Old Dominion University ODU Digital Commons

Sociology & Criminal Justice Theses & Dissertations

Sociology & Criminal Justice

Summer 2013

Rez Realities: Exploring the Perceptions of Crime and Justice Among Tribal Police Officers in Indian Country

Favian Alejandro Martin Old Dominion University

Follow this and additional works at: https://digitalcommons.odu.edu/ sociology_criminaljustice_etds

Part of the <u>Criminology Commons</u>, <u>Indigenous Studies Commons</u>, <u>Race and Ethnicity</u> <u>Commons</u>, and the <u>Regional Sociology Commons</u>

Recommended Citation

Martin, Favian A.. "Rez Realities: Exploring the Perceptions of Crime and Justice Among Tribal Police Officers in Indian Country" (2013). Doctor of Philosophy (PhD), dissertation, Sociology/Criminal Justice, Old Dominion University, DOI: 10.25777/kn8r-1k08 https://digitalcommons.odu.edu/sociology_criminaljustice_etds/30

This Dissertation is brought to you for free and open access by the Sociology & Criminal Justice at ODU Digital Commons. It has been accepted for inclusion in Sociology & Criminal Justice Theses & Dissertations by an authorized administrator of ODU Digital Commons. For more information, please contact digitalcommons@odu.edu.

REZ REALITIES: EXPLORING THE PERCEPTIONS OF CRIME AND JUSTICE

AMONG TRIBAL POLICE OFFICERS IN INDIAN COUNTRY

by

Favian Alejandro Martín B.S. May 2007, Pennsylvania State University, Harrisburg M.A. May 2009, Pennsylvania State University, Harrisburg

A Dissertation Submitted to the Faculty of Old Dominion University in Partial Fulfillment of the Requirements for the Degree of

DOCTOR OF PHILOSOPHY

CRIMINOLOGY & CRIMINAL JUSTICE

OLD DOMINION UNIVERSITY August 2013

Approved by:

Mona J. E. Danner (Director)

Travis Linnemann (Member)

Scott Maggard (Member)

antRoss (Member)

ABSTRACT

REZ REALITIES: EXPLORING THE PERCEPTIONS OF CRIME AND JUSTICE AMONG TRIBAL POLICE OFFICERS IN INDIAN COUNTRY

Favian Alejandro Martín Old Dominion University, 2013 Director: Dr. Mona J. E. Danner

Although American Indians and Alaskan Natives (AI/NA) compose just over 1% of the general population in the U.S., they experience higher rates of crime and violence than the total population and are dramatically over-represented in the criminal justice system. In light of these realities, the paucity of research on AI/NA crime, violence, and justice problems is appalling. What research does exist suggests that AI/NA crime and victimization is correlated with social problems such as poverty and illegal drug use which are linked to the social ills of colonialism. Drawing on the work of Loader (1997), this dissertation examines the perceptions of crime and justice among tribal police officers in Indian Country.

Police officers are charged by the state with the responsibility to maintain social control and seen as the authoritative discourse of crime and justice in society. Within their position, they have the ability to "diagnose" all crime related problems (Loader 1993). Given their unique position in society, the police also have the power to frame the discourse on crime and justice and their experiences influence public opinion and policy. Throughout Indian Country, tribal police officers possess unique knowledge about all crime-related problems and offer a discourse on crime and justice on reservations.

This dissertation uses in-depth interview methods to address the research question: What are the perceptions of crime and justice in Indian Country among tribal police officers? Interviews with 27 tribal police officers on a southeastern Indian reservation were conducted to explore the tribal police officers perceptions about crime and justice. The tribal police officers' perceptions revealed a tribal community suffering from high rates of prescription drug use that were correlated with property crimes and high levels of domestic violence and assaults. This is substantiated by the current crime data: as creators of discourse and community perceptions, the view of the criminal justice personal are reified in the community. The participants spoke about the tribal court system as dysfunctional due to inter-tribal politics, federal restrictions, and lenient tribal judges. Tribal police officers also portray the federal government as too often failing to prosecute serious crimes on the reservation. As a result, they believed that both the tribal and federal criminal justice systems did not adequately address crime on the reservation. This dissertation is dedicated to my parents Jennie and Carlos Martin

.

ACKNOWLEDGEMENTS

First, I thank my dissertation Committee: Dr. Mona J.E. Danner, my Dissertation Chair, for her patience, guidance, and constructive comments. Dr. Travis Linnenman for his in-depth knowledge of critical criminology that proved to be extremely helpful during the theoretical analysis. Dr. Scott Maggard for his helpful and insightful comments on the dissertation research. Lastly, Dr. Jeffrey Ian Ross for serving as my external committee member and sharing his knowledge on American Indians and their social problems.

Next, I thank my mentors for their invaluable advice. First, Dr. Mona Danner for her unwavering support in my academic endeavors from providing words of encouragement to finding sources of funding for various research projects. To Dr. Dawn Rothe, for her guidance and life-changing opportunities such as studying aboard, which exposed me to different cultures and perspectives. I am thankful for your openness and your ability to challenge my preconceived notions on all things in life. Thanks to Dr. Travis Linnenman, for your advice on life in academia. Taken together, my mentors encouraged me to find my "academic voice," and for that, I am eternally thankful.

I am also indebted to my family and friends. To my family, whose encouragement and support greatly helped me throughout the last four years. The many "care packages" and cards significantly helped during those bleak moments and brought much relief. In addition, many thanks to my friends Victoria Collins, Amanda Farrell, and Jesse McKee for their friendship and support. Lastly, to Jeff for being supportive and offering words of encouragement.

To you all, I am eternally thankful.

TABLE OF CONTENTS

۲

Chapter	Page
I. THE PROBLEM OF CRIME AND JUSTICE IN INDIAN COUNTRY	1
SUMMARY OF CHAPTERS	
II. COLONIALISM – THE STORY OF CONFLICT	10
COLONIALISM	10
THE CURRENT STATE OF AMERICAN INDIANS IN THEIR	
COMMUNITIES	
AMERICAN INDIAN CRIME AND VICTIMIZATION	
POLICING AMERICAN INDIANS	
TRIBAL JUDICIAL SYSTEM	
CONCLUSION	44
III. THE LITERATURE ON POLICE PERCEPTIONS	16
THE POLICE	
POLICE PERCEPTIONS	
CONCLUSION	
IV. THE THEORETICAL FRAME – THE SYMPOLIC POWER OF POLICING	G61
SYMBOLIC POWER OF THE POLICE ON THE REZ	
V. METHODS	
RESEARCH PARTICIPANTS	
EASTERN BAND OF CHEROKEE INDIANS	
IN-DEPTH INTERVIEWS	
RESEARCH PLAN	
PROTECTION OF SUBJECTS	79
VI. FINDINGS	0 0
POLICE DEPARTMENT DEMOGRAPHICS	
COMMUNITY LIFE	
CRIME AND VIOLENCE	
ILLEGAL DRUG USE	
CRIME	
PROPERTY CRIMES	
VIOLENT CRIMES	
CRIME CAUSATION	
CRIME PREDICTIONS	
TRIBAL JUSTICE	110

INFORMAL RESOLUTIONS	124
IDEAS FOR IMPROVEMENT	
SUMMARY OF FINDINGS	
VII. DISCUSSION	
RELEVANCE OF THEORETICAL FRAMES	
COLONIALISM	133
SYMBOLIC POWER OF POLICING	
POLICE PERCEPTIONS	
POLICY IMPLICATIONS	
CONTRIBUTIONS AND FUTURE DIRECTIONS	155
CONCLUSION	158
APPENDIX	
INTERVIEW SCHEDULE	160
REFERENCES	162
VITA	209

CHAPTER I

THE PROBLEM OF CRIME AND JUSTICE IN INDIAN COUNTRY

American Indian and Alaskan Natives (AI/NA) living on reservation communities are surrounded by the natural beauty of the landscape and the richness of their ancestral culture and heritage. Sadly, these idyllic realities stand in sharp contrast to the crime commonly found on those same lands. Simply put, AI/NAs experience some of the highest rates of crime and violence of any group in the nation and violent crime rates among these individuals remain high, despite a decrease in the violent crime in the U.S. generally (Perry 2004). Crime and violence among AI/NAs is even more pervasive in reservation communities, as reported by Williams (2012a:1): "Indian reservations across the United States have grappled for years with chronic rates of crime higher than all but a handful of the nation's most violent cities." Other journalists have characterized reservation communities as "war zones" to describe the high rates of crime, violence, and victimization that occur in Indian Country (Moore 1990). Scholars would argue that the hegemonic use of this metaphor serves as a propaganda tool to serve political interests. As Steinert suggest (2003:265-266):

Methaphors of war have longed plagued thinking about crime – in the USA predominantly, but now that the "Fifty Years Peace" is over, Europe is fast catching up. The metaphor of war has acquired a solid place in hegemonic thought – perhaps akin to it role during the first half of the 20th century. It has invaded (!) the public discourse in economics, international relations, sport, even some medical specialties – and, of course, crime policy and policing.

Indeed, the reservation war zone analogy "feeds on dramatizations and the creation of events that can be dramatized" (Steinert 2003:266). That is to say, that this characterization suggests that reservation communities are similar to cities and towns in

war torn countries such as Syria, Egypt, and Afghanistan. One newspaper article compared several Indian reservations' crime-fighting initiative to the Iraq war military "surge" strategy. As Williams (2012c:1) notes:

Nicknamed "the surge," it was modeled after the military's Iraq war strategy, circa 2007, which helped changed the course of the conflict. Hundreds of officers from the National Park Service and other federal agencies swarmed the reservations, and crime was reduced at three of the four reservations – including a 69 percent decline at Mescalero Apache in New Mexico, officials said. Wind River, as has been true for much of its turbulent history, bucked the trend: violent crime there increased by 7 percent during the surge, according to the Department of Justice.

Returning to the "war zone" characterization, many commentators on Native social problems have used this term to describe the high rates of violence against American Indian women. As Chief Elk (2012:1) argues, "those of us in the Native American community know we are being targeted—and as I read about sexualized violence against civilians in war zones, I can't help but draw parallels to the brutality against Native women." Similarly, Sarah Deer (2006:1), an attorney for the Tribal Law and Policy Institute states, "American Indian and Alaska Native women are living in a virtual war zone, where rape, abuse, and murder are commonplace and sexual predators prey with impunity." Further, media headlines such as "Brutal Crimes Grip an Indian Reservation," "Death Toll Increases in Indian Reservation Rampage" and "Mass Murder on Indian Reservation in North Dakota, Victims Includes Children" conjure images of American Indians living in war zones. Equally alarming are narratives of a failed criminal justice system in Indian Country. The Denver Post reported that law enforcement in Indian Country is "dysfunctional" (Riley 2006a:1) and the failure of the system has contributed to "lawless lands" where violence and victimization is part of everyday life on the reservation. Terms such as "war zones" and "the surge" also invoke images of

American Indians as being savages wreaking havoc and destruction within their communities. Further, it gives the impression that reservation communities are on the verge of, if not already experiencing, total anarchy. Seemingly, Indian reservations and war-like violence appeared to be inseparable concepts in the media. Despite these realities, AI/NA's crime problems "remain among the least studied" in the field of criminology (Morris, Wood, and Dunaway 2006 2006:573).

In her address upon receiving the American Society of Criminology's 2011 Sutherland award, Peterson (2012) argued for the need to examine race and crime issues. Citing several studies that show the criminalization of minorities, she illustrated the ways that racialized policies within the criminal justice system have contributed to their high incarceration rates. In her critique Peterson focused on African-Americans and Latinos within the criminal justice system. Indeed, African-Americans and Latinos experience high rates of crime, violence, and incarceration, but the AI/NA population is increasingly experiencing even higher levels of these social problems than other minority groups. Despite their high rates, criminology continues to overlook these problems. As Perry (2009:3) asserts:

The paucity of research on Native American issues is more than simple perplexing. It is disturbing. Native Americans represent the most disproportionately over-represented population both in terms of victimization, and arrest and incarceration rates.

AI/NAs are increasingly involved in the criminal justice system. As Greenfeld and Smith (1999:26) note:

The estimated 62,600 American Indians with a correctional status accounted for just over 4% of the American Indian adult population. By comparison, an estimated 2% of white adults, 10% of black adults, and less than a half of 1% of Asian adults were under correctional supervision.

Recent figures suggest that on average 1 in 25 AI/NAs are under the auspices of the criminal justice system (Perry 2004). They are also subjected to over-policing which contributes to high levels of incarceration (Perry 2009). AI/NA arrest rates are substantial because on average they are first arrested at the age of 14 and can have on average an additional 19 arrests during their lifetime (Grobsmith 1994). The high level of involvement in the criminal justice system among AI/NAs is correlated to un- or underemployment, drug and alcohol use, lack of education attainment, poor parenting, and child abuse (Bonta 1989; Grobsmith 1994; National Institute for Justice (NIJ) 2005).

Al/NAs are more likely to be involved in crime as both the offender and/or the victim than are the general U.S. population (Nielson and Silverman 1996; Greenfeld and Smith 1999; Perry 2004). More importantly, while the national violent crime rate declined in the 1990s, violent crime rates among Al/NA remained high even when compared to other racial/ethnic groups in the U.S. (Greenfeld and Smith 1999; Perry 2004; Toth, Crews and Burton 2007; Feagin and Feagin 2008). Given the extraordinary rate of crime, violence, and victimization among Al/NAs, several scholars argue that these issues are a race and crime problem which can be traced to the effects of colonialism (Gabbidon and Greene 2005).

Blauner (1972:53) defines colonialism as, "nations [that] incorporate new territories or peoples through processes that are essentially involuntary, such as war, conquest, capture, and other forms of manipulation." The mechanisms of colonialism preserve the belief that minority groups are more predisposed to crime behaviors than their white counterparts and represent the foundation of policies that are directed towards minorities in order to address their "criminal behavior." Some criminologists utilize the colonel model to explain the high rates of crime and criminalization among minorities. Fanon (1963) asserts that the colonization process impacts the cultural and structural status of minority groups and that the process occurs when the dominant group forcefully takes over the territory of an indigenous group. He claims that in colonial societies European whites are economically and politically dominant, which allows them to maintain their culture and values, and impose them on minority groups. During the colonization process, the minority group's culture is reduced to a lower status or destroyed, thus forcing the group to adhere to the culture, beliefs, norms, and rules of the colonizer. The minority group has no other option but to assimilate into the dominant culture.

The colonial process also establishes a caste-like system. Fanon (1963) argues that the colonizers maintain control over the resources in terms of economic, political, and social benefits, thus limiting the resources available to minority groups. Colonizers utilize the social construct of race to delineate the minority group to a low-class status. In order to maintain a system of subordination, the dominant group establishes "representatives of the power structure," such as police and other law enforcement officers who reinforce the status of minority groups and protect the assets of the dominant group. The structural relationships in colonial societies impact the behavior of minority individuals, and lead to feelings of alienation among them. In turn, alienation from one's self, racial group and culture may produce assimilation resistance, or violence among individuals. When violence does occur it is more likely intra-group, rather than intergroup.

Fanon (1963), Blauner (1994) and Staples (1987) suggest that minority groups in the U.S. are internal colonies that are controlled and exploited systematically by whites who are the majority group. Additionally, they argue that there is an unequal structural relationship between majority and minority groups in terms of economic, political, and social power. This unequal structural relationship leads to frustration and alienation among minority individuals, thus causing them to engage in criminal behaviors (Tatum 2000). The colonial model holds that racial inequality in the criminal justice system is created by whites (the colonizers) who target minority groups (the colonized); the structural conditions result in those colonized experiencing high rates of arrest and imprisonment. Due to these disparities, minorities recognize that they are less likely to receive equal justice. The colonial model holds that whites systematically control and exploit minorities through the criminal justice system to ensure white power over valued resources. Minority criminal behavior is seen as a threat to the status quo of white control. AI/NAs have been subjected to anti-Native governmental policies that subjugate them to the lower fringes of society, in order to maintain white superiority and power, and that result in significant social problems including geographical isolation, substance abuse, and mental health problems, among others.

AI/NA crime and violence are correlated with social problems such as poverty (Bachmann 1991a, 199b, 1992; Fitzgerald and Carrington 2008; Lanier and Huff-Corzine 2006; Painter-Davis 2012) and illegal drug use (Winfree, Griffiths, and Sellers 1989; Silverman 1996; Whitbeck, Hoyt, McMorris, Chen, and Stubben, 2001; Bohn 2003; Wahub and Olson 2004; Morris et al., 2006; Yuan, Koss, Polacca, and Goldman 2006; Bachman Zaykowski, Lanier, Poteyeva, and Kallmyer 2010). In turn, these problems are linked to the effects of colonialism (Snipp 1992). Several scholars argue that substance abuse serves as the impetus for a majority of crimes that occur in reservation communities (Greenfeld and Smith 1999; Perry 2004), and Indigenous activist Noel Pearson (2001) acknowledges that substance abuse is a leading facilitator in Native violence and crime. Further, Native substance, violence, and crime are traced to historical trauma and cultural oppression. As Pearson (2001:4) notes:

We must understand that trauma, dispossession et cetera make our communities susceptible to grog and drug epidemics, they do not automatically cause abusive behavior...when abusive behaviors is deeply entrenched in our communities it is not the material destitution, the social ills and historical legacy that fuel the abusive epidemics. It is the epidemics themselves.

In response to high rates of crime and violence in reservation communities, many tribal governments implemented their own judicial system to address these problems. Tribal courts are public and formal sites where generational conflicts may occur with respect to ideas about crime, violence, and tribal culture. Historically, AI/NAs have administered justice and resolved conflict via ancestral customs, traditions and laws. While a majority of the 560 federally recognized American Indian tribes utilize traditional methods of resolution to address conflicts and disputes, many tribal governments have adopted Anglo-American judicial systems to address criminal and deviant behaviors. Tribal courts have limited jurisdictional powers and may only hear misdemeanor cases, protection orders, and cases involving juvenile offenders, in contrast to the federal government's unlimited jurisdiction to prosecute crimes that occur on tribal lands (Pevar 2004). As a result, many serious offenders are hardly prosecuted for their crimes, which instills a sense of vulnerability among tribal members (National Victim Assistance Academy 1999). Despite having a tribal justice system, many communities are unable to effectively address crime and victimization due to federal laws. Unfortunately, the violent and dysfunctional predicaments experienced within reservation communities are not widely discussed and the needs of AI/NAs are often neglected.

This dissertation examines the perceptions of crime and justice among tribal police officers. Within their occupation, tribal police officers are able to provide insight about crime, violence, and justice in their reservation community. Loader (1997) and Loader and Mulcahy (2003) illustrate the cultural processes which allow the police to speak on all crime related matters within society. Referred to as the "symbolic power of policing," these authors suggest that police officers are the "experts" and able to "diagnose, classify, authorize and represent both individuals and the world" (Loader 1997:3). Similar to their mainstream counterparts, tribal police officers have "ownership and framing of relevant issues, and whose individual and collective utterances circulate meanings that contribute in potentially telling way to the formation of opinion and belief" (Loader and Mulcahy 2001:42). In this research, the tribal police officers offer an "diagnosis" on all crime related matters taking place within their reservation community. This dissertation research draws on the works of Loader (1997) and Loader and Mulcahy (2003), to explore the dynamics of crime, and justice on the Eastern Band of Cherokee Indian Reservation in North Carolina. An in-depth interview method was used to address the research question: What are the perceptions of crime and justice in Indian Country among tribal police officers?

SUMMARY OF CHAPTERS

The following chapters will provide the frame of understanding in investigating the perceptions of crime and violence among tribal police officers in Indian Country. Chapter II provides a socio-historical analysis of the treatment of American Indians, with specific attention to the role of colonialism. The colonial legacy and its social ills have been detrimental to the Indigenous population historically and continue to be a source of contemporary social problems for them. Due to colonialism, American Indians experience a myriad of social problems such as poverty, suicide, and substance abuse, all of which are strongly correlated to their high rate of crime and victimization. This chapter will also provide a discussion on the frequency and characteristics of AI/NA crime and victimization. In addition to these statistics, the chapter provides an overview of the tribal judicial system. Chapter III provides a literature review on past research on police perceptions. Chapter IV offers a review of Loader's (1997) the symbolic power of policing perspective as it relates to the research question: What are the perceptions of crime and justice in Indian Country among tribal police officers? Chapter V provides an overview of the methodological design of in-depth interviews that was utilized to address the research question. Chapter VI presents the analysis including major themes that emerged from data and Chapter VII contains a summary of the dissertation research as well as a consideration of the contributions and implications of the project.

CHAPTER II

COLONIALISM - THE STORY OF CONFLICT

American Indian/Native Alaskan (AI/NA) social problems have their origins in colonialism. As Ross and Gould (2006:3) note, "the study of law, justice, and criminal justice issues in Indian Country is complex and laden with cultural, economical, political, and social undertones that are bound in a deeply rooted historical context." Social, political, and economic institutions subjugate AI/NA to the lower fringes of society, and as a result allow these individuals to experience crime and violence as well as victimization. To understand how these mechanisms evolved, this chapter examines the role of colonialism as it relates to the current state of AI/NAs because the realities and perception of crime and justice are shaped by the effects of colonialism.

COLONIALISM

Prior to the European conquest of the Americas, an estimated 15 million AI/NAs inhabited North America (Sale 1990); four centuries later, by the late 1800s, this population was substantially reduced to just 250,000 (Healey 2003). During the colonization period and thereafter, the AI/NA people experienced an array of victimizations at the hands of Anglo Whites, ranging from acts of genocide to forced relocation which contributed to the decline of the population (Salisbury 1996; Churchill 1997; Smith 2005). Not only did the effects of colonization decimate the indigenous population, it also shifted the cultural and structural status of these individuals, which allowed the dominant Anglo group to forcefully take over their ancestral lands. In colonial societies, European Anglos were economically and politically dominant, affording them opportunities to maintain their group's culture, beliefs, and values, while simultaneously imposing these attributes on the minority population (Fanon 1963). The colonization process afforded Anglo Whites an opportunity to destroy the minority group's culture and coerce them to adopt and adhere to their cultural system (Blauner 1972, 1994; Staples 1987).

The colonization process affords the colonizer an opportunity to establish a castesystem to ensure white dominance over economic, political, and social resources, which continues into the present by limiting minority groups' access to valued resources such as education and employment (Fanon 1963, 1967; Pencak and Richter 2004). This mechanism uses race to relegate minority groups to the outer fringes of society (Fanon 1963, 1967; Massey 2007), it is a system of subordination based on race. Additionally, colonialism is a systematic method of social stratification. Massey (2007:242) argues that the social stratification system hinges on three factors: a social structure that categorizes individuals based on a combination of their achieved and ascribed traits, the stereotyping of exploited groups into a lower-class status, and the restriction of socially valued resources to the colonizers. As Massey (2007:243) notes:

Social groups can be classified along two fundamental dimensions that define the conceptual space of social cognition: warmth and competence. People naturally frame themselves and other like them as both warm (likable, approachable, trustworthy) and competent (efficacious, capable, astute). People perceived in this way are seen as in-group members, or at least as members of groups that are very similar to the in-group. People who are framed as lacking either warmth or competence are socially defined as members of the out-group – as others who are not perceived as "people like us."

In conjunction with placement of individuals based on warmth and competence characteristics, the in-group also implements social mechanisms to ensure the protection of resources for themselves, while resources are extracted from the out-group without compensation. For Massey, there are two mechanisms that enforce and reproduce stratification to benefit dominant groups: opportunity hoarding and exploitation. Massey (2007:244) refers to opportunity hoarding as, "the monopolization by in-group members of access to a resource so as to keep it for themselves." This mechanism allows the ingroup (whites) to dominate valued resources such as educational and employment opportunities so that they are not in competition with out-group (minorities). Massey (2007:244) suggests that exploitation "is the expropriation of resources from an outgroup by members of an in-group such that out-group members receive less than full value for the resources they give." This mechanism permits whites to pay members of the out-group lower wages, for instance. Massey's theory of social stratification is especially evident in the historical and current oppression of the AI/NA population. Throughout the history of the United States the AI/NA population has been subjected to mechanisms of exploitation and opportunity hoarding. Conflicting views of human nature between AI/NAs and Anglo Whites contributed to the conflict.

The conflicting views between Anglo Whites and AI/NAs "has to do with the deeply rooted religious and spiritual underpinnings on which the various European and native cultures are based" (Ross and Gould 2006:4). While Anglo Whites were deeply patriarchal, practiced Christianity, and abided by individual property rights (Washburn 1971; Flint 1971; King 1999; Ross and Gould 2006), the American Indian population was matrilineal, viewed the land as being sacred, and did not understand the concept of landownership (Washburn 1971; Snipp 1992; Churchill 1997, 2004). With such conflicting and opposing ideologies and beliefs, and in the face of superior Anglo

firepower, the AI/NA population experienced social disruption and ethnic violence (Perry 2006).

As a result of colonization, AI/NAs experienced the most severe forms violence and ethnocide at the hands of Anglo Whites (Perry 2002), such acts included genocide (Snipp 1992; Churchill 1997), government sponsored warfare (King 1999), enslavement (Calloway 2006), and forced relocation (Perdue 1985). However, the most detrimental factor in the decline of the AI/NA population was transmission of European diseases (Snow and Lanphear 1988; Snipp 1992; King 1999; Calloway 2006). AI/NAs were not immune to diseases such as smallpox and measles, and in some cases entire tribes were lost to these infections. For example, from 1618 to 1619, a small pox epidemic killed nearly 90% of Mohawk Indians in Massachusetts. Realizing the destructiveness of European diseases upon the indigenous population, the federal government actively engaged in biological warfare to exterminate these individuals (Churchill 1997). While AI/NAs suffered from these diseases, Anglo Whites organized to gain landownership of tribal lands. Many of the social ills within reservation communities can be traced to government policies and initiatives that led to the development of Indian reservations.

Although the United States created treaties with American Indian tribes to establish land boundaries (Ponnersheim 1995), many of these agreements were violated in order to accommodate a growing white and immigrant population in the country. For instance, in the 1840s the United States operated under the belief of "Manifest Destiny" that it was America's right to move into the western territories of the country, which provided a justification for western expansion (King 1999). In order to circumvent existing treaties with Native tribes, the federal government created many Indian removal policies such as the Indian Removal Act of 1830, which oversaw the relocation of 100,000 Natives (Perdue 1985; McEwan 2000; Perdue and Green 2005; Conley 2005). Under the Act, the President had the authority to negotiate treaties with Indian tribes for their relocation to west of the Mississippi River. While some tribes voluntarily relocated, others were pressured or deceived into entering an agreement (Conley 2005). The relocation of the Cherokee Nation, for instance, occurred when federal agents duped a dissident group of Cherokee Indians, none of whom were tribal leaders, into signing the Treaty of New Echota (Perdue 1989). As a result, the treaty forced the relocation of over 17,000 Cherokee Indians in the southeast to an Indian reservation located in the Oklahoma territory (Perdue and Green 2005). Nearly 5,000 Cherokee Indians died from exposure, exhaustion, and disease from this forced relocation, now referred to as the "Trail of Tears" (Carter 1976; Perdue and Green 2005). The federal government's relocation policies still did not quell the thirst for more land to support a growing U.S. population.

By the 1800s, many mid-western American Indian tribes found their ancestral lands encroached by an influx of white settlers. To accommodate the influx of settlers, companies and corporations established railroads and telegraph poles on tribal lands (Banner 2005). Similar to their eastern American Indian counterparts, western Native peoples found themselves increasingly forced off their lands. Perhaps more detrimental was that many Anglo Whites actively hunted and nearly wiped out buffalo, an important source of food and clothing for AI/NAs. Although many tribes were engaged in several skirmishes with Anglo Whites, it was not until after the American Civil War that the federal government directed its full attention to diminishing Indian resistance in the Plains. By the end of the 1800s, the federal government and many Indian tribes found themselves engulfed in the "Indian Wars," which included notable battles such as the Sioux Uprising (1862), the Sand Creek Massacre (1864), the Battle of Little Big Horn (1876), and Wounded Knee (1890). Fearing a long war, the federal government engaged in a series of genocidal practices as Wellman (1934:8) notes:

The Indian [was thought] as less human and worthy only of extermination. We did shoot down defenseless men, and women and children at places like Camp Grant, Sand Creek, and Wounded Knee. We did feed strychnine to red warriors. We did set whole villages of people out naked to freeze in the iron cold of Montana winters. And we did confine thousands in what amounted to concentration camps.

After their defeat in the "Indian Wars," the AI/NA peoples were relocated to Indian reservations.

The federal government passed numerous laws that greatly reduced their rights and severely restricted Native access to resources. In 1871, Congress passed the Indian Appropriations Act which ended the government's recognition of AI/NA tribes as sovereign nations and banned additional Indian treaties. In years to follow, the federal government passed aggressive assimilation policies in hopes of "transforming" Indians into farmers. For instance, Congress passed the Dawes Act of 1887, which divided tribal lands into allotments and distributed them to Native families (Haas 1957) to encourage farming (Healey 2003). The Act was a failure because the Native population did not have a history of farming, the government did not prepare Natives for the transition, and the land was inadequate for agriculture (Carlson 1978). In reality, the underlying intention of the Act was to restrict Native access to their ancestral lands while making these areas available to Anglo Whites. As a result of the Act, nearly 900,000 Natives were displaced and lost an estimated 90 million acres of land, most of which were later sold to White settlers (Feagin and Feagin 2008). As such, the Dawes Act served as a mechanism of exploitation of AI/NAs.

In 1928 the Miriam Report noted the severe abuses that were caused by the Dawes Act, and led to further investigations of fraud and corruption. In one state, tribal lands were reduced from 138 million acres to 47 million acres after the passage of the Act. The Miriam report served as the impetus for the Indian Reorganization Act of 1934, which attempted to correct and stabilize Native communities. Under this Act, the federal government appropriated funds to recover lands that were sold under the Dawes Act and provided federal protection (Haas 1957). For the first time, the government realized that "federal policy had failed because it was based upon the false premise that all Americans should conform to a single, uniform cultural standard" (Kelly 1975:291). By the 1950s, many tribal communities faced a series of setbacks as the federal government reduced the "special treatment because of their status" (Walch 1983:1188) including reducing tribal authority over their members, social service funding, and land holdings. The vagaries of US government policies towards AI/NAs are perhaps best represented by subjugation of Native youth to Indian boarding schools.

Indian boarding schools have been used extensively since the colonization of the Americas. In 1723, the College of William and Mary created one of the first Indian boarding schools in the US (Swem 1928). With the financial backing of Sir Robert Boyle, the College initiated a program that promoted Christian beliefs among the Native juveniles. In addition, the school taught these juveniles the English language and other Anglo cultural norms. The Royal Governor of Virginia, Francis Nicholson, supported the College's goal of educating Native juveniles. In fact, the Governor went as far as to encourage local Indian tribes to send their tribal youth to be educated at boarding school in hopes that they would go back to their ancestral homes and educate their people of the "English ways." Despite the intention of the school, the program failed to assimilate Natives into the Anglo culture as many Native juveniles abandoned Christian beliefs and resumed their tribal customs and rituals upon reentering their tribal communities. While the College of William and Mary's Indian boarding school might have been viewed as a failure, the idea of educating Native juveniles did not lose momentum until years later.

During the late 19th and into the 20th centuries, many Indian boarding schools were established in the U.S. to educate Native juveniles. Initially, Christian missionaries constructed boarding schools on several Indian reservations for juveniles who were unable to receive an education due to the geographical isolation of their tribal community (Churchill 1997; Smith 2005). Years later, the U.S. government saw the usefulness of the Indian boarding schools and allocated resources to these institutions (Noriega 1992). For instance, the government approved the construction of the Carlisle Indian Industrial Boarding school on the grounds of the Carlisle War Barracks in south central Pennsylvania (Witmer 1993). Founded in 1879, the school was one of the first boarding schools to operate outside of Indian reservation lands. The institution's main goal was to assimilate Native juveniles into the dominant Anglo culture. The school's overseer, Captain Richard Henry Pratt, believed it was the mission of the school to remove Native juveniles from their tribal culture, customs, language and rituals (Witmer 1993). He openly admitted that it was the school's goal "to civilize the Indian, get him into civilization. To keep him civilized, let him stay" (Pratt 1964: 283). By 1894, the Carlisle Indian Industrial School had 818 students who represented 53 tribes such as the

Chippewa, Oneida, Sioux, Apache, Seneca, Ottawa, Cherokee, and Nez Perce (Reyhner and Eder 2004). Pratt (1910:856) also set the tone for Indian Boarding Schools for decades to come, by stating:

We shall not succeed in Americanizing the Indian until we work on him exactly the same way...By means of the public school, Native Americans would take their place in the mainstream of American culture.

As a result, the Carlisle Indian boarding school served as a model for other schools in an

attempt to assimilate Native juveniles into the Anglo culture.

Following the Carlisle Indian boarding school, the Federal Board of Indian

Commissioners provided a rationale for more schools:

As a savage, we cannot tolerate him any more than as a half-civilized parasite, wanderer or vagabond. The only alternative left is to fit him by education for civilized life. The Indian, though a simple child of nature with mental facilities dwarfed and shriveled, while groping his way for generations in the darkness of barbarism, already sees the importance of education (Prucha 1986:194).

As a result, U.S. Congress passed the Compulsory Indian Education Act of 1887 and allocated funds to build more Indian boarding schools (Child 1993). At these schools Native juveniles were coerced to assimilate into the Anglo culture, including adopting new traditional European names, speaking English instead of their native language, wearing European/American style clothing or uniforms, getting haircuts, and accepting Christianity as their sole religion (Churchill 1997; Smith 2005). They were told by their school teachers that their ancestors were evil and hedonistic and their schoolbooks depicted their ancestors as demonic and responsible for the murder of white women and children (Churchill 1997; Smith 2005). The hostile depictions of American Indians were an attempt to persuade Native juveniles to adhere to the dominant culture and norms

(Henson Taylor, Curtis, Cornell, Grant, Jorgensen, Kalt, and Lee 2007), as well as convey the idea that AI/NA peoples were inferior to the Anglo society.

The popularity of Indian boarding schools grew and by 1909 the Bureau of Indian Affairs (BIA) established twenty-five non-reservation Indian boarding schools with a total of 6,000 students enrolled at these institutions (Adams 1995). Indian boarding school enrollment peaked in the 1970s, when 60,000 Native juveniles attended these institutions. Despite the wide perception that these schools had a positive impact on these juveniles, there were also reports of abuse.

By all accounts the experiences at these schools were harsh. In some Indian boarding schools, the Native youth were trained to do manual labor or farming. Many Native female juveniles were leased to white home owners to do housework (i.e. cleaning, sewing and washing). However, several reports found that Native juveniles experienced a myriad of abuses in these institutions that ranged from neglect to sexual abuse (Smith 2005, 2007). The Meriam Report (1928) documented the problems within these schools including high levels of infectious diseases due to lack of nutritious meals, inadequate funding for medical supplies, poor sanitary conditions, and overcrowding problems in these institutions. The report also found that the death rate for Native students was six times higher than for non-Native students. Despite the awareness of abuses occurring within these schools, many Native juveniles continued to experience physical, emotional, and sexual abuse. In some cases, Indian students were sexually abused school by officials (Smith 2005) and in other cases older Indian students were forced to physically or sexually abuse younger students (Churchill 1997, 2004; Smith 2005).

Although school officials and the BIA were aware of these abuses, they largely ignored the complaints coming from students and parents. Federal Bureau of Investigation (FBI) investigations found that a small number of Indian boarding school teachers were responsible for sexually abusing hundreds of students (Smith 2005). For example, in 1987 a school teacher at the Hopi Indian boarding school in Arizona was found to have sexually abused over 140 male students. In another case, a school official working at the Navajo Reservation School was investigated amid reports that he molested twelve students. It was not until the passage of the Indian Child Protection Act of 1990 that the BIA and the Indian Health Service (IHS) put in place accountability measures to ensure that Native juveniles were protected from abuse. Perry (2002: 236) contends that the use of Indian boarding schools represents "a history of violence, intimidation, and repression." Other scholars argue that it was in these boarding schools, while alienated from their reservation communities and families, that Native juveniles learned abusive behaviors (Noriega 1992; Brave Heart-Jordan 1995). Upon returning to their reservation communities, they turned to alcohol and other vices to cope with feelings of isolation and despair. Indian boarding schools have negatively influenced generations of AI/NAs.

The historical anecdotal narratives and reports pertaining to the impact of colonialism among the AI/NAs demonstrated the disastrous results for the population. Aside from being victimized through extermination and forced assimilation policies, generations of AI/NAs witnessed a substantial decline in both their population and their people's participation in the ancestral culture. Colonialism continues to be a source of historical trauma and cultural oppression, and is reflected in the current state of the AI/NA population.

THE CURRENT STATE OF AMERICAN INDIANS IN THEIR COMMUNITIES

Census figures report that there are 5.2 million AI/NAs in the U.S. comprising about 1.7% of the total population (Norris, Vines, and Hoeffel 2012). Almost half of the total AI/NA population lives in one of the 562 Indian reservations located throughout the U.S. (Ogunwole 2006). Unfortunately, past federal policies of forced relocation and isolation continue to cause AI/NA peoples to experience economic deprivation and social exclusion.

Poverty

Recent poverty statistics paint a grim situation for many AI/NA families as this segment of the population experiences the highest rate of poverty in the U.S. (27.3%), almost two times higher than the national average (14.2%) (U.S. Department of Commerce 2009). Economic problems on the reservation are further exacerbated by the lack of employment opportunities available for the AI/NA population (Sandefur 1989). Employment figures reveal that only 36% of all American Indian males living in reservation communities are able to secure full-time employment. On a community level, the rates of poverty are equally high. The Blackfoot Reservation in Montana, for instance, experiences an unemployment rate of nearly 70%. In order to sustain themselves, nearly half of the AI/NA population relies on federal assistance in the form of food stamps and welfare checks (Feagin and Feagin 2008). Almost 40% of Native juveniles attending school live with poor families (Orfield and Lee 2005). Over 60% of all Native eighth graders receive free or reduce lunch (Moran and Rampey 2007). The impoverished conditions among Native peoples extend to the infrastructure of many tribal homes. For

instance, many reservation homes lack electricity and almost 12% of these buildings do not have adequate plumbing or sewage facilities (Pickering 2000). These realities are contrary to the popular belief that many reservation communities are wealthy due to Indian gaming casinos.

Indian Gaming

Indian gaming casinos grew popular during the Reagan administration. Initially, the federal government promoted the belief that Indian gaming casinos would allow reservation communities to become economically sufficient and enhance tribal sovereignty. There are nearly 400 Indian gaming casinos across the U.S. operated by 224 tribal governments in 28 states, according to the National Indian Gaming Commission (2004). While popular belief holds that Indian gaming casinos generate a substantial amount of funding for reservation communities, the reality is otherwise. According to the United States General Accounting Office (1998), the Indian gaming industry averages nearly \$4.5 billion per year, but only eight tribes account for 40% of these gaming profits. Although the National Indian Gaming Commission projects that Indian gaming casinos will generate \$18.5 billion per year, only 12% of all Indian gaming casinos will account for 65% of the projected rate. Thus, the majority of Indian gaming casinos are not sufficiently successful to help alleviate the impoverished conditions that plague many reservation communities. Research indicates that Indian gaming casinos have led to an increase in compulsive gambling among AI/NA peoples and exacerbated other health problems (Wardman, El-Guebaly, and Hodgins 2001).

Health Problems

A myriad of health problems among AI/NA people results in a lower life expectancy compared to the general U.S. population (Indian Health Service 2007). These health problems can be traced to the legacy of colonialism. Nearly 80% of the Native population died following exposure to European diseases such as tuberculosis, measles, smallpox and syphilis. Although the health status of the AI/NA population has improved, they continue to suffer significant health disparities. For instance, AI/NAs suffer from much higher rates of infant mortality, sudden infant death syndrome, diabetes, heart disease, and tuberculosis than the general population (Indian Health Service 2007). Diabetes emerges as one of the most prevalent diseases that inflict the Native population. David Satcher (2000), the Surgeon General of the United States, noted that "the diabetes rate for American Indians and Alaska Natives is more than twice that for whites. The Pima of Arizona have one of the highest rates of diabetes in the world." Similarly, diabetes inflicts nearly 15% of all AI/NA individuals who seek treatment from Indian Health Services (United States Renal Data System 2003). Although recently in decline, tuberculosis remains a serious health issue as the Native population is twice as likely to be inflicted by tuberculosis as the general population. These high rates largely stem from reservation homes that are typically overcrowded with poor ventilation.

The AI/NA population also experiences higher rates of mental health problems than the overall population. Native youth suffer from a wide array of problems including anxiety disorders, pathological gambling, mood disorders, conduct and oppositional defiant disorders, disruptive behavior disorders, and separation anxiety disorders (Walker, Lambert, Silk-Walker, Kivlahan, Donovan, and Howard 1993; Duclos, Beals, Novins, Martin, Jewett, and Manson 1998; Costello, Farmer, Burns, and Erkanli 1997; Beals, Piasecki, Nelson, Jones, Keane, Dauphinais, Red Shirt, Sack, and Manson 1997; Novins, Beals, Roberts, and Manson 1996). Unfortunately, only one in seven will receive treatment (Costello et al., 1997). In general, many Native juveniles are hesitant to seek professional treatment because they feel that such services represent the "White man's" culture (Gone 2004) and they are embarrassed to ask for assistance (Freedenthal and Stiffman 2007). Structural conditions also impede their ability to receive adequate treatment.

The federal government established the Indian Health Service (IHS) in 1955 to provide healthcare to the Native population. Although the IHS was created to assist AI/NAs, the program has always been underfunded. As a result, many IHS facilities face numerous obstructions that impede their ability to adequately address the needs of AI/NAs including budgetary constraints, managerial shortcomings, limited personnel, and poor or nonexistent emergency care (Henson et al., 2007). Since the 1980s, the federal government has reduced funding even more for these health clinics. Recently, the Indian Affairs Congressional Committee reported that the IHS has become "completely dysfunctional" (Wallbank 2010). The report uncovered a litany of problems such as IHS employees with a "record of misconduct" who were being placed on paid administrative leave rather than being terminated, and employees who stole narcotics from facilities. Many of the IHS facilities were also in danger of losing their Medicare and Medicaid certification because of their poor conditions, which would be devastating to the vast majority of AI/NAs enrolled in these programs.

Suicide is a major health problem for AI/NAs who experience a suicide rate of 14.68 per 100.000, substantially higher than the national rate of 11.15, according to the Centers for Disease Control and Prevention (2011). Moreover, young males (20 - 24)years) have the highest rate of suicide among AI/NAs at 47.47 per 100,000; in fact, this rate is the highest for all groups in the United States. Among adults in the Northern Plain Indian tribe, 11% of males and 7% of females reported suicide attempts (Garroutte, Goldberg, Beals, Herrell, and Manson 2003). These high rates of attempted and completed suicides are also prevalent among Native youth. Suicide is so widespread among Native juveniles that it is the second leading cause of death for those 15 to 24 years (Centers for Disease Control & Prevention 2003) who experience suicide at a rate that is two times higher than their non-Native peers (Centers for Disease Control and Prevention 2005). Gender appears to influences suicide rates, with Native male juveniles experiencing a higher rate than their female counterparts (Indian Health Service 2008), while Native female juveniles have a higher rate of attempted suicide (Patel, Wallace, and Paulozzi 2005; Centers for Disease Control and Prevention 2007). Native juveniles living in urban areas are more likely to commit suicide than their reservation counterparts, perhaps because these youths are geographically removed from their ancestral communities. As Johnson and Tomren (1999:298) note:

Risk factors for American Indian suicide including stress, rapid economic and cultural change, acculturation and loss of culture, isolation and lack of strong identity are increased among American Indian youth in urban areas.

The high rates of suicide for all Native juveniles stem from several factors including problems with family and friends and their own emotional health (Borowsky, Resnick, Ireland, and Blum 1999). These factors may also influence alcohol and illegal drug use.

Alcohol and Substance Abuse

Substance abuse problems can be traced to the introduction of alcohol to the Native population by European colonists during the 1600s and 1700s. Trading posts served as an avenue for Natives to trade their goods in exchange for alcohol, specifically brandy and rum. The fur trade of the 1700s also provided an opportunity for Natives to obtain alcohol because European traders would barter alcohol for furs. As a result, the Native population has a long history of alcohol abuse that continues to this day and AI/NA adults and juveniles experience higher rates of alcohol binges (and illegal drug use) than the general population (Substance Abuse and Mental Health Services Administration 2010). Native juveniles are over two times more likely to use alcohol than other non-Native juveniles (National Institute on Drug Abuse 2003) and they begin to consume alcohol at an earlier age than their peers (Barnes et al., 2002). Native juveniles also consume illegal drugs.

Inhalants are frequently used by AI/NA individuals, especially among Native youth (Beauvais, Oetting, Edwards 1985; Okwumabua and Duryea 1987; Howard, Walker, Silk Walker, Cottler, and Compton 1999). Recent years have seen an increase in illegal drug use such as marijuana, cocaine, non-prescription drugs, and methamphetamine (National Congress of American Indians 2006). According to the National Survey on Drug Use and Health Report (2011), Native juveniles (12 to 17 years) have higher rates of marijuana use (13.8% vs. 6.9%) and non-prescribed prescription drug use (6.1% to 3.3%) than non-Native juveniles. Methamphetamine is also frequently used among AI/NAs. Over 70% of tribal police agencies reported that methamphetamine was highly prevalent in their reservation communities (United States Bureau of Indian Affairs 2006); the Navajo Nation reported a 100% increase in methamphetamine in recent years (Vigil 2006). This drug has devastated AI/NA families. Californian Indian Services reported that whenever a Native child is taken out of their home, either one or both parents are methamphetamine addicts (NCAI 2006). Alcohol and illegal drug consumption among AI/NAs have led to high rates of crime and recurring encounters with the criminal justice system (Perry 2004; Snipp 1992). The effects of colonialism are noticeably present in the AI/NA population generally. Poverty, substance abuse, and physical and mental health issues abound for the AI/NA population, especially those living in reservation communities. The following section delves further into the current conditions of many American Indians living in reservation communities ranging from crime to the 'lack' of 'justice.'

AMERICAN INDIAN CRIME AND VICTIMIZATION

Crime and victimization are among the most serious problems confronting American Indians and Native Alaskans (AI/NA). According to the Bureau of Justice Statistics, "the annual average violent crime rate among AI/NA was twice as high as that of blacks (50 per 1,000 persons), 2½ times higher than that for whites (41 per 1,000 persons), and 4½ times than for Asians (22 per 1,000 persons)" (Perry 2004:5). AI/NAs are more likely to experience violent crime victimization than any other group in the United States and one in ten AI/NAs are likely to experience a violent crime in their lifetime (Perry 2004). AI/NAs between the ages of 18 and 24 experience higher rates of violent victimization per capita than all age groups in the total population; one in four will experience a violent crime in their lifetime. Contrary to the general population, AI/NAs are more likely to be victims of inter-racial violence. In fact, almost 70% of violent crimes suffered by AI/NAs are committed by a non-Native offender (Perry 2004), a significantly higher level of inter-racial violence than that experienced by Blacks or Whites (Greenfeld and Smith 1999).

Homicide

۰.

Reservation communities experience higher levels of homicide than any other community in the US. The Ute Mountain Indian reservation's homicide rate is almost 50 times higher than the national average (Burnett 2006), prompting state officials to label the reservation community as "the murder capital of Colorado." While the homicide rate among AI/NAs has decreased in recent years, they continue to experience higher levels of lethal violence than the general population (Perry 2004) and their homicide rate is two times higher than the national average. These acts of lethal violence also occur among Native juveniles who most often fall victim to other AI/NAs (Wallace, Calhoun, Powell, O'Neil, and James 1996): "from 1976 and 1999, 7 in 10 American Indian juvenile murder victims were killed by another American Indian" (Perry 2004:iv).

Native Women Victimization

Scholars suggest that the effects of colonialism have contributed to the high rates of violence against Native women (Chester, Robin, Koss, Lopez, and Goldman 1994; Neithammer 1997; Smith 2005); such victimizations range from domestic violence to homicide. Irrespective of whether colonialism is a contributor for violence, Native women experience higher rates of violent victimization than any other female group in the U.S. population (Greenfield and Smith 1999; Perry 2004; Simoni, Sehgal, and Walters 2004; Magen and Wood 2006; Saylors and Daliparthy 2006; Yuan, Koss, Polacca, and Goldman 2006). In fact, violence against Native women surpasses the rate of African-American male victimization by 50% (Greenfeld and Smith 1999).

Native women experience high rates of rape and sexual assault (Bohn 2003; Harwell, Moore, and Spence 2003; Wahub and Olson 2004); in fact, Native women are two and one-half times more likely to be a victim of rape or a sexual assault than are other women (Perry 2004). Nearly 35% of all Native women will experience rape in their lifetime, significantly higher than the 20% of non-American Indian women who will suffer from a rape occurrence (Tjaden and Thoennes 2000). The majority of AI/NA victims (86%) are raped by a non-American Indian, in sharp contrast to the general population which is more likely to experience intra-racial rather than inter-racial sexual assault (Perry 2004). Still, these high numbers may mask the true extent of these victimization as the official data "underestimate[s] the true number of rapes committed each year, because [it] excludes rapes of children and adolescents, as well as... anyone living in... households without telephones" (Tjaden and Thoennes 2006: iii). The typical AI/NA sexual assault victim is poor, receives government assistance (Fairchild, Fairchild, and Stoner 1998; Malcoe, Duran, and Montgomery 2004), young (Fairchild et al., 1998), and in either a cohabitating relationship, separated, or divorced (Yuan et al., 2006).

Native women experience similar spousal abuse rates as the general population, but are more likely to be injured; 75% of these victims claim that their offender is a non-American Indian (Greenfeld and Smith 1999). Over 30% of Native women reported that they experience physical spousal abuse, compared to the national rate of 22% (Tjaden and Thoennes 2000). Community studies reveal the extent of the problem. Almost 50% of the San Carlos Apache Reservation residents reported that they experienced an intimate partner assault (Hamby and Skupien 1998). Similarly, 52.5% of women residing on the Navajo Indian Reservation claimed to experience at least one episode of intimate partner violence and 16.4% reported that their current partner physically abused them (Fairchild et al., 1998). In another study, researchers found that nearly 60% of American Indians in a southwest tribal community experienced physical or sexual violence (Malcoe et al., 2004). Unfortunately, high levels of violence and victimization also extend to Native juveniles.

Native Juvenile Delinquency & Victimization

Generally, AI/NA juveniles experiences higher rates of crime than the overall juvenile population (Greenfeld and Smith 1999; Perry 2004). Several reports have found that Native youths are responsible for committing a wide array of crimes that range from vandalism and vagrancy to aggravated assault and robbery (Greenfeld and Smith 1999; Wakeling, Jorgensen, Michaelson, and Begay 2001; Perry 2004; Arya and Rolnick 2008). More importantly, Native juveniles' violent crime rate is two and half times higher than the national average (Perry 2004). Self-report data also suggest that Native youth's crime and delinquency rates are much higher than AI/NA adults (Wakeling et al., 2001; Frank and Lester 2002). These crime statistics overlook Native youth living in non-reservation communities who are at even a greater risk of engaging in delinquency (Peak and Spencer 1987; Grossman, Krieger, Sugarman, and Forquera 1994). While ambiguity exists about the actual extent of Native youth delinquency (Zatz, Chiago, and Snyder-Joy 1991; Armstrong, Guilfoyle, and Melton 1992), official and reservation data paint a grim situation for these young individuals. For instance, Warm Spring Reservation in Oregon experiences a delinquency rate that is nearly six times higher than the national average (O'Brien 1977).

Native youth also commit high rates of alcohol and illegal drug violations (Beauvais and LeBoueff 1985; Beauvais, Oetting, and Edwards 1985; Beauvais, Oetting, Wolf, and Edwards 1989; Snyder and Sickmund 2006). For instance, Native youth commit offenses such as liquor law violations at rates almost three times higher than do other non- Native youth groups (Snyder and Sickmund 2006). Over 30% of the arrests of Native youth during an eleven year period (1988-1998) were for alcohol violations (consumption and possession) (Bond-Maupin, GoodTracks, and Maupin 2006). Peak and Spencer (1987:401) examined crime and delinquency rates in over 200 reservation communities and found "69 percent of the actual offenses that were investigated in 1982 involved the use of alcohol or drugs."

The sexual abuse of Native youth is also pervasive, although the true extent is difficult to estimate (Malley-Morrison and Hines 2004). Nationally, about 14% of Native female and 2.4% male young individuals reported that they had experienced sexual abuse

(Pharris, Resnick, and Blum 1997). In Alaska, nearly half (45%) of all sexual abuse victims are Alaskan Native, although Alaskan Natives comprise just shy of 17% of the state's total population (Vernon and Bubar 2001). Retrospective studies paint a grim picture about child sexual abuse. For example, 56% of American Indian women in Montana reported to have been victims of child sexual abuse (Hobfoll, Jackson, Hobfoll, Pierce, and Young 2002). A study of child sexual abuse occurring on Indian reservations in New Mexico found that the majority of the victims were female juveniles (77%) between the ages of six and eleven, and the perpetrator in these crimes ranged from extended family member (43%) to parent/stepparent (22%) (Steele, Damon, and Denman 2004). Despite these high rates of child abuse among American Indians, the offenders are seldom brought to justice. For example, from 2003 to 2006, the Federal Bureau of Investigation (FBI) responded to 1,658 child abuse cases that occurred within tribal communities but only made 537 arrests for these crimes (Burrus 2006). Realizing the lack of federal assistance, many tribal communities have established their own policing units.

POLICING AMERICAN INDIANS

The interaction between American Indians and law enforcement remains a difficult challenge in Indian Country. While policing is a universal concept in mainstream society, it is unfamiliar for many American Indians (Luna 1998; Perry 2009). As Luna-Firebaugh and Walker (2006:118) contend:

The concept of professionalized policing, with its emphasis on technology, specialized police activities, and restricted use of police discretion, does not conform with the style of peacekeeping most commonly used in traditional or rural communities.

To understand the conflict between American Indians and law enforcement, it is important to examine the historical development of policing of Native peoples in the United States, which can be traced to colonialism. For the Anglo population, policing American Indians was essential in securing tribal lands. As Perry (2009:35) notes:

Law enforcement bodies – in varied guises – supported and often facilitated the colonialist project of the state. The social and legal order defended by the police was also that of a colonial state, seeking to exclude or at least regulate native peoples who were in the way of "progress." Specifically, law enforcement became the key means by which to "police" racialized spaces, and by which to facilitate the assimilative process.

Initially, American Indians were policed by the U.S. military and were subjected to a myriad of abuses which were attempts "civilize" them (Peak 1989; Stannard 1992; Million 2000; Perry 2009). As years passed, policing units replaced the military's role of being an "agent of civilization" (Harring 1994). Again, as Perry (2009:39) points out:

Where the military had served its purpose in restricting the Native American land base, the police would serve to similarly restrict their cultural base. Increasingly, police became a key authority in the project of assimilation.

The mission of tribal police was twofold: (1) segregate American Indians to reservation communities and (2) ensure they comply with assimilation policies (Peak 1989; Stannard 1992; Perry 2009).

In 1869, Thomas Lightfoot, Indian Agent to the Sac and Fox Tribes proclaimed, "I have appointed a police, whose duty it is to report to me if they know of anything that is wrong" (United States Indian Office 1869: 356). Thus, Lightfoot is credited for establishing the first formal law enforcement organization in Indian Country (Peak 1989; Perry 2009). This event marked a shift in policing American Indians. Initially, the colonial powers treated American Indian tribes as sovereign nations. During the late 1800s, the federal government altered its approach and began to view American Indians as individuals rather than sovereign nations. As a result, American Indians lost greater autonomy with respects to administering justice among themselves (Peak 1989; Stannard 1992). In fact, "Native sovereignty would by turns be contracted and expended over time, with the net result of a significant loss of autonomy in decision making and control with respect to issues of crime" (Perry 2009:35). In years to come, the federal government (as in all American Indian matters) adopted a paternalistic ideology in addressing crime and violence in Indian Country (Skoog 1996). Public concern about safety in U.S. western territories heightened when Commissioner of Indian Affairs Ezra Hayt testified before Congress that (Peak 1989:396):

Civilization has loosened...evil continues unabated. Women are beaten and outraged, men are murdered in cold bold, children are molested on their way to school. It is a disgrace to out land...as long as by the absence of law Indian society is left without a base.

These imageries suggested that tribal lands experienced high rates of crime and violence. Following Lightfoot's lead, the federal government allocated funding to establish several policing organizations in Indian Country in 1878 and by 1880 there were over 480 police officers patrolling tribal lands (Peak 1989; Million 2000). However, it was the Crow Dog case that caused the government to create a more efficient and organized policing presence in Indian Country (Luna-Firebaugh and Walker 2006).

In 1881, Crow Dog, a Lakota Sioux, murdered principal chief Spotted Tail, also a Lakota Indian on the Rosebud Indian Reservation. Crow Dog was tried by his tribe and ordered to pay "Blood Money" (a form of restitution) to the victim's family. The federal government believed that this form of punishment was not satisfactory and arrested, tried for murder, and sentenced Crow Dog to death in Dakota Territorial court in Deadwood, South Dakota (Luna-Firebaugh and Walker 2006). Crow Dog appealed his conviction to the Supreme Court on the grounds that the federal court system did not have the right to try him again for the crime because the offense was already settled by the Lakota tribal council. The Supreme Court sided with Crow Dog and stated that the federal court did not have jurisdiction to hear the case unless it was authorized by Congress (Peak 1989; Million 2000). The Supreme Court's decision led to a public outcry that an American Indian was able to escape a "suitable" punishment for murder (Luna-Firebaugh and Walker 2006). As a result, Congress passed the Major Crimes Act of 1885, which granted the federal government more jurisdictional powers to try serious crimes that occur on tribal lands. The Act also called on the federal government to allocate resources to establish a more professional law enforcement presence in Indian Country. Within decades of the Act's passage, over 100 law enforcement agencies provided assistance to reservation communities (Peak 1989).

The federal government later reduced funding and resources to adequately address crime and victimization within Indian Country. By the 1950s, frightened individuals voiced concerns about the lack of police officers in these areas (Peak 1989). To address these shortcomings, Congress increased funding for equipment and resources as well as hired more officers in tribal communities and authorized the establishment of a U.S. Indian Police Training and Research Center to properly train tribal police officers (Million 2000). During the 1970s, Congress passed Senate Bill 1228 which federally legitimized the arresting power of tribal police officers. Despite receiving recognition and funding, current tribal police officers struggle to protect American Indians living in Indian Country.

Current State of Tribal Law Enforcement

Tribal police officers perform various duties; "the most common functions [are] executing arrest warrants (95%), enforcing protection orders (92%), serving process (89%), apprehending fugitives (88%), and providing court security (75%)" (Reaves 2008:3). In the midst of fulfilling policing obligations, many tribal police departments are understaffed and over-stretched in Indian Country. According to the Department of Justice, in 2000 there were 3,462 tribal police officers in Indian Country (Reaves 2008); the number of tribal police officers dropped to 3,000 in 2012. These figures suggest that tribal police departments have lost a substantial number of officers in a ten year time span. Unfortunately, tribal police departments have consistently experienced problems in maintaining police officers. High turnover rates within tribal police departments are persistent in Native Alaskan Villages. For instance, "for every one hundred [Native Alaskan Village police officers] to have served in a calendar year, thirty-six either quit or were fired" (Wood 2002:197). During the 1970s, these communities experienced such an extreme high police turnover rate that it was suggested that "entire village populations" may eventually be trained as village police officers in the continuing effort to keep trained officers in each village" (Angell 1978: 67). Explanations for the high rate of attrition included stress, poor wages, dissatisfaction with police training, and lack of support among reservation community members (Wood 2002).

Tribal Police Department Problems

Reservation communities are situated in large rural areas, where many tribal members can be reached "by unpaved and unmarked roads" (Luna-Firebaugh and Walker 2006:119), which proves to be difficult for the limited number of tribal officers to adequately patrol these reservation communities. Recent statistics suggest that "11 of the 25 largest tribal law enforcement agencies serve jurisdictions covering more than 1,000 square miles" (Reaves 2008:2). The federal government is fully aware of this predicament. During a Senate committee hearing on Tribal Law Enforcement, Tracy Toulou, the Director of the Office of Tribal Justice, stated that "the typical [tribal police] department serves an area the size of Delaware, but with a population of only 10,000, that is patrolled by no more than three police officers and as few as one officer at any one time" (United State Senate Committee on Indian Affairs 2002). For example, the Fort Apache Indian Reservation in Arizona has a population of 12,500 individuals and a land area consisting of 2,600 acres, but its police department consists of only 30 officers, which proves difficult to meet the needs of tribal members (United State Senate Committee on Indian Affairs 2002). The Pine Ridge Indian Reservation in South Dakota also experiences understaffing problems. As Williams (2012a:1) notes:

Despite growth in both population and crime [in the Pine Ridge Indian Reservation], the number [of tribal police officers] had fallen to 49. F.B.I. agents have told officers on Pine Ridge that the reservation needs at least 140 officers to handle an epidemic of violence that includes 3,000 child abuse cases and more than 20,000 arrests each year — nearly one arrest for every other resident.

Given these realities, many individuals (including government officials) suggest that the lack of tribal police officers allows opportunities for crime and violence to occur in Indian Country. Similarly, Senator Al Franken (2010:3) in a Senate Committee Hearing on Tribal Police asserted:

Crime on Indian reservations is an enormous problem in so many states, certainly in Minnesota. For example, on the Red Lake Reservation, which I recently visited, the crime rate is eight times higher than the national rate. There are so many different reasons for this, but one is the simple lack of police officers at Red Lake and in Indian Country. To obtain minimal officer safety standards as determined by the BIA, Red Lake would need a minimum of 32 officers. However, currently they have 15, less than half of what they need.

American Indians in some reservation communities rely on state and local police departments for assistance, but these services are extremely limited. In a recent study, American Indians had unfavorable view about these law enforcement agencies (Goldberg and Champagne 2007). Many of these individuals cited slow response times and lack of police responses. Tribal police departments also lack the funds needed to hire officers and acquire equipment.

Tribal Police Budgets

The federal government has reduced the size of the police force in Indian Country by decreasing funds for law enforcement "even as rates of murder and rape there have increased to more than 20 times the national average" (Williams 2012b:1). Although President Obama referred to violence in Indian Country as "an affront to our shared humanity" (Williams 2012b:1), his administration has reduced the budget for the BIA and funding for tribal law enforcement. In fact, "the federal government allocates far less money for public safety on Indian lands than what cities of similar size devote to fighting crime" (Williams 2012b:1). While the BIA oversees the safety for over 1.6 million American Indians living in over 56 million acres in Indian Country, the agency has a \$322 million budget for public safety in the 2012 FY (Williams 2012b). By comparison, the city of Philadelphia with a similar population of 1.5 million and covering 142.6 square miles of land has a \$540 million budget to support a police department of 6,400 officers. Given the impoverished conditions within Indian Country, many tribal police departments lack the resources necessary to obtain policing equipment or technology to address crime and violence (Wakeling et al., 2001). Some tribal police departments rely on older equipment, while others do not have working patrol vehicles or access to police radios. Some communities even lack 911 response systems which makes it difficult for individuals to call for assistance (Luna-Firebaugh and Walker 2006). As a result of the lack of resources and manpower, many tribal police departments are unable to properly record all crimes that occur within reservation communities. As Wakeling et al (2001:14) note:

Because the small size of police departments in Indian Country places time constraints on staff, data collection is extremely expensive relative to the overall budget of the department. Police officers and other staff commented that the time needed for specialized administrative tasks competes with, and often loses out to, fighting crime and providing emergency services. When police officers also must serve as jail staff, as is the case in approximately 50 percent of all departments surveyed, their time is stretched even further. Simply put, because these small departments cannot afford to dedicate staff to any specialized function, the cost of data.

Several tribal police departments document crime and arrests through handwritten records, many of which are seldom consistent or properly maintained making it difficult to understand the actual extent of crime. Several organizations admit that the lack of resources hinders the ability of tribal police departments to adequately address crime. For instance, the International Association of Chiefs of Police (2001:12) note:

In other words, resource limitations place direct constraints on the ability of law enforcement and justice agencies to protect the safety of Indian country residents and to prevent crime and victimization. Increased monetary resources – and the translation of these resources into manpower, training, facilities, equipment, program development, research and evaluation, and community outreach – are critical to improved safety in Indian country. In addition to a lack of resources, tribal police departments experience a strained relationship with reservation community members.

Tribal Police and Community Relations

Historically, American Indians distrust tribal police officers. For many American Indians, the tribal police represent the state whose purpose was "to force the imposition of Western values and behaviors, and the suppression of traditional ways of living" (Perry 2009: 40). As agents of the state, tribal police officers enforced anti-Native laws that prohibited American Indians from practicing their religion, customs, and using their Native languages. As Perry (2009: 40) notes:

The Americans were at least intelligent enough to know that in religion were to be found the foundations for a host of other societal norms. In the case of Native Americans, these norms were held to be a dangerous contrast to Western values, and thus a brake on assimilation.

Charged with the task of enforcing anti-Native laws, tribal police were responsible for ensuring that Indian cultural practices such as the ghost dance and smoking peyote were not celebrated among American Indians (Peak 1989; Skoog 1996). Ironically, "where many immigrant and ethnic groups have found their traditional life patterns despised and ignored in the United States, Indians are unique in that their traditions have been criminalized" (Bracey 2006:49). Luna-Firebaugh and Walker (2006) assert that the killing of Sitting Bull, a Sioux Medicine Man, continues to be a source of resentment towards tribal police officers. In 1890, Sitting Bull voluntarily surrendered himself to federal authorities for practicing in the Ghost Dance ritual and was later killed by Indian police officers during his imprisonment in an Indian jail. The federal government quickly painted Sitting Bull as a "troublemaker" and asserted that the tribal police fulfilled their obligation in maintaining law and order in Indian Country. In the aftermath of the killing, Massachusetts Senator Henry Dawes declared that tribal police are "the bulwark of the Government in the administration of justice and in the preservation of order on the reservation" (Hagan 1966:159). For many American Indians, Sitting Bull's death symbolizes "bad policing" in reservation communities (Luna-Firebaugh and Walker 2006). Anti-police sentiments continue into the present as many American Indians are unwilling to cooperate or provide information about criminal activity occurring within reservation communities. In extreme cases, tribal police officers "themselves become victims of violence crime in retaliation for their role" (Perry 2009:41). Some American Indians viewed them as:

Servants of the Great Father in Washington...Indian police officers not only gave up their braids and moccasins, but also campaigned against aspects of their own culture including ceremonial dancing, raiding for horses, and the influences of witches (Knepper and Pucket 1995:15).

American Indians have also experienced police brutality and racial profiling in Indian Country. Perry (2009) found that American Indians living in or around reservation communities are subjected to over and under-policing which often perpetuates criminalization and victimization of these individuals. American Indians reported that tribal police officers neglected to respond to the needs of victims. In other cases, the participants reported that the tribal officers harassed and physically assaulted them. These experiences may be part of tribal police departments' recruitment problems.

Tribal Police Recruitment

The federal government actively recruits American Indians to fulfill positions within tribal police departments, however in the past, this recruitment served as a means to control American Indians. As Perry (2009:41) notes:

It was intended to reinforce the imposition of western jurisprudence over Native American values and strategies of social control, that is, to render social control more effective through the employment of "insiders" who knew intimately both the people and their life styles.

Thus, Native police officers were used to legitimize the law.

Despite animosity towards tribal police officers, many American Indian males believe that becoming an officer will fulfill Native cultural responsibilities. Washburn (1971:171) found that Native males enlist in the tribal police department to reclaim their "warrior status." While Indian males believe that they are regaining a cultural status, Indian females have also enjoyed promotional success within these police departments. Luna-Firebaugh (2002) found that tribal police departments had a greater representation of females than non-tribal police departments and that females were five times more likely to be elevated to command and supervisor positions in tribal police departments than non-tribal police departments. In addition, females within tribal police departments were less likely to experience gender-based hostility in their occupation. Despite positive advancements among females in tribal police departments, these institutions continue to experience hardships that impede their ability to provide assistance to reservation community members.

DiGregory and Manuel (1997) suggest that "there is a public safety crisis in Indian Country" pointing to several problems confronting tribal law enforcement such as being fragmented, lack of resources, and poorly coordinated in responding to crime. Despite these weaknesses, tribal police departments are in position to meet the needs of the American Indians. As Luna-Firebaugh and Walker (2006: 133) suggest:

They are close to their communities and generally reflect the diversity of those communities and are in the early stages of development...[and] bad habits and tolerance of misconduct have not yet had the opportunity to become deeply rooted in the departments.

Tribal police departments continue to provide security to American Indians living in reservation communities despite confronting many obstacles.

TRIBAL JUDICIAL SYSTEM

The federal government recognizes tribal governments as autonomous entities capable of "[making] their own laws and to be ruled by them" (Williams v. Lee, 358 U.S. 217, 1958). As a result, many tribal governments have implemented their own tribal justice system. Currently, there are 291 tribal courts in the United States (Saunooke 2008), staffed by over 500 tribal judges (American Bar Association 2001). Although the federal government acknowledges tribal governments as sovereign nations, they have severely limited the power of the tribal judicial system. While the federal government has unlimited jurisdiction to prosecute crimes occurring in Indian Country, tribal governments are limited in their jurisdictional powers. In *Duro v. Reina*, the Supreme Court ruled that tribal governments do not have the authority to prosecute the criminal conduct of non-AI/NAs. Thus, tribal courts are unable to hear cases involving non-AI/NAs, even if the victim is an AI/NA. In addition, although tribal governments are able to prosecute AI/NAs for crimes, this is limited as tribal courts are restricted to hearing misdemeanor cases. The federal government bears sole responsibility for prosecuting felony crimes occurring in Indian Country but fails to do so nearly 75% of the time (Burrus 2006). The tribal government may prosecute those cases declined by the federal government as long as the perpetrator is an AI/NA, but the crime, regardless of its seriousness will be tried as a misdemeanor offense.

To address the shortcomings of the tribal criminal justice system, President Obama signed into law the Tribal Law and Order Act of 2010 which holds federal agencies responsible for investigating and prosecuting crimes in Indian Country. The Act includes several provisions to provide and maintain security in Indian Country such as creating and maintaining a database of crimes occurring on tribal lands, requiring federal prosecutors to share evidence for prosecution in the tribal judicial system, and mandating that federal agents testify in tribal courts. More importantly, the Act increased the sentencing powers of the tribal court system to include the ability to impose the maximum sentence of three year imprisonment and fines up to \$15,000. Although many tribal courts have adopted Anglo-American philosophies of justice, others have implemented traditional AI/NA methods to settle disputes.

CONCLUSION

The AI/NA population experiences higher rates of crime and victimization than any other group in the United States yet are unable to receive adequate resources or social services address their victimization. In addition to inadequate treatment, many AI/NA victims of crime often find that justice is elusive due to the complexities of the tribal court system as it operates under the confines of the federal government. While these problems are well known, tribal police officers are able to provide an additional discourse about crime and justice in Indian Country. The next chapter provides an overview on the criminological literature on police perceptions.

CHAPTER III

THE LITERATURE ON POLICE PERCEPTIONS

From directing traffic to answering calls for assistance, the police have a ubiquitous presence in society. Empowered by the state, they are required to enforce the law, protect state and individual properties, and maintain civil obedience (Barak, Leighton, and Flavin 2007). With that being said, the police have become symbols of law, justice, order, and stability (Taylor 1999; Waddington 1999; Reiner 2000; Loader and Mulcahy 2003). As Jackson and Brafford (2009:497) note:

More prosaically the police present a highly visible instantiation of state power, with which it is nonetheless possible to interact on a face-to-face basis -a position which contrasts with that of many other representations of the state.

Some argue that the main focus of the police is crime prevention, which has been a popular topic of criminological scholarship for the past 50 years. In fact, scholars have investigated the effectiveness of preventive patrol (Kelling, Pate, Dieckman, and Brown 1974; Sherman 1990, 1995; Bayley 1994; Sherman and Weisburd 1995; Marvell and Moody 1996), rapid responses (Isaacs 1967; Clawson and Chang 1977; Klockars and Mastrofski, 1991; Bayley 1994; Sherman 1995), criminal investigations (Chaiken, Crabill, Holliday, Jaquette, Lawless, and Quade 1977; Greenwood, Chaiken, and Petersilia 1977; Bayley 1994), and reactive policing (Tittle and Rowe 1974; Logan 1975; Brown 1984; Sherman and Berk 1984; Greenberg, Kessler, and Logan 1979; Chamlin 1988, 1991). Despite an overwhelming concentration on crime prevention strategies, some criminologists argue that the police are symbolic representations of power and authority (Loader 1997). The police wield considerable power in shaping feelings and

thoughts on crime and everyday life. This chapter provides a critical overview of policing in the United States, with special attention to research on police perceptions on crime. In addition, the chapter discusses the framework of Loader's (1997) symbolic power of policing as it relates to the current dissertation research.

THE POLICE

The police have multiple responsibilities. Michalowski (1985:182) recognizes these as "dramatizing state power, legitimizing state power, maintaining order, controlling crime, and labeling criminality." Above all else, the police are agents of the state. As Michalowski (1985:170) suggests, "policing is unique to state societies. In...non-state societies...rules and customs are enforced and order maintained by ordinary individuals in the process of conducting their daily lives, not by designated representatives of state power." State societies establish police forces to maintain the interests of the powerful and protect property rights. In essence, "policing arises when one group develops an interest in controlling other groups" (Lynch and Michalowski 2006: 218-219). While the concept of policing can be traced to ancient times, the police initiated during the industrial revolution and the rise of capitalism (Chambliss 1964).

In the United States, contemporary policing evolved during the 1960s, an era of great social and political upheaval generated from the civil rights movement, women's rights movement, the Vietnam War, human rights violations, and the questioning of power in Washington D.C. (Bernard et al., 2010; Williams and McShane 2010). Lynch and Michalowski (2006) recognize three factors that contributed to the development of modern policing: (1) social turmoil on college campuses and inner city communities, (2)

the advent of broadcast television which emerged as a primary news source for many Americans, and the most important, (3) a concerted conservative effort to fracture the political capital held by the Democratic Party. Following WWII, the Democratic Party embarked on a progressive political agenda to assist minorities, women, and the poor in achieving social equality (i.e. education, employment, and standards of living). Despite positive improvements, minority groups continued to be suppressed and took to the streets to vent their frustration in the form of marches and protests. In addition, students on college campuses were protesting against the Vietnam Conflict. Nightly, Americans were confronted with these images of disturbance on their televisions which portrayed an America "coming apart" (Lynch and Michalowski 2006:215). Conservatives used these social upheavals as an opportunity to regain political power (see Garland 1990, 2001; Tonry 1994). To achieve political dominance, conservatives demonized minority groups as criminals and argued for a return to law and order, which required a transformation of the criminal justice system and American policing (Lynch and Michalowski 2006).

Tonry (1994) agrees that the power of conservative politics weighs heavily on the criminal justice system. He argues (1994:496) that "conservative Republican politicians have, since the late 1960s, used welfare, especially Aid to Families with Dependent Children, and crime as symbolic issues to appeal to anti-Black sentiments and resentments of White voters." Tonry recognizes two developments that led to stringent policing policies and mass incarceration. Nixon and fellow Republicans ran on an anti-crime platform and used the "race card" to create an anti-Black sentiment to appeal to whites and attract them to the Republican Party, and both Democrat and Republican policies.

these factors "ultimately resulted in the transformation of urban police forces into militarized strike forces organized and trained according to national policies and national standards" (Lynch and Michalowski 2006:215). The social movements of the 1960s were seen as a threat to those with power and wealth. Moreover, Senator Samuel Huntington told Americans that there was "an excess of democracy and that what was needed was a greater degree of moderation in democracy" (Lynch and Michalowski 2006:216). Scholars argue that the criminal justice system was transformed to operates as a system of stratification to ensure that certain groups were regulated to the lower fringes of society.

Massey (2007) asserts that criminal justice system is a vehicle that maintains control over minorities. For Massey, the criminal justice system embodies two mechanisms that enforce and reproduce stratification to the benefit white males: opportunity hoarding and exploitation. Massey (2007:244) refers to opportunity hoarding as "the monopolization by in-group members of access to a resource so as to keep it for themselves." This mechanism allows the in-group (whites) to dominate resources such as social services and housing opportunities so that they are not in competition with minority groups and the poor. Exploitation, "is the expropriation of resources from an out-group by members of an in-group such that out-group members receive less than full value for the resources they give" Massey (2007:244). This mechanism permits whites to exploit the out-group (minorities) in terms of resources such as housing, employment, and money. The criminal justice system serves as a systematic mechanism of inequality, because it subjects members of the out-group to extraordinary policing that results in exclusion from opportunities to achieve resources; those resources then ultimately go to whites.

Historically in the U.S., the police have been used to maintain "unequal class structures by focusing on behaviors most likely to be engaged in by working-class and marginalized populations" (Lynch and Michalowski 2006: 222). During the 19th century. the police were used to suppress worker strikes who were protesting against economic and political exploitation (Harring 1981, 1983). Harring (1976:56) argues that "without real economic and political exploitation there would have been no riots, nor any need to create a police department to control them." That is, the police were created to maintain the social, political, and economic dominance of the powerful (Lynch and Michalowski 2006). In a similar vein, Gabbidon and Greene (2005) assert that antebellum slave patrols in the south serve as a precursor to the modern form of policing. These patrols were responsible for enforcing slave codes, apprehending runaway slaves, and providing punishment towards Blacks caught violating the law. Presently, many Blacks (similar to their ancestors) find themselves subjected to extraordinary policing. As Lynch and Michalowski (2006: 223) note, "police use of public-order arrest to control the lower classes has not disappeared in the modern world." Although research has provided a deep historical context on the use of the police in society, studies have also sought to understand police perceptions of crime, violence, and justice.

POLICE PERCEPTIONS

Police officers have a distinct view on crime, violence, and justice. As Fielding and Fielding (1991:39) note:

Hardness and pragmatism mark the stereotype of police attitudes on crime and punishment; for many, police are the epitome of the 'law and order' mentality. Police are seen to subscribe to a harsh, narrow, and unforgiving view of human nature and to do so consistently. These attributes contribute to the "police mind" (Colman and Gorman 1982), which often shapes the realities of crime and justice among police officers. Scholars have underscored the significance of examining the perceptions of the police. Fielding and Fielding (1991:40) assert that police perceptions can uncover their "working codes and conceptions of human nature." Indeed, the examination of police perceptions can provide an insight about the nature of policing and their response to crime. The inquiry of these perceptions has led to several studies examining police perceptions on different types of crime such as gangs (Quinn and Downs 1993), domestic violence (Friday, Metzgar, and Walters 1991; DeJong, Burgess-Proctor, and Elis 2008; Schafer and Giblin 2010), Stalking (Weller, Hope, and Sheridan 2013), intimate partner violence (Gracia, García, and Lila 2011), sexual assault (Campbell and Johnson 1997; Davies, Smith, and Rogers 2009; Wentz and Archbold 2012), computer crimes (Holt and Bossler 2012), and drug crimes (Gardiner 2012). Before exploring police perceptions of crime, it is important to understand the formation of these perceptions.

The Formation of Perceptions

Ikner, Ahmad, and del Carmen (2005:83) suggest that "police officers, like most people, are socialized in both personal and professional settings, which affects their decision-making." Indeed, the police are "hard-wired" to detect certain traits or characteristics that are perceived as criminal. However, some of these perceptions are based on prejudices and stereotypes. As Massey (2007:10) notes, "all human beings, whether they think of themselves as prejudiced or not, hold in their heads schemas that classify people into categories based on age, gender, race, and ethnicity." Several studies have examined the psychological processes that contribute to prejudices of individual groups (Taylor, Fiske, Etcoff, and Ruderman 1978; Stangor, Lynch, Duan, and Glas, 1992; Bargh 1996; Fiske 2004). Again as Massey (2007:9) asserts:

Since human memory is finite and cannot be expanded, if the brain is to remember more things it must combine or "chunk" bits of information into larger conceptual categories (schemas), using common properties to classify a much larger number of people, objects, and experiences into a small number of readily identifiable categories for recall.

With respects to police officers, they rely on these prejudices and stereotypes to make judgments. That is, their perceptions of crime and justice may reflect their own biases, prejudices, and stereotypes. Aside from prejudices and stereotypes, other factors are influential in the shaping of police perceptions.

The police sub-culture greatly impacts the police perceptions. Westley (1970:198)

defines police sub-culture as:

The particular definitions of the public and of his conduct that the policeman develops out of this interaction assume a collective and cultural character because the police hold the experience in common, because of the strong consensual bond developed by the felt hostility of the public.

Research has examined the dynamics of the police subculture (Skolnick 1966; Westley

1970; Savitz 1971; Brown 1988). This culture is responsible for the transmission of a

distinct set of cultural norms among police officers. That is, it serves as a socialization

mechanism which provides individuals with "organizational cues and boundaries" (Britz

1997:128). More importantly, newly appointed police officers are "groomed" to think

and perceived differently than others. As Britz (1997:129) suggests:

Across organizations, the osmosis of norms, ideals, and perceptions of organizational incumbents is essential to a newcomer's occupational longevity. Newcomers to all organizations must adhere to established modes of behavior or risk ostracism from the group. Indeed, the police sub-culture plays a significant role in shaping their perceptions and realities about human nature. Several studies have examined the impact of the police sub-culture in the formation of police attitudes and behaviors (McNamara 1967; Westley 1970; Rubinstein 1973; Van Maanen 1974; Sparrow, Moore, and Kennedy 1990; Skolnick 1994; Manning 1995; Kappeler, Sluder, and Alpert 1998). More importantly, these police values and beliefs often contribute to "distrust and suspiciousness of citizens and a prescription to assess people and situations in terms of their potential threat" (Paoline 2004:207) and creates the "us vs. them" mentality among police officers. Other than the police sub-culture, individual differences such as race/ethnicity, age, gender and past experiences also impact police perceptions (Britz 1997). Education level also effects police perceptions on crime-related matters.

Research has examined the role of education on police perceptions (Smith, Locke, and Walker 1968; Smith, Locke, and Fenster 1970; Weiner 1976; Tyre and Braunstein 1992; Truxillo, Bennett, and Collins 1998). An earlier study found that college educated police officers were significantly more likely to understand social problems and shown to be more sympathetic towards minority groups (Weiner 1976). In a similar vein, educated police officers appeared to be more professional and less likely to engage in unethical behaviors (i.e. police brutality and corruption) (Tyre and Braunstein 1992; Truxillo et al 1998). Educated police officers appeared to be less authoritative than those with limited college education (Smith et al 1968; Smith et al 1970). More importantly, college educated police officers have been shown to adjust to the dynamics of society. That is, police officers with "college degrees had the most open belief systems and highest levels of job performance, indicating that college educated officers were better able to adapt to the complex nature of the police role" (Roberg 1978:344). Other research suggests that environmental factors may also impact police perceptions.

Environmental Factors

Environmental factors can greatly impact perceptions of human behavior, more specifically in urbanized areas. As Massey (2007:7) note:

Human interactions increasingly occur within urban environments of great size, density, and heterogeneity, and the ecological settings that individuals find themselves adapting to – psychologically, socially, culturally, and physiologically – vary greatly depending on whether the individuals are rich or poor, light or dark, male or female.

Indeed, the characteristics of an area can greatly impact police perceptions and attitudes. Sun and Triplett contend that criminological research on police perceptions of neighborhood crime is limited. However, the few studies available do suggest that the structural characteristics of a neighborhood may impact police perceptions on crime. For example, Hagan, Gillis, and Chan (1978) found that a neighborhood's socio-economic status effects police perceptions on crime and deviance. They found that poor neighborhoods were seen by the police as "offensible space" and perceived to be more prone to crime and violence. As a result, the police are more likely to aggressive force (Terrill and Reisig 2003) and make more arrests (Smith, Visher, and Davidson 1984) in these perceived "offensible spaces." Moving beyond space and location, research has also found that the characteristics of the individual such as race and gender are factors on police perceptions. Race

Gabbidon and Greene (2012:1) note that "the notion of race seems to permeate almost every facet of American life." Unfortunately, individuals often rely on "racebased inferences" (Cosmides, Tooby, and Kurzban 2003:173) or stereotypes in the formation of their perceptions of human behavior (Massey 2007). Research has found that the police rely on the race of the individual in their decision making which contributes to the use of racial profiling, excessive force, and targeting racial/ethnic communities (Cohen 1996; Barlow and Hickman-Barlow 2000; Barlow and Hickman-Barlow 2002; Dowler 2005; Ioimo, Tears, Meadows, Becton, and Charles 2007). In an earlier study, Skolnick (1966) suggested that the police frequently target Black males because they are perceived as the "symbolic assailant." According to Skolnick, the police are trained to be vigilant and suspicious of individuals by their appearance (i.e. clothing, race, language, and body language). As Skolnick and Gray (1975:45) note:

The policeman, because his work requires him to be occupied continually with potential violence, develops a perceptual shorthand to identify certain kinds of people as symbolic assailants, that is, as persons who use gesture, language, and attire that the police man has come to recognize as a prelude to violence. This does not mean that violence by the symbolic assailant is necessarily predictable. On the contrary, the policeman responds to the vague indication of danger suggested by appearance.

For Skolnick, police officers are trained to detect the attributes and characteristics of the "symbolic assailant." Research has also shown that the racial composition of a neighborhood greatly impacts police perceptions (Smith 1986). One study found that police officers are more likely to be more abusive and coercive towards criminal suspects in areas with high rates of heterogeneity (Smith 1986; Sung 2002). Given the recent war on terrorism, the police have been trained to detect "terrorists" by targeting individuals

who appear to be Arabic or Muslim (Onwudiwe 2005; Keeling and Hughes 2011; Gabbidon and Greene 2012). That is, anyone who appears to be Arabic or Muslim is perceived to be a terrorist and a threat to national security. That is, race is an important variable in police perceptions, research also found that gender significantly impact these views.

Gender

Perceptions of masculinity and femininity are established by dominant societal groups (Fortier 1975) which "serve male interests" (Page 2008:399). For example, characteristics of masculinity are aggressiveness, intelligence, and virility; however, femininity traits are the opposite and often "reflect subjugation, docility, passivity, and naiveté" (Page 2008:399 also see Johnson 1997). Given the patriarchal emphasis in the United States, individuals are socialized to accept these gendered traits. Sociologists turned to the concept of hegemonic masculinity to explain the social position of males and the subordination of females (Connell 1987, 1995, 2002).

[Hegemonic masculinity] is form of masculinity, emphasizing dominance, aggression, heterosexuality, and a lack of emotion. Hegemonic masculinity promotes but does not determine behavior that supports the traditional gender order... In other words, hegemonic masculinity acts as a script for how men should behave and or think in various situations (Page 2008:404).

Scholars argue that the police project hegemonic masculinity (Martin 1989; McNamara 1999). Fielding (1994:47) argues that the policing sub-culture "may be read as an almost pure form of hegemonic masculinity." As result, police officers embrace heterosexist attitudes which often leads to "patriarchal and misogynistic attitudes toward women" (Fielding 1994:47). These attitudes often lead to negative police perceptions about

violence against women (Page 2007). Others contend that it may be situational factors that influence police perceptions of violence against women, more specifically domestic violence (Rigakos 1997; Sinden and Stephens 1999; Robinson and Chandek 2000). For example, police responses to these crimes often rely on their perceptions of the crime scene and other factors such as neighborhood characteristics, visible injuries, previous calls for assistance, illegal drug and alcohol use, and witnessing the crime (Belknap 1995; Rigakos, 1997; Buzawa and Buzawa 2003). DeJong, Burgess-Proctor, Elis (2008:692) found that "some officers expressed attitudes demonstrating negative or stereotypical perceptions of IPV; other officers exhibited a sophisticated understanding of the dynamics of abusive relationships." These attitudes may rely on the level of education an officer attains. For example, police officers with limited professional education were more likely to endorse "rape myths" (Page 2007). With respects to the characteristics of other sex crimes, prostitutes and male rape victims are perceived as not being "real victims" (Gilmartin-Zena 1988; Miller and Schwartz 1995; Mitchell, Hirschman, Nagayama-Hall 1999). More nuisance police perception research explores police perceptions on individuals with mental health problems.

Mental Health

Recent research sought to examine police perceptions of individuals suffering from mental illness issues (Ruiz and Miller 2004; Wells and Schafer 2006; Godfredson, Thomas, Ogloff, and Luebbers 2011). The value of this research lies in the fact that a disproportionate rate of those incarcerated suffers from mental health issues. During the 1970s, the U.S. government initiated the deinstitutionalization of mental health institutions, which required the removal of individuals with mental health problems from state mental health facilities to community-based treatment facilities. This transfer to community-based facilities would allow "the greatest degree of freedom, selfdetermination, autonomy, dignity, and integrity of body, mind, and spirit" (Report to The President's Commission on Mental Health 1978). Unfortunately, many removed mental health patients never received treatment in these community-based facilities. As a result, a significant portion of these individuals found themselves under the auspices of the criminal justice system (Jemelka, Trupin, and Chiles 1989; Steinwachs, Kasper, and Skinner 1992; Lauriello, Bustillo, and Keith 2000; Stephan 2004). In fact, many correctional facilities turned into "mental health facilities" by default because of the high rate of prisoners with mental health problems. For these individuals, their initial encounter with the criminal justice is initiated by their confrontation with police officers. In a recent study, police officers appeared to be frustrated in addresses problems surrounding individuals with mental health issues (Cooper, Mclearen, and Zapf 2004). These feelings of frustration stems from the fact that the police have to devote more resources and time in dealing with the mentally ill, which also takes them away from other responsibilities (Godfredson, Thomas, Ogloff, and Luebbers 2011).

Police Perceptions of Crime

There is much research on public perception of the police (Bordua and Tifft 1971; Jacob 1971; Smith and Hawkins 1973; Albrecht and Green 1977; Manning 1977; Gaines, Kappeler, and, Vaughn 1994; Correia, Reisig, and Lovrich 1996; Huang and Vaughn 1996; Boni and Packer 1998; Dantzker and Kubin 1998; Perlmutter, 2000); however, the research on police perceptions on crime is limited and they tend to focus on job performance (Fry and Greenfield 1980; Greene 1989; Worden 1995; Kakar 2003; Vila 2006). A handful of studies seek to investigate police perceptions of crime. For example, Paulsen (2004) examined whether crime statistic maps effected police perceptions of crime. For this study, Paulsen distributed GIS and GPS daily crime maps to a control group of police officers and the non-control group did not receive any crime level information. After two months of reviewing the information, Paulsen found that the influence of geospatial crime maps did not change the perceptions of crime among the police officers.

In another study, Craglia, Haining, and Signoretta (2005) interviewed several police officers about their perceptions of crime and compared them to the official crime rate in a U.K. city. The findings found a significant difference between the police officer's perceptions of crime to the official crime rates. The researchers found that neighborhoods characterized as being low levels of socio-economic areas were perceived as having higher rates of crime. While crime statistic maps were found not be a factor, police calls for assistance appear to be a significant in police perceptions of crime. Lopez and Lukinbeal (2010) found that the police officer's perception of crime were significantly influence by the amount of reported crimes. Sun and Triplett (2008) compared police and community member perceptions of social problems occurring in mid-western neighborhoods. Findings found that the police perceived that there were more serious problems occurring within the neighborhoods than the residents. The researchers believe that the police's perception may rest on two assumptions: 1) the police are exposed directly to the neighborhoods' social problems, and 2) the "police may over-rate neighborhood problems to justify the need for police presence and to seek more resources" (Sun and Triplett 2008:450). Indeed, police perceptions of crime can be questioned; however, society has also taken for word their thoughts about crime, violence, and justice (Erikson 1994; Loader 1997). For better words, they are "cultural producers" with respects to framing the narratives on crime and violence (Loader 1997).

Within their position, the police regulate political and social order, reiterate the state's right to control its members (Habermas 1979), and motivate them to obey the law (Hough 2007). Given their powerful position, sociologists investigate the cultural significance of policing (Ericson and Banton 1964; Skolnick 1966; Bittner 1970; Cain 1973; Haggerty 1997; Loader 1997; Manning 1997; Waddington 1999; Reiner 2000; Freiberg 2001; Loader and Mulcahy 2003; Innes 2004; Goldsmith 2005). As agents of social control, the police enforce law and social norms to maintain the status quo of society. That is, "we need laws to govern human behavior; and we need state force to ensure compliance with those laws" (Jackson and Bradford 2009:494 see also Hough 2003, 2004).

CONCLUSION

Given the above literature, the question remains as to how the police and police perceptions can come to be reified as realities for the criminality within the community. Indeed, the police are key actors in maintaining law and order, and as a result, they provide a discourse on all crime-related issues. The following chapter provides an overview of the theoretical frameworks drawn on here to explain the phenomenon of police perceptions, the carceral state and symbolic power of policing.

CHAPTER IV

THE THEORETICAL FRAME - THE SYMPOLIC POWER OF POLICING

Criminologists argue that the police are symbols of authority and social order (Manning 1997; Freiberg 2001; Loader 1997; Loader and Mulcahy 2003). However, society's view of the police appears to be a mixture of contradictions. Indeed, "the police protect us from crime but they also intrude into our lives. We want the police to target others – those we hold responsible for crime and disorder – and we clamour for more visible deterrent patrolling and a style of policing more responsible to local needs" (Jackson and Bradford 2009:494 see also Hough, Joseph, and Qureshi 2002). Irrespective of these attitudes, the police are formal agents of social control and are a fixture in society to ensure that individuals comply with the law that governs their behavior (Hough 2003, 2004). Above all else, the police hold a symbolic position within society.

The nexus between the symbols and human behavior have long been a topic of discussion among scholars. Some sociologists believe that symbols can send subtle unconscious messages throughout society that directs human behavior. To understand human behavior requires scholars to think beyond the position of individuals within society. As Durkheim (1970:250) notes "that social life must be explained, not by the conception of those who participate in it, but by deep causes which lie outside of consciousness." Realizing this, Bourdieu (1991) seeks to understand the power of symbols on human behavior. For Bourdieu, symbols can hold substantial power which

can impact an individual's daily routine. To explain this, he develops the idea of symbolic power. Bourdieu (1991:170) defines symbolic power as:

A power of constituting the given through utterances, of making people see and believe, of confirming or transforming the vision of the world and, thereby, action on the world and thus the world itself, and almost magical power which enables one to obtain the equivalent of what is obtained through force (whether physical or economic), by virtue of the specific effect of mobilization.

By its virtue, symbolic power can shape individuals' behaviors and realities about life in society. Similar to other institutions within society, the police exemplify symbolic power.

Drawing on the work of Bourdieu (1991), Loader (1997:2) attempts to "illustrate the symbolic power" of policing. Loader argues that the police possess both coercive and symbolic power. While police coercive power is seen through the use of arrest and physical restraint, the display of symbolic power is different. Loader (1997:3) views "symbolic power as an invisible power, inculcated through instruction, habit and routine, as power misrecognized as such, even exercised by those who are subject to it." For Loader (1997:3), symbolic power grants police the "a power to diagnose, classify, authorize, and represent both individuals and the world, and to have this power of 'legitimate naming' not just taken seriously, but taken-for-granted." It is within this capacity, the police are entitled to discuss crime related problems. More importantly, their ideas about crime are rarely challenged (Loader 1997:3). These uncontested thoughts about crime are related to the display of police symbols. Police Symbols

Recall that Bourdieu (1991) argues that symbolic power illustrates the unconscious methods of cultural and social control over individuals. With respect to the police, images of power extend beyond policing ceremonies to include tangible items such as police uniforms, equipment, and weapons. As Loader (1997:4) notes:

The iconography of policing - the handcuffs, fingerprints, cop shows, uniforms, photo fits, picture postcards, memoirs, cars, sirens, helicopters, riot shields and so forth - connect with and re-articulated is positions towards, and fantasies of, policing that already pertain within the wider culture.

These police icons project the symbolic power of the police towards societal members.

The mere presence of the police reminds individuals about the dangers of criminal

victimization and the need for their protection (Loader 1997). Also, the power to

intervene (with force) functions as a reminder that the police have the authority to resolve

conflict. Loader (1997:9) notes:

The craft skills and coercive powers that police officers deploy on a daily basis are not just goal-oriented. They serve too to communicate meaning, not only about the police and their role, but also about power and authority in society. Similarly, the symbols and signs within which police work is encoded are not merely decorative, an epiphenomenal gloss on the material practices of policing. They too have a practical effect and are mobilized for instrumental purposes.

While coercive skills are necessary to convey images of power and authority, the police

also rely on subtle methods to ensure control.

Police Voice and Legitimate Namers

Coercive tactics to control society is not enough, instead police rely on the "police voice" (Loader 1997). To further understand the cultural significance of policing, Loader also draws on the work of Turner (1974), more specifically his ideas on "condensation

symbol." According to Turner, condensation symbols combine the diverging cultural meanings of a subject. Simply put, condensation symbols "condense many references, unifying them in a single cognitive and affective field" (Turner 1974:55). With respects to policing, Loader (1997:8) contends that the police "evoke – at a paleo-symbolic level – some deeply felt fears and anxieties, and some equaling compelling hopes and fantasies." That is, the police serve as cultural produce agents with the mission of producing, reinforcing, and circulating cultural and societal norms. More importantly, the police enjoy "a privileged place in constructing representations of crime events" (Loader 1997:10), which is frequently utilized by the media to present crime and justice news. Ericson (1994) refers the police as "knowledge workers" as they are able to produce and circulate information about crime and security. Loader (1997:10) contends that "as a body possessing an aura of authority and knowledge ability, the police are well-placed to 'name' contemporary problems and diagnose their causes." Also within their role, the police are able to frame individuals as criminals or the "other."

Loader (1997:5) argues that the police enjoy the "power of legitimate naming." In fact, the police are a "constant reminder[s] of the existence of the undesirable, criminal other" (Loader 1997:8). The "other" are characterized as the indigents living in poor areas and are often perceived to have norms and cultural attributes that are immoral and defective (Young 2003:395). Such labels often serve as a mechanism to ensure that these individuals are unable to participate in cultural values or norms (Massey 2007). For instance, the Reagan administration constructed racist ideologies to legislate discriminatory drug policies (Culverson 1998). In their quest to pass stringent drug policies, conservative politicians demonized crack cocaine users who were typically African-American. Walker (2011) argued that the number of crack babies being born was grossly exaggerated by politicians and the media to scare the public and justify drug policies directed towards African-Americans. For example, the Narcotics Penalties and Enforcement Act of 1986 and the Omnibus Anti-Drug Abuse Act of 1988 established harsher penalties for the possession and trafficking of drugs used by African-Americans (Mahan 1996; Bush-Baskette 2000). As a result, the police targeted black communities in inner city areas based on these racial stereotypes, which reinforced the negative beliefs about African-Americans and drug use. Given their quest to win the war on drugs and crime, the police have served as a barometer in detecting the characteristics of illegal drug users. In fact, Loader (1997) would argue they have become the legitimate namers within the war on drugs. Just as African Americans have been subjected to racist stereotypes, American Indians also suffered from negative labeling which is used to relegate them to the lower fringes of society.

The perpetuation of negative stereotypes about American Indians and Native Alaskans (AI/NA) has its origins in the early founding of the United States. In fact, Anglo perceptions of American Indians evolved to suit the needs the dominant societal group or provide a rationale for anti-Native policies (Churchill 2004; O'Brien 2011). Early colonists needed assistance from the Indigenous population for survival purposes (i.e. trade, learning fishing and hunting techniques or farming) and often depicted American Indians as being "brave" and "noble savages" (Churchill 2004). Then, during the "Manifest Destiny" period, many whites viewed American Indians as being an obstacle in the acquisition of tribal lands, and as a result, the Native population was negatively stereotyped as being "blood thirsty savages" and "animalistic." Other negative stereotypes of American Indians include "drunken Indians" who are receiving free money from the government and are wealthy from revenue generated from Indian gaming casinos. Negative American Indian stereotypes contribute to their increased involvement in the criminal justice system. Given the long history of negative Native stereotypes portrayed by the public and media, Leonard Peltier (2001) suggested that these perceptions contribute to the criminalization of American Indians and that American Indians are increasingly criminalized for their participation in their culture and ancestral traditions; therefore, the criminal justice operates as a mechanism that criminalizes individuals for simply being "Indian." In conjunction with the criminal justice system, the media and society perpetuates the "Other" effect on AI/NA population by promoting negative stereotypes such as "drunken Indians" and "savages" which makes them appear to be subhuman and prone to criminality. These negative minority stereotypes serve as a constant reminder of the "other" and the need for the police to subdue these individuals.

Policing and Revealing "Stories"

Loader (1997:2) suggests that the actions and policies of the police can reveal "stories" about society. By understanding the nature of policing within society, sociologists can learn about the condition of society as well as the placement of individual groups. As Loader (2006:211) notes:

Every stop, every search, every arrest, every group of youths moved on, every abuse of due process, every failure to respond to call or complaint, every racist ... sexist ...homophobic, every diagnosis of the crime problem, every depiction of criminals – all these send small, routine, authoritative signals about societies conflicts, cleavages and hierarchies about whose claims are considered legitimate within it, about whose status identity is to be affirmed or denied as part of it. That is, the police enforce "structures of feeling" as they are the focal point in the "production and reproduction of order and security" (Loader 1997:3). For instance, Loader (1997:7) states that "the way we police, represent policing ourselves, and position it within an overall sense of order, makes a difference to both the construction of individual subjectivities and the quality and character of social relations." In the U.S., the over and under-policing of women and minorities suggest that there are gender and racial inequalities. That is, these policing strategies tell "stories" of society's treatment of women and minorities.

From unequal employment opportunities to the lack of accessible healthcare, gender inequality is pervasive in society. Although women have made great strides in gaining equality, the patriarchal social structure continues to oppress women (Barak, Leighton, and Flavin 2010). Given the ubiquitous nature of patriarchy, it is not surprising that many women within the criminal justice system experience gender inequality. Indeed, gender plays a significant role in the criminal justice system's response to women (Ferraro 1996; Chesney-Lind 2002; Belknap and Potter 2005; Humphries 2005; Peterson and Dixon 2005; Belknap 2007; Gruber 2007. For instance, many sexual assault and intimate partner violence (IPV) victims are subjected to under-policing (Pagelow 1987; Bowker 1982; Oppenlander 1982; Brown 1984; Ferraro 1989; Websdale 1995; Websdale and Johnson 1997).

Loader (1997) would suggest that the lack of adequate police responses to women victims of crime illustrate the placement of individuals within society based upon their gender. That is, the under-policing of women depicts a society heavily under the influence of patriarchal social structures. Loader (1997) views the police as agents of the state with the purpose of producing, circulating, and reinforcing gender inequalities. From the hesitation of enforcing protection orders to dismissing claims of sexual abuse, the lack of police protection for women is troubling. Indeed, the criminal justice system and policing strategies are forms of patriarchal social control directed towards women. Although females experience issues with policing, minorities are subjected to extraordinary policing through the use of racial profiling.

Despite the passage of Civil Rights Legislation in the 1960s, minorities continue to experience unequal treatment within the criminal justice system. As previously mentioned, the criminal justice system operates as system of social stratification that frequently captures minorities (see Massey 2007). Unfortunately, minorities are increasing targeted by the police (Gabbidon 2003; Russell-Brown 2004; Gabbidon & Higgins 2007; Gabbidon & Higgins 2008; Khoury 2009). As a result of being subjected to extraordinary policing, minorities are disproportionately represented in the criminal justice system (Khoury 2009). Loader (1997) would argue that the over-policing of minorities reveals racial inequalities. Indeed, the police and their strategies serve as a mechanism to ensure the oppression of minorities. Racial profiling allows police officers to use race as a component of criminality (Meehan and Ponder 2002; Russell-Brown 2004; Welch 2007).

For Loader (1997), the police use of racial profiling serves as a tool of exclusion. As Loader (2006:212) notes:

Police utterances and practices can operate to communicate powerful exclusion signals, practices that indicate that particular groups – teenage boys, or Black males, or Muslims – are not considered full members of society. The patterns of oversuspicion and overprotection that long marked police relations with disadvantaged groups send an oblique signal of this kind.

Additionally, the use of racial profiling relegates minorities to lower status in society. These policing strategies operate as a mechanism of social stratification. Moreover, the use of racial profiling allows police to label individuals as criminal based first and foremost on their race/ethnicity.

SYMBOLIC POWER OF THE POLICE ON THE REZ

Given the nature of their occupation, tribal police officers are in position to provide insight about crime and justice related problems in Indian Country. As previously mentioned, this dissertation investigates the perceptions of crime and justice among tribal police officers. Similar to inner cities, reservation communities experience high rates of crime and violence. As one journalist reported, "as far as criminality, this is the pinnacle [describing the rate of violence on the Wind River Indian Reservation]" (Williams 2012c:3). As a result, tribal police officers are on the "frontlines" when dealing with crime and violence in Indian Country. More importantly, tribal police also serve as guardians for American Indians living in reservation communities. Loader (1997:8) notes the significance of the "guardians" moniker with respects to policing:

The police, in their omnipotence and potential 'everywhereness' are imagined as 'guardians.' This may take a number of forms. It may be embodied, for instance, in the dramatic appeal of bandit-catching as a form of policing, where the police are enacted as immediately and actively apprehending the 'other.'

The "guardian" label may serve as a twofold purpose for tribal police officers. First, the police are the "bandit-catchers" in Indian Country and a constant reminder of the presence of the "other." Second, the tribal police may also serve as protector of the Native culture. That is, the tribal police are responsible for defending and protecting

individuals and their way of life within tribal communities scattered across Indian Country. That is, tribal police officers are seen as a powerful condensing symbol.

Loader notes (1997:16) "the police – like the monarchy, a powerful condensing symbol – provide an enduring and affectively appealing means by which a certain strata of English society is able to imagine 'England' as a community." Although Loader's reflections focus on English society, his observation may have usefulness in examining the dynamics of life within Indian Country. It could be argued that tribal police officers serve as condensation symbols of law and order as well as establishing parameters of tribal sovereignty. That is, the tribal police are symbols of tribal sovereignty by allowing American Indians to police themselves within their own communities.

Additionally, tribal police officers are seen as the "experts" and have the ability to "diagnose, classify, authorize and represent both individuals and the world" (Loader 1997:3). It is within this capacity that tribal police officers have "ownership and framing of relevant issues, and whose individual and collective utterances circulate meanings that contribute in potentially telling was to the formation of opinion and belief" (Loader and Mulcahy 2001:42). Irrespective of these cautions, the perceptions and opinions among tribal police officers can offer an insight about life on reservation communities. As Loader (1997:2) suggests, the way a society polices its people can "tell stories about itself." Drawing this concept to the present study, the manner in which policing on the reservation occurs can reveal "stories" about life and social problems in Indian Country. It is with this understanding that Loader's policing perspective may help to understand the perceptions of crime and justice among tribal officers working within a southeastern reservation community.

CHAPTER V METHODS

This chapter discusses the method utilized in the present dissertation research. This study draws on multiple in-depth interviews of tribal police officers working within the Eastern Band of Cherokee Indians' police department to address the research question: What are the perceptions of crime and justice in Indian Country among tribal officers? The chapter first provides a brief description of the participants, then proceeds to explain the nature of focus group and in-depth interviews as well as the rationale for utilizing this method. Lastly, the chapter discusses the research plan.

RESEARCH PARTICIPANTS

Participants for this dissertation research were recruited from the Eastern Band of Cherokee Indians Police Department. This police agency provides the tribal community with various services that range from patrolling reservation roads to providing security to the tribal courthouse. The police department is responsible for providing assistance to 15,000 tribal residents scattered across 82.6 square miles of tribal lands. The department is headed by a police chief who oversees 60 tribal police officers. Unlike many tribal police departments, this department is able to establish specialized units focusing on specific crimes such as illegal drugs, child abuse, and serious crimes. For the purposes of this study, all policing personnel were invited to participate. Before discussing the research plan, it is important to provide a historical overview on the tribe.

THE EASTERN BAND OF CHEROKEE INDIANS

The Eastern Band of Cherokee Indians (EBCI) is a federally recognized American Indian tribe in the United States located in Western North Carolina. This tribe is part of the Cherokee people and closely related to the Cherokee Nation and United Keetoowah Band of Cherokee Indians in Oklahoma. Similar to other AI/NA tribes, the EBCIs have deep historical and cultural roots. Archeologists found artifacts associated with the Cherokee Indians that date over 11,000 years ago - at the end of the last ice age - and they believe that by 1500 B.C. the Cherokee Indians developed their language (Perdue 1985). However, historians believe it was not until 200 A.D. that the tribe evolved into establishing their own distinctive culture in the southeastern United States (O'Brien 2001). For instance, the Cherokee culture established working roles based on gender. Cherokee women tended to crops – beans, corn, and squash also known as "the three sisters" (Perdue 1989) - while males carried out hunting and fishing duties. The tribe maintained a matrilineal societal structure in which the women served as head of the house and elected tribal chiefs (Perdue and Green 2005). Similar to other AI/NA tribes, the Cherokee practiced ceremonies, festivals, and rituals to honor their gods, tribal people, and the seasons (Perdue 1989, Perdue and Green 2005); some of these activities are still practiced by the EBCI. At the height of their power, the Cherokee were one of the dominant tribes in the southeastern United States, to the extent that the tribe inhabited over 140,000 square miles stretching from Virginia to Alabama (Perdue 1985; Perdue and Green 2005). Cherokee Indians lived in "towns" or villages that were comprised of 30 to 60 homes and a council house. The council house served as a government and ceremonial building and also the site for the scared fire where ceremonies and rituals

were preformed. Although the Cherokees commanded extraordinary power in the southern region of the United States, their power greatly diminished with the arrival of European settlers.

In 1540, Spanish explorer Hernando De Soto was one of the first Europeans to interact with the Cherokee, but it was not until the 1690s that the tribe began to trade with the colonists. The fur trade of the 1700s and 1800s greatly increased the Cherokee's interaction with the colonists. As O'Brien (2011:113) notes:

Europeans traveled to Indian villages to obtain fur and created companies to manage the immense fur trade that became crucial to the colonial economies...Indian people adapted easily to the new demand for animals that they already hunted on a limited scale, and the trade enabled them to acquire a wide arrange of European-made merchandise.

The Cherokee traded deerskin with local settlers in exchange for tools and other goods (Perdue 2005). Deer were an important source for food for the Cherokee and the animal "provided as much as a third of the annual diet for most Indians" (O'Brien 2011:115). The Cherokee Indians also used the various part of the animal for clothing, tools, and ritual objects. Unfortunately, relations between the Cherokee Indians and Europeans deteriorated, leading to conflict among the two groups as well as inter-tribal fighting. As a result of tensions, the Cherokee tribe splintered into several groups. Historians assert that the building tension between the colonists and Cherokee Indians led to the Tuscarora War of 1711 and the Yamassee War of 1715 (Powell 2006; O'Brien 2011). Although the Cherokees sustained substantial losses during these wars, European diseases proved to be more responsible for the decline in their population than warfare (Perdue 1985; Perdue and Green 2005; Calloway 2006). Between 1738 and 1739, nearly half of the Cherokee population succumbed to smallpox (Calloway 2006). In another setback, the

Cherokee people were forced to concede over 50,000 square miles of land to colonial governments to appease white settlers following the French and Indian War (Perdue and Green 2005). During the American Indian Revolutionary War, the British government approached the Cherokees to wage war against the colonies in exchange for loss tribal land. In retaliation, the colonists destroyed over 50 Cherokee villages during the summer of 1776, and the Indians had to concede even more land to the colonists. In the face of adversity, the Cherokees were able to initiate several progressive initiatives.

To maintain their political autonomy, the Cherokee entered into several government treaties; however, a stipulation for their ability to self-govern was to concede more land to the government (Perdue and Green 2005). The 1820s is characterized as a "Cherokee renaissance" (Powell 2006: 209) which saw the advent the Cherokee written language by Sequoyah in 1821. The development of a written language allowed the entire Cherokee population to become literate, which also lead to *The Phoenix* a Cherokee newspaper which was published in their native language as well as English. The Cherokees also adopted their own constitution based on the U.S. Constitution. Interestingly, the Cherokee constitution made tribal land sales illegal (O'Brien 2001). The Cherokees embraced other American traits such as implementing school and government systems, owning African slaves, and adopting the latest advances in agriculture. As a result of these developments, the tribe nearly assimilated into the white Anglo culture. Their quiet existence in the Appalachian Mountains made it possible for them to actively trade with White settlers (Perdue 1989). The tribe's cohesion with the dominant culture was reflective in the numerous inter-marriages between the two groups (Perdue 1989; Finger 1981). Although the Cherokees were assimilating into the Anglo

culture, they were not immune to the takeover of their ancestral lands. As O'Brien (2001:164) suggests, "such expressions of civilization ironically made them more vulnerable to removal because Americans wanted their productive land." The 1828 election of Andrew Jackson changed the lives of many Cherokee Indians.

President Jackson and other leading politicians wanted to remove American Indian tribes from the eastern United States largely for agricultural and mining purposes. In 1830, the U.S. Congress passed the Indian Removal Act which forcefully removed the Cherokee Indians and other tribes to Oklahoma. This passage of this Act led to the Trail of Tears of 1838 to 1839 which was responsible for the deaths of numerous Cherokee Indians and "the loss of property at the hands of rapacious whites" (O'Brien 2001:166). However, a group of 400 Cherokee Indians remained in the mountainous region of Western North Carolina. While attempts were made to remove this small group of Indians from the Appalachian Mountains, the band was able to circumvent the federal government's Indian removal policies for a while. The band of Indians were successful enough to maintain their ancestral lands through the help of Principal Chief Yonaguska and William Holland Thomas (Finger 1981) by lobbying government officials to allow them to live on their lands (Perdue 1985). Thomas and the Cherokees were able buy land, which is referred to as the Qualla Boundary, thus establishing the Eastern Band of Cherokee Indians tribe. The tribe was recognized by the state of North Carolina in 1866 and received federal recognition in 1868 (Perdue and Green 2005).

Currently, the EBCI live on a 56,500 acre reservation located in the western portion of North Carolina. The reservation area is mountainous; in fact, a substantial amount of the reservation land borders the Great Smoky Mountain National Park. 75

Presently, there are over 13,000 enrolled tribal members (Perdue and Green 2005). Although the tribe receives funding from both the federal and state governments, a substantial amount of funds comes from tourism. During the summer months, thousands of tourists descend onto the reservation community to visit attractions such as the Museum of the Cherokee Indian and an outdoor drama titled "Unto These Hills" which recounts the tribe's history. However, the most notable attraction in the community is Harrah's Cherokee Casino and Hotel which generates a substantial amount of revenue for the tribal community. Funds generated from the casino are used to improve the reservation community's infrastructure. Remarkably, the tribe used in the current study has a long historical legacy. The following sections will discuss the methods that were drawn upon to investigate the research question.

IN-DEPTH INTERVIEWS

This dissertation research relied on in-depth interviews to understand the factors that contribute to the perceptions of crime, violence, and justice. In-depth interviewing involves conducting an intense conversation with a participant in order to understand their perceptions, opinions and thoughts about a subject (Creswell 1998, 2007). This qualitative approach provides a glimpse into the realities experienced by the participant. The participant is considered to be the expert about the problem being studied and the interviewer is seen as the student (Berg 2007; Creswell 2007). The typical format of indepth interview is as follows: (1) the interviewer poses a question to the participant, (2) the observer listens intently to the participant's responses and records the response, (3) and the interviewer can ask follow up questions based on the participant responses (Berg

2007; Creswell 2007). This type of interview requires a face-to-face conversation and involves one interviewer and one interviewee. Typically, these interviews can last a couple minutes to several hours in length.

There are several benefits associated with in-depth interviewing. It provides participants an opportunity to express their thoughts, opinions, and reflections about themselves or about a particular social issue (Creswell 2007). Additionally, many individuals participating in an interview often find it cathartic to discuss their life experiences and thoughts and have their views shared with an individual (Berg 2007). Another benefit of in-depth interview methodology is that the data are much more detailed compared to other data collection methods (i.e. survey research) (Berg 2007; Creswell 2007). Aside from these benefits, there are also some limitations with this approach. Critics of in-depth interviewing argue that this technique does not yield generalizable data and findings cannot be inferred to that of the general population (Berg 2007). This method can also be time consuming because it takes a considerable amount of time to conduct interviews, transcribe the dialogue, and analyze the data. Lastly, this technique requires the interviewer to be trained in interviewing techniques including to avoid relying on leading questions, appear interested in the participant's responses, and keep their personal thoughts and opinions to themselves during the course of the interview. Forgoing these limitations, this technique is appropriate in exploring tribal police officers' perceptions of crime and justice.

RESEARCH PLAN

To understand the perceptions of criminal justice officials on crime, violence, and justice among American Indians, a series of in-depth interviews was conducted among tribal police officers. This dissertation research relied on convenience sampling – a non-probability sampling technique – where participants were recruited based solely on the convenient accessibility of the researcher (Creswell 2007). Simply put, "this category of sample relies on available subjects – those who are close at hand or easily accessible" (Berg 2007: 43). Moreover, "this strategy is an excellent means of obtaining preliminary information about some research question quickly and inexpensively" (Berg 2007:43). The participants will share insight about their thoughts on problems confronting their tribal community.

The goal was to conduct at least 20 in-depth interviews among tribal police officers. A couple of weeks prior to the interviews, the participants were asked if they were willing to participate in the study. In order to recruit these participants, the researcher asked the tribal police chief to notify potential participants about the interviews and encourage them to participate. Interviews lasted up to an hour. Additionally, the participants were informed about the nature of the study as well as their rights and that they could remove themselves at any point during the interview. Prior to the interview, the researcher introduced himself and provided participants with a brief overview about the interview process, such as establishing rules and encouraging their involvement. Once the participants agreed to the interview, the interview began. The interview schedule can be found on Appendix A. These interviews were audio recorded and later transcribed. Once all of the data were transcribed, the researcher checked the transcriptions with the audio recording to ensure reliability. In agreement with the Eastern Band of Cherokee Indian Tribe, a final report will be given to the tribal community.

PROTECTION OF SUBJECTS

Old Dominion University and the Eastern Band of Cherokee Indian's Tribal Cultural Resource Center reviewed and approved this dissertation research project in the summer of 2012. Judge J. Mathew Martin (tribal court justice) and Dr. Mona Danner (dissertation advisor) provided letters of support and Dr. Danner served as the coinvestigator. The research was conducted ensuring that all IRB rules and regulations will be enforced.

During the data collection, participants were notified about the nature of the study. In compliance with ethical research protocol, they were also be informed of their rights in the study, including their option to abstain from answering any questions and/or that they may remove themselves from the study at any time. The researcher also informed the participants that the interviews were to be recorded and upon completion of the research, the data will be destroyed to enhance anonymity. To enhance the anonymity of the participants, pseudonyms were used during the transcription process as well as in the final report.

CHAPTER VI FINDINGS

This study explores the perceptions of crime and justice among tribal police officers in one reservation community. The following chapter presents the findings from the field interviews. First, the demographics of the participants are introduced, followed by a presentation about life on the reservation community. The chapter then provides a thematic overview on the police officers' perceptions of crime and justice.

POLICE DEPARTMENT DEMOGRAPHICS

Data collection for the research occurred in the fall of 2012. The field interviews took place at the tribal police department and courthouse. The actual time spent for the interviews ranged from nine minutes to over an hour. In an agreement with the tribal government, the participants for this study were restricted to tribal employees. The recruitment procedure yielded 27 tribal police officers who held various ranks from police chief to patrol officers. Four employees of the tribal justice system provided additional contextual insight and their perspectives are also included where appropriate but their responses are not tallied in any of the quantitative results. The age of the 27 police officers ranged from 20 to 62 years old (mean 35.75, median 33) and the gender composition of the group was 24 males and 3 females. Twenty-three of the participants were enrolled members of the tribe and five were non-tribal members who worked within the police department. The years of service among the participants ranged from one month to over 20 years (mean 8.80, median 8).

The interviews were conducted at the convenience of the participants. The researcher noted personal reactions of the participants as they responded to the interview questions, and recorded the full interview. Following data collection, the interviews were transcribed and coded. Strauss and Corbin (1990:61) define coding as "the process of breaking down, examining, comparing, conceptualizing, and categorizing data." With this technique, the researcher identified themes and patterns. For coding purposes, the research identified concepts by key words (i.e. crime, violence, justice etc.) and grouped them into major categories and subcategories. To ensure confidentiality, the names and identifying characteristics of the participants are not included. Because the majority of participants are male, the male pronoun is used in all cases.

COMMUNITY LIFE

To build a relationship between the participant and researcher, the interviews initiated with several background questions, which Hill, Thompson, and Williams (1997) suggest can allow the participant to become comfortable with the interviewing process and help in gathering demographic information. After asking the participants questions about the tribal community, the interviews focused on crime, violence, and justice within the reservation community.

The participants were asked whether they would raise children within the reservation community, in order to allow the participant to describe the characteristics of the community. Several themes emerged from this question. First, an overwhelming majority of officers deemed the community a good place to raise children (n=22). These individuals provided several positive attributes of the reservation that made it conducive

to child rearing such as the rich ancestral history (n=6), close relationships and family ties (n=4), low rates of crime (n=4), "it's home" (n=3), the beautiful landscape of the area (n=2), and/or adequate schools (n=1). Several tribal police officers shared their justifications for decisions to raise their children on the reservation.

There's a lot of good qualities [reservation community]. As far as close-knit community, everybody knows everybody here, we kind of look after each other, a lot of people here are related, you know...but it's not like an urban setting or a mid-sized town or whatnot... I like to think we're in our own world. You know what I'm saying? (Police Officer #12)

It's a real pretty tight community here, everybody knows everybody and somewhere down the line you're related to one another. It's like, it goes back to tradition. You know, like the clan system, the culture, that kind of thing. (Police Officer #2)

These descriptions portray a tight-knit tribal community. In addition, they suggest that

there is a sense of willingness among the tribal members to work together towards

helping their community strive. A small group of officers thought it was safer to raise a

child on the reservation than in the neighboring communities (n=4). As one police officer

pointed out:

Well, I consider it like, pretty much, a low crime rate and you know... but I think overall this is a good place. Because, [I was] in the service, I been all over the place, and I still say [name of reservation] is the best place to raise kids ... got ... everything they need here from our own school system, like I said the crime isn't bad around here. (Police Officer #1)

While this officer had traveled extensively, he believed that the tribal community is an

exceptional place to raise children. Likewise, another officer suggested that the well-

staffed tribal police department was a good quality. He posited, "I think it's a better

place to raise children because there's more law enforcement presence in [reservation's

name] than anywhere else in [part of the state]" (Police Officer #12). While a majority of

the participants thought the reservation was a good place to raise children, a group of police officers suggested otherwise.

A small number of police officers thought that the tribal community was not a good place to raise children (n=5). All of these individuals cited the reservation's high rate of crime as the primary reason for their decision. One police officer observed:

Well I would say it's safe but really it's not all that safe. There's a lot more that goes on here than people know about. If you're not a cop you don't really know that much. I don't know...it's hard to answer that one. I would say family values but it's getting worse. I mean, everywhere is worse. (Police Officer #15)

Coupled with high rates of crime and the lack of family values, this police officer believed that the reservation was not a safe environment to raise children. Mirroring the previous statement, another individual thought that the inconsistency of the tribal criminal justice system created an unsafe community for his children. For this reason, he remarked "that's why I wouldn't send my kids up here for nothing" (Police Officer #13). Finally, another individual felt that his position as a police officer could expose his children to victimization. He conveyed his concerns:

I wouldn't necessarily say it's a good place [reservation community], I'd actually like to live where I don't enforce the law. So I'd like to live, not, I don't wanna say the real world, but in non-native territory... it's because I don't want [to]. I don't even shop locally [on the reservation]. I shop at Food Lion [off the reservation]... Um, I don't want my children associated with me because I don't' want them victimized or perped on because I am what I am. (Police Officer #22)

Without a doubt, being a police officer exposes family members to unintended risks. The initial question on the characteristics of the community yielded several interesting findings. First, all of the non-Native police officers believed that the tribal community was a good place to raise children. While there were several positive and negative

attributes about the tribal community, the interviews also provided an insight about crime and violence on the reservation.

CRIME

To gauge the extent of crime and violence within the reservation, the researcher asked the participants whether it was a fair characterization to label all of Indian Country as a "war zone" to describe the high rates of crime and violence. An overwhelming. majority of the participants thought the analogy was an unfair characterization for all tribal communities in Indian Country. This question elicited several passionate responses from the participants. A police officer denounced this analogy:

I think that's an UN-fair assessment. You know, it's not a war zone here, we got problems that everybody else, you know, every city off the reservation has. You have your breaking and entering, you have your assault. We don't have that many assaults, we don't have really... And usually it's just regular intoxicated person, you know, type deal. Everybody deals with that. Person gets drunk and... so it's no different. I mean, people act the same... (Police Officer #6)

This participant contested the analogy equating his tribal community as a "war zone." Indeed, the tribal officer pushed back the notion that his reservation was in a state of anarchy and believed the tribal community's crime rate was similar to the neighboring outside communities. For this participant, the analogy portrayed American Indians as "savages" and appears as subhuman. That is, the analogy framed his fellow tribal members as the "other," whose transgressions wreaked havoc within the confines of the reservation. However, the tribal officer argued that the people in his community acted and behaved similar to those living anywhere in the country. Although the participant revealed that crimes do occur on the reservation, these events are also similar to one occurring in small town across the United States. Indeed, the tribal officer's narrative portrayed a community that was similar to the fictional town of Mayberry with respects to the local town drunk causing a nuisance within the community. Another officer suggested that the analogy was "harsh" and thought it invoked images of war torn areas of the world. He replied:

No, I don't think so. I mean, that's [like] Afghanistan or something because we're no... I've been here pretty much all my life...A lot of tourists I've talked to enjoy being in this area, we have a low crime rate, and as far as law enforcement, we take care of what needs to be taken care of when it needs to be taken care of. I don't... I wouldn't consider this a war zone. (Police Officer #12)

Again, this narrative rebukes the "war zone" analogy. For this participant, the analogy invoked images of war-torn communities in Afghanistan. The tribal officer suggested that this was not the case in his community. By pointing out that tourists enjoy visiting the community, he tried refute the claims that his tribal community was plagued by high rates of crime and violence. Moreover, this officer felt the tribal police department is equipped to address any problem on the reservation. Indeed, it appeared that many officers were appalled by the analogy. While a majority of the participants felt the "war zone" comparison was unfair to their tribal community, a small group of police officers thought it was a reasonable assessment of their reservation.

Only four police officers believed that the "war zone" analogy described the level of crime and violence in their community. Upon being asked, an officer revealed that outside law enforcement agencies receive fewer calls for assistance than the tribal police department. He stated, "I've got buddies that are state cops, sheriffs, they can go a week, have maybe five calls. I usually average eight a night" (Police Officer #9). In this statement, the tribal officer believed that the war zone analogy captures the level of crime in his tribal community. The participant revealed that the tribal police department receives a high volume of calls for assistance than his law enforcement colleagues working in neighboring jurisdiction outside of Indian Country. The high call volume exemplifies the war zone analogy as these "frontline soldiers" are on call to protect the people on the reservation from crime and violence. In his analysis of the usage of warlike metaphors, Steinert (2003:271) asserts that the service the police afford is "understood to be in the common service of defending 'us' against an enemy: crime and criminals." Indeed, the calls for assistance can be symbolized as the tribal need for police protection to defend them from crime and violence. In a separate conversation, a tribal officer draws on another war-like metaphor to describe the manner of policing on the reservation. A ranking officer mentioned that his patrol officers often complete a tour of duty (patrolling for seven consecutive days) and experience a high call volume. He noted, "we could be on call, we could do seven days, which is one tour of duty, we can do seven days of our tour and in that seven days charge only three people, but we've dealt with a hundred and fifty calls" (Police Officer #22). For this narrative, it is important to note the military idiom "tour of duty," is reminiscent of the "war zone" that suggests the community experiences high levels of crime and violence. Additionally, these participants believed the reservation has a crime rate substantially higher than the neighboring communities. Perhaps one of the most fervent responses to this question did not come from a tribal police officer, but from one of the tribal judicial personnel. The participant appeared aghast by the comparison of his community to worn torn areas in Afghanistan and Iraq. He claimed that such comparisons perpetuate negative American Indian stereotypes. The participant recalled an event where he had to "correct" a non-Native about his view of the community.

Because it happens (crime) on white man's land too. And buddy, I put one person who used to work here in his place. He was from [neighboring community], he was a non-Indian, he walked here (tribal courthouse) and said, "Oh today's Indian love court." Oh my god. Oh my god. I, excuse my language, but I said, "What the fuck did you just say?" I said, "Get in here and close that damn door." And I said, "Don't you ever let me hear you say that again. Because let me tell you something, white people fight and beat up each other just like Indians do Don't you ever fucking say that again." And I'll tell anybody that. Hell no. It ain't just Indians, white people, Mexicans, blacks, they all fight, drink, and do everything and Indian does, but we just get tagged. (Tribal Judicial Personnel #1)

For this participant, crime and violence transcends all race and ethnic groups and not solely experienced among one group. The participant tried to rebuke the stereotype that American Indians are more violent than other racial/ethnic groups in the United States. As previously mentioned, American Indians are stereotyped as the "drunken Indian" and "blood thirsty savages," which makes American Indians appear as subhuman (Churchill 2004; Lujan 2006). Taken together these stereotypes are "absurd and damaging because they perpetuate a distorted image of American Indians" (Lujan 2006:131). Quintero (2001:58) notes that social constructs through these stereotypes "are systematically utilized in specific contexts to disempower the colonized by characterizing Indian people as dysfunctional, pathological, or weak, thereby reinforcing the power of the colonizer." Unfortunately, these stereotypes continue to be perpetuated through the media, which propagates into the general population into thinking that American Indians are violent. With that being said, American Indians continue to overcome negative stereotypes that demonize and cast them as the "other." The participant felt compelled to "set the record straight" with the "outsider" who had a false perception about the people living in her community. Although a majority believed their community did not suffer from extraordinary high rates of crime, they admitted the tribe did have an illegal drug problem.

ILLEGAL DRUG USE

Perhaps one of the most striking patterns to emerge from the interviews was the high rate of illegal drug activity within the reservation community. The police officers identified several illegal drugs that were being abused among the tribal members: prescription drugs (n=22) methamphetamines (n=13), marijuana (n=7), cocaine (n=4), heroin (n=1) and bath salts (n=1). According to the participants, the pervasiveness of illegal drug use has reached epidemic levels. In fact, the officers believed a substantial part of their time is devoted towards addressing illegal drug activities in their community. As one individual noted "we're doing a lot of car stops and they're [tribal police officers] finding syringes and, you know, just drug paraphernalia...where we might not find drugs on them [suspects]...they had the paraphernalia where they'd been using it" (Police Officer #1). Much of the illegal drug activity within the tribal community involved illegal prescription drug use.

Illegal prescription drug use emerged as a major theme among the police officers. As previously mentioned, 22 officers thought illegal prescription drug consumption was a dire problem in the tribal community. Upon being asked about illegal drug use, a participant simply quipped, "we have a lot of pill[s], prescription abuse... prescription pills" (Police Officer #1). Indeed, many officers simply replied that "pills" were being abused among the tribal members. An officer thought that much of the daily activities of the police department centered on reducing illegal prescription drug use. The participants identified several prescription drugs that were being abused such as morphine, codeine, oxycodone, oxycotin, and Vicodin. Although illegal drug use emerged as a major problem, the officers were divided on which age group abused this form of drug frequently.

Over half of the police officers believed illegal prescription drug use was commonly abused by both young and old tribal members (n=15). One of these individuals suggested this form of illegal drug was abused by "sixth graders to sixty or seventy years old" (Police Officer #15). However, another half of the participants thought illegal prescription drug use was solely abused by the Native youth (n=11). Some of these officers blamed "pill parties" for the high level of Native youth consumption of illegal prescription drug (n=5). For many participants, peer pressure served as an impetus for the consumption of this form of illegal drug. A tribal judicial personal admitted that his son heavily abused prescription drugs and he recounted to him about a "pill party."

When my son was on them pills...One day he's talking to me. He said, "let me tell you how bad it was, I'd get invited to a party and it was bring your own pill instead of bring your own beer, bring your own pill." They had a big bowl. Everybody threw whatever, they didn't know what... they'd crush them, they'd snort them. They never shot up, he always snorted the pills, but who's to say that could have been a blood thinner or, you know, a heart pill. He just said that he'd get invited to pill parties. "Whatever kind of pills we had, we took them. We'd crush them all, we'd crush every one of them together and we'd start snorting them..." I said, "You're stupid." He went, "yeah, I was." I can tell now in his speech, it has affected his speech and you can tell him something and he don't (snaps fingers) grasp it like we do. It about ate his damn brain up. (Tribal Judicial Personnel #1)

"Pill parties" consists of obtaining copious amounts of prescription drugs and crushing pills into a powder in a bowl, which is snorted among individuals. Unfortunately, a tribal justice personnel disclosed that his son suffers from brain damage due to an addiction to prescription drugs. Being geographically isolated, Native youth struggle to find activities to overcome the monotony of living in a rural community. All too often, they find themselves attending "pill parties" for entertainment purposes. Aside from the combination of peer pressure and boredom, the police officers noted that tribal members sold prescription drugs within the tribal community for financial gain.

Tribal members selling prescription drugs were discussed among a group of police officers (n=5). Given the recent economic turmoil, many tribal members resorted to selling their prescription drugs. As one participant pointed out, "they're [tribal members] not working. Plus...you know, [the] recession puts a lot of [people out of work]... and I'm thinking the high price of gas has triggered a lot of it" (Police Officer #6). For many, selling prescription drug provided tribal members with "fast and easy money" (Police Officer #5), which afforded them an opportunity to obtain food and other necessities. Another attraction to selling prescription drugs was that it could yield a substantial sum of money.

The pills, you know, they pay a dollar a milligram, so if you've got a 35 milligram Oxycodone, one pill, 35 milligrams, you can sell that on the street for 35 dollars. Now if somebody's got 200 pills that are 35 milligrams each... Now you figure the math on that; \$35 a pill. And they can sell those within, well within an hour if they wanted to. (Police Officer #3)

Given the figures, a tribal member has an opportunity to earn \$7,000, which is highly profitable considering they have free access to healthcare and medication. Several participants suggested that tribal members are exploiting their access to free healthcare by obtaining prescription drugs to sell for monetary profit. Unlike other reservations, this tribal community has an Indian health clinic which is available to all enrolled members of the tribe. Some tribal members view the clinic as an opportunity to obtain prescription drugs to sell to others. Realizing that the tribal healthcare system is being abused, the tribal government has implemented initiatives to reduce the distribution of medication to tribal members. In addition to obtaining prescription drugs from other tribal members, some individuals have resorted to property crimes to obtain illegal drugs.

Another theme that emerged is that tribal members were stealing prescription drugs (n=7). An officer revealed, "they don't steal, rob, and cheat for pot, but they do for the pills" (Police Officer #9). The participants admitted prescription drug addicts were stealing narcotics out of tribal members' homes. An officer reported "we have reports of people's medicine getting stolen from the medicine cabinets [and] their homes" (Police Officer #3). Given that older tribal members rely on medication to overcome their ailments, illegal drug users see this as an opportunity to obtain prescription drugs. Additionally, tribal members have used their care-taker position to exploit these individuals for their medication. Although some tribal members rely on theft to secure prescription drugs, others have utilized innovative methods such as doctor shopping.

Doctor shopping emerged as a theme in these illegal prescription drug use discussions. At least nine police officers suggested that tribal members relied on doctor shopping to obtain prescription drugs. Doctor shopping is the practice of seeking care from multiple physicians in order to acquire prescription drugs. An officer provided his definition on this technique.

Where you go from one doctor to another doctor to another doctor and you tell every doctor, "My back's hurt, I hurt my back..." And so this doctor gives you a prescription, and then you go to another doctor that's in another town and say, "Hey, you know, I tried to get in with the doctor where I live, and it's gonna be two weeks before I can get an appointment." So then that doctor gives you another prescription. You start doctor shopping different counties, or even different states. And so there's actually no test that's, you know... so if I was to go to the doctor and say I hurt my back mowing, there's no test to show how much pain I'm in for them to say, "okay, with these x-rays..." but if it looks normal, I just say, "but I'm really hurting," they give you oxycontin. (Police Officer #13) According to the participants, it was common among tribal drug addicts to rely on doctor shopping to fulfill their addictions. The heighten need for prescription drugs also lured drug traffickers to the tribal community to distribute illegal drugs.

Only a small group of police officers mentioned the role of drug traffickers in the tribal community (n=5). Drug traffickers view the reservation as an opportunity to sell their illegal goods. The enticement stems from the "per cap" check, which is a stipend given to enrolled tribal members which can range from \$4,000 to \$6,000. Tribal members receive a "per cap" check twice a year, typically in June and December. The funds for the "per cap" check come from the revenue that is generated from the Indian gaming casino. With respects to drug traffickers, they are drawn to the reservation community in June and December because it is when the "per cap" checks are released.

Opportunity is more because, like I said, somebody's spreading the word that there's per cap [stipend] and things like that. And they know that there's money available up here, so we're actually being targeted as far as bringing the pills here. Why not sell the pills here? Oxycodone right now... a 30 mg Oxycodone can sell for \$45 a pill. (Police Officer #21)

Indeed, there is an opportunity to generate a substantial amount of money from selling narcotics on the reservation. A police officer stated that "in the month of June we arrested 21 people for drugs. For pills...it was after per cap and... so between June first to July first we arrested 21 people for pills" (Police Officer #21). The "per cap" check impacts the distribution of illegal drugs to the tribal community.

A couple of the participants provided details about illegal drug trafficking in the community. The frequency of drug trafficking reached a fevered pitch. As one participant noted, "every week we'll catch somebody ... with [about] 200-300 pills coming up here to sell" (Police Officer #21). The same participant also recalled an event

related to drug trafficking. He stated "a guy came from Albuquerque on a motorcycle with 36 pounds of marijuana and delivered it to me and my partner in a barn" (Police Officer #21). Interestingly, a week prior to the field interviews, the tribal police department intercepted a substantial amount of narcotics coming into the tribal community. A participant reported that tribal police officers seized nearly 8,000 prescription pills a week before the field interviews. Combined with convenience and product demand, many traffickers are distributing prescription drugs. The tribal officers also shared their thoughts about the negative consequences of illegal drug use among tribal members.

The harmful effects of substance abuse within the tribal community abounds. The participants identified several negative consequences of illegal drug use such as restructuring tribal families (n=11), destroying opportunities (n=5), and health problems (n=2). Some of the participants voiced their frustrations about the use of illegal drugs among tribal members. The tribal community offers numerous resources for tribal members which include free college education, vocational training, employment opportunities within the tribal government and Indian gaming casino, and access to healthcare. However, a substantial portion of the Native youth refuses to participate in these entitlements. A police officer remarked how substance abuse is a socialized process. He explained that Native children observe their parents engaging in substance abuse, and later as adults these children adopt such behaviors.

It's a learning process. My children know that alcohol's not okay, but I had to teach them that. I had to not use alcohol, and... so in turn my children know that there's a different way. But if I was an alcoholic parent and taught my children, "hey, you know, I drink an 18 pack a day or a 24 pack a day, you know, it's okay, I'm all right, you know, what's wrong with it?" then that child learns [it]. (Police Officer #22)

For this participant, American Indian substance abuse is a cycle that is instilled among Native youth and later acted upon as adults. The next line of questions focused on crime within the reservation community.

COMMUNITY CRIME

Before going in-depth about the dynamics of crime on the reservation, the researcher asked the participants if they believed that the tribal community's crime was similar to the neighboring communities. This question revealed several themes. First, the majority of the police officers thought that their tribal community's crime rate was similar to neighboring communities (n=16). An officer stated that "there's any more or any less crime on the reservation than there is in the city population or a county" (Police Officer #7). Similarly, participants believed the reservation's crime rate appeared parallel to other communities across the country. While the crime was similar to other communities, the police officers acknowledged crimes occurred within their community.

PROPERTY CRIMES

Property crimes abound within the reservation. Upon being asked the types of crimes that occur within his community, an officer replied, "what types of crime DON'T I see on the reservation (laughs). We see everything" (Police Officer #22). Indeed, the police officers identified several property crimes such as breaking and entering (n=15), larceny (n= 8), theft (n=7), and burglary (n=1). A substantial number of police officers believed that these crimes are drug related (n=17). As one participant put it "there are crimes here, just little piddy stuff...stealing weed eaters, lawn mowers, something to

pawn [to obtain] drugs. Meth, pills, whatever they can get their hands on" (Police Officer #26). Similar to the previous statement, another officer offered his thoughts about the connection between property crime and illegal drugs:

When we conduct our interviews, usually, it's because they need a fix. You know, they might steal somebody's \$1800 TV or \$2000 TV and for exchange, it might be for 3 pills. Which is, in my mind, to a working person who has spent that money wisely to live the way they choose to, then when somebody takes advantage of them, and... I don't know. It's the pill. Like I said, in our interviews they'll usually say it's for a quick sale or a trade. (Police Officer #4)

This narrative is important for several reasons. First, it captures the police's virulent populism surrounding the use of illegal drugs. Loader would argue that the tribal officer's narrative is producing, reinforcing, and circulating the war on drugs rhetoric through the use of fear. That is, the participant believed that "hard working people" are being victimized through the theft of their personal property by drug addicts. Similar to the Bennett and colleagues (1997) assessment on drug use and crime, the tribal officer suggested that drug addicts are preying on these "hard working" individuals in order to obtain illegal drugs. Also within this narrative, the police officer frames drug addicts as the "other," whose addiction and behavior causes problems on the reservation. Recall that Loader (1997) contends that the police frame the individuals as the criminal of the "other." The tribal officer's tone in this conversation revealed feelings of frustration, which is consistent with previous research on police officer (Gardiner 2012). Moreover, this narrative suggest that perpetrators are willing to break into homes, which shows the power of addiction with respects to committing crimes to obtain illegal drugs.

Breaking and entering emerged as a theme in these conversations. Several police officers admitted that the majority of their calls involve breaking and entering crimes (n=8). A participant claimed that whenever he turns on a police scanner the majority of

calls are breaking and entering. As he said, "it's B&E, B&E, B&E. You know. All the time" (Tribal Judicial Personnel #2). While several participants voiced frustration about the frequency of breaking and entering crimes within the community, they appeared concerned about the loss of the tribal community's ancestral heirlooms and crafts.

The reservation community has a long artisan history which continues into the present. Many tribal members make baskets, wood and stone craving crafts, and pottery to sell to tourists. Additionally, tribal members possess century old family heirlooms, which chronicle the history of the tribe. These artisan crafts are worth hundreds, if not thousands of dollars. One police officer suggested that a deceased craftmaker's crafts can be worth a substantial sum of money (Police Officer #4). Aside from home invasions, reservation craft dealer shops have been targeted by perpetrators.

Here in town we've got a gentleman that owns [Native craft shop], I'll use that for an example. He deals with a lot of Native American type handmade crafts. And probably within the year he had been hit twice. A lot of valuable stuff had been taken. (Police Officer #4)

Some of the participants have taken steps to protect ancestral heirlooms within their family. As one participant revealed, "we got a security system because...me and my husband bought a bunch of baskets from two people. Bought them for what, a couple years or so… we spent a lot of money" (Tribal Judicial Personnel #1). Similar to other stolen items on the reservation, Native ancestral heirlooms and crafts are pawned to obtain illegal drugs. The conversations with the tribal officers suggest that there is a nexus between property crimes and illegal drug use. The tribal officers also provided their perceptions about violence within their reservation community.

VIOLENT CRIMES

Participants revealed different perceptions about the extent of violence within the reservation community. A majority of the non-enrolled tribal police officers (n=4) thought that the level of violence in the tribal community is not as extensive as in neighboring communities. An officer recalled "I don't think it's as violent as most people think it was, like when I started here, they're like, 'Oh, you're gonna have a hard time with this'...and it's not been that, I don't think it's that bad" (Police Officer #21). The officer revealed that his friends cautioned him about being a tribal police officer because they thought the community experienced high levels of violence. However, other officers suggested the there is a substantial rate of violence within the reservation community. As one tribal officer pointed out "yeah, there's definitely crimes of violence that occur here" (Tribal Judicial Personnel #2). Similar to the previous statement, another participant reflected back on his previous occupation as a dispatcher and realized the extent of violence in his community.

I mean, there's things that happen here that you just don't know about. And I never really seen it before I was in [here]. It was like this quiet little town, but then I worked in [here] and really seen how much everything happens all the time. And now I have a completely different outlook. It's really hard to see it. Like for you, you come in, you stay a week or two and it's really hard to see. And it was really hard for me to see even though I lived here for twenty years. But...you can't leave, like... where I live, you know, we have built security systems because somebody will come to your house and steal your TV while you're home. That's pretty much the way it is. (Tribal Judicial Personnel #2)

In this narrative, the participant revealed that people working in the criminal justice system have a greater understanding about the frequency and the characteristics of crime taking place within the community. The officers identified several violent acts that are common within the reservation community: domestic violence (n=17), assaults (n=11)

murder (n=5), child abuse (n=4), gang activities (n=4) rape (n=2), armed robbery (n=2), and shootings (n=2). Although tribal members claim that violence occurred in the tribal community, many of the officers failed to provide details about these events within the tribal community.

Domestic Violence

The police officers identified domestic violence as the most frequently committed violence act within the reservation. An officer suggested, "oh yeah. There's a lot... but it's not as bad as it is out on them reservations out west... But yes, it is pretty bad" (Police Officer #2). The participants claimed the acts of domestic violence included physical assaults (i.e. hitting, kicking, and shoving), sexual and emotional abuse. Additionally, the officers thought that domestic violence was triggered by alcohol and financial hardship.

Given the violent nature of many domestic cases, the police department takes extra precaution when dealing with these incidents. As Police Officer #22 reported "when we deal with a domestic violence situation, we all go in there, four man, three man, five man strong or whatever." These precautions are for good reason as one individual revealed, "there's been domestic violence murders that have occurred here" (Tribal Courthouse official #1). Two participants admitted that domestic violence cases resulted in the victim's death. During the time of the interviews, tribal police officers wore purple ribbons on their uniforms to support domestic violence awareness month. Several police officers acknowledged the tribal community implemented programs for domestic violence victims as well as awareness campaigns to address the issue. A group of participants also suggested that domestic violence extends beyond females to include male victims. Three individuals said that tribal females were the aggressors in many domestic violence cases in the tribal community. An individual noted "you have [Native] women that will fight back" (Tribal Judicial Personnel #1). At least two participants provided a discussion on domestic violence in their community.

Tribal Judicial Personnel #2: There's probably more men filing DV against women...

Tribal Judicial Personnel #1: You know who started that? My son. There was hardly no man... because they felt like they would be tagged as wusses. But my son got all his front teeth knocked out by his baby's momma, with a baseball bat...And she [the perpetrator] runs and tries to do charges. Well, the investigator comes to me and says, "I can't find [Ned], where's he at?" I said, "I don't know." And they said, "Well, she's in there doing charges." I said, "She's in there doing charges? Why?" And he said, "She says that she's been beat up." And I said, "Well, does she have any marks?" And he was like, "No." And I said, "Go to their apartment... and find him." He found [Ned] lying there bleeding out. So [Ned] was the first one, I said, "[Ned], you're gonna do this." And he just looked at me, and from that time on, even the judges will tell you that. My son was the first real guy that stepped forward and wasn't ashamed to come to court and say, "She done this to me."

This narrative is of interest for several reasons. Historically, on this reservation, few tribal men have been willing to press domestic violence charges against a female in the tribal community due to gender stereotypes. This hesitation is consistent with previous research suggesting that male domestic violence victims are reluctant to notify authorities about their victimization (Felson and Paré 2005; Belknap 2001; Dobash and Dobash 1998). Due to masculine cultural influences, male victims may feel they are not "less of a man" or "less masculine" if they came forward about their victimization (Felson and Paré 2005; Dobash and Dobash 1998). For American Indian males, they have a tendency to embrace masculine cultural values (Brown 1997), which may serve as a barrier in reporting the offense. Although reservation communities are physically and socially isolated from dominant society, masculinity cultural values and traits are present for

many American Indians living in these areas. As a result, American Indian male victims are reluctant to hold their perpetrator responsible for the crime. Only recently, has male domestic violence victims began to come forward about their victimization. Irrespective of these challenges, domestic violence continues to remain a problem within the community. Unfortunately, these crimes are relatively unreported to the tribal police department because they often occur within intimate relationship and the victim is reluctant to come forward about their victimization. Aside from domestic violence, the police officers also suggested that the tribal community experiences assaults.

While the participants believed that domestic violence occurred frequently on the reservation, they failed to provide a meaningful in-depth discussion about the nature of these crimes. In fact, some tribal officers briefly mentioned domestic violence and switched the conversation to another topic. Research suggests that police embrace hegemonic masculinity attitudes (Martin 1989; McNamara 1999), which may impact police perceptions about violence against women. Fielding (1994) notes that the police hold misogynistic attitudes (Fielding 1994), which may explain their hesitance in discussing domestic violence within these conversations. Another explanation for the lack of discourse on this subject may rests on the gender of the assailant. It could be argued that the tribal police do not see domestic violence as a crime when the victim is a male, this too is consistent in previous research (Mitchell et al. 1999; Gilmartin-Zena 1988). Aside from domestic violence, the tribal police officers thought that assaults also occurred frequently within the reservation community.

Assaults

Almost half of the tribal police officers mentioned assaults in the discussions (n=11). At least four of these officers believed that these acts were fueled by alcohol. Two of these officers thought these assaults were more violent on the reservation than outside non-tribal communities. At least one participant acknowledged that assault victims require medical attention due to their injuries. In another conversation, an officer thought that assaults were facilitated by alcohol.

It's a high level of assault- with or without a weapon. But here's the kicker though: if we're cousins or we Indian people, we can get in a fight, I can cut you, you can cut me, we're mad at each other for that ten or fifteen hours until we sober up. God dang, we cut each other? But we'll be right back at it the very next night, we'll be drinking side by side, "Oh I got you good." "I got you good." And then sometimes you don't have that, but a great majority of the time that's how it works out. So victims refuse to cooperate because they believe in that "I'll get him back" or "naw, we were just drunk. (Police Officer #22)

The above narrative is consistent with past research on the correlation between alcohol and assaults (Livingston 2008; Gruenewald, Freisthler, Remer, LaScala, and Treno 2005; Brecklin 2002). Given that American Indians experience high rates of alcoholism (Duran, Oetzel, Lucero, Jiang, Novins, Manson, and Beals, 2005; Beauvais 1998; May 1986), it is not surprising that assaults occur frequently on the reservation. Scholars argue that the high rate of alcoholism is contributed to the legacy of cultural oppression and historical trauma that is experienced among American Indians (Brave Heart and Debruyn 1998). That is, American Indians indulge in alcohol to settle unresolved grief from the social ills of colonialism (Quintero 2001; Brave Heart and Debruyn 1998). With that being said, it is not surprising that the tribal police officer revealed that assaults on this reservation are part of the tribal norm. All too often, the victim is reluctant to report the event to the tribal police department. As a consequence, the police department is unaware about the true extent of these crimes. The police officers also shared their thoughts about the causes of crime in the reservation community.

CRIME CAUSATION

Several police officers offered their thoughts on the causes of crime in their tribal community. The participants identified substance abuse (n=17), poor parenting (n=6), and mental health (n=1) problems as main contributors to crime on the reservation. The social problems that were mentioned within these discussions are correlated with the historical legacy of colonialism. As previously mentioned, a majority of tribal police officers thought illegal drugs and alcohol facilitated criminal activity on the reservation. In addition to substance abuse, several of the police officers thought that poor parenting contributed to crime and violence in their community (n=7). For many participants, poor home environment is conducive to socializing Native youth to non-conforming norms and expectation, and as a result, these individuals engage in criminality and deviance. The police narratives on the nexus between broken homes and crime supports criminological research that suggests that negative family and home environment may impact an adolescent's propensity to commit crime or abuse illegal drugs (Gibson 1969; Wells and Rankin 1986; Petrosino, Derzon, and Lavenberg, 2005; Boutwell and Beaver 2010; Theobald, Farrington, and Piquero 2013). However, research also suggests that Indian boarding schools greatly impacted the dynamics of many Native families for generations (Smith 2005). It could be argued that Native families in this tribal community are still negatively affected from the social ills of colonialism. Another police officer thought that there is a persistent pattern of behavior within Native families

that are related to future crime behavior. In fact, one of the tribal officers conveyed his

thoughts about the historical trauma that was experienced in these institutions.

We are a very violent people. And I don't know if that's because of the historical trauma, what we've endured... that's why the Indian Welfare Act, I saw that on Dr. Phil...drove me crazy. They were talking about Indians being placed in Indian homes because it's so important...but what they seem to forget back in the day is. when we were trying to be Christianized, or to be whitenized, our children were taken. Were forcibly taken from families and homes and put in boarding schools where they experienced severe sexual abuse, severe physical abuse, starvation, mutilation, you name it. That's what happened to these kids. And as they aged out, some of these kids went on and didn't return back to the reservation, but a lot of them did. So they brought that with them with the trauma they experienced, plus the trauma that their parents experienced, feeling helpless because they couldn't protect their children. That's your divine right as a parent, is to protect your children. They couldn't do neither. So you have this angry, and it just kind of evolves. And you can kind of see it in abuse. They were abused, they were neglected, their parents were alcoholics, they were sexually molested... (Police Officer #22)

The above narrative supports prior research that suggests that Indian boarding schools are responsible for anti-social behaviors and mental health problems among American Indians (Henderson, Kunitz, Gabriel, McCright, and Levy 1998). Indeed, the historical trauma that was experienced among former Indian boarding students manifests through alcoholism (Henderson et al 1998), depression (Manson, Ackerson, Dick, Wiegman, Baron, and Fleming 1990), suicide (Dinges and Duong-Tran 1992) and domestic and family violence (Smith 2005). For this participant, the horrors of Indian boarding continue to impact the members of his community. Realizing the historical legacy of colonialism, the researcher inquired whether contemporary sources of colonialism impact the behaviors of people within this reservation community.

The researcher asked the police officers whether outside influences (mainstream culture and the media) impacted crime on the reservation. Scholars argue that the media does have an impact on human behavior and their tendency to engage in transgressions (Kidd-Hewitt 1995; Dotter 2002; Kort-Butler 2012). Given the magnitude of social media followers and the internet, it could be possible that tribal members adopt outside cultural traits rather than their native cultural values. That is, tribal members could be exposed to media depictions of pro-deviance traits and attitudes that may impact their behavior. In the current research, an overwhelming majority of the participants thought that outside influences was not a factor in crimes occurring on the reservation (n=18). However, some participants felt that outside influences were present in the reservation.

A small group of police officers believed that outside influences were responsible for deviant and criminal behaviors occurring on the reservation (n=5). The narratives offered from these tribal officers suggested that "outsiders" and their culture have seeped into the reservation community. These police officers thought that elements of urban culture were being embraced by some tribal members. For these individuals, they were disheartened that tribal members were accepting outside influences. One tribal officer suggested that American Indians were losing their cultural identity as many of them accepted outside cultural values rather than embrace Native ancestral traits. Some of these officers felt that the Native youth try to mimic culture styles of outside groups.

We have a, like everywhere else, we have that certain element that wants to be gangsters running around with their red bandanas and things like that, so I'm sure that probably comes from TV and music. (Police Officer #21)

This narrative suggests that the Native youth adopt cultural norms that are conveyed through the media. Ironically, in midst of data collection, the tribal police department initiated a task force to provide security to the Indian gaming casino for the upcoming concert of an American rapper.

A well-known American rapper was performing at the Indian gaming casino during the data collection for this research project. Upon questioning the tribal officers about their perceptions about the upcoming concert, they revealed several interesting thoughts. First, some of the participants revealed their disappointment in tribal members adopting elements of urban culture that is connected to rap music. Symbolically, the rapper represented the antithesis of the tribal culture for some of the participants. Indeed, it could be argued that the rapper represented a contemporary form of colonialism as he promoted urban culture and glamorized organized crime lifestyles. Indeed, there has been considerable research on examining the nexus between rap music and social identity (Bennett 1999; Kubrin 2005). The tribal officers demonized urban culture in their discussion about outside influences. A participant remarked that the presence of a rap music artist in the tribal community will cause tribal member to "wanting to dress like black people. Walk around showing the crack of their ass, that type of stuff... well, when they have the rappers come in, all the kids show up, wanna be rappers." (Police Officer #9). As this narrative suggests, tribal members adopted clothing styles that is associated with urban culture and rap music. Although not as critical in the last statement, other tribal officers shared their thoughts on elements of urban culture emerging in the tribal community.

They see so much stuff on the TV, I mean, the saggy pants and the long hair... [the rapper] He's a rapper, you know. And just that type of concert will bring that type of people. Pants down to their...There's a bunch that does that, that was what I was getting at. You know, they see all this on TV, you know, Hollywood, and everywhere, in the cities and they got their hat crooked and they got big old long baseball looking jersey and pants down to their knees and some kind of big fancy looking tennis shoes. I mean, they see that, so that's the way they wanna dress. (Police Officer #1) Once again, this narrative suggests that the rapper and media influence the behaviors and

appearance of tribal members in this tribal community. Simply put, outside cultural

influences symbolize an intrusion of outside groups into the reservation. While these

participants were critical about the rapper and unban culture, one tribal officer offered an

explanation for these negative views toward "outsiders" and their culture.

During a conversation with a tribal officer, he revealed that American Indians

distrust all groups within society. He stated:

Police Officer #15: We have, I don't know if you've heard, but we have a way high crime rate than any of the surrounding counties because all of the surrounding counties are a lot bigger than here and the populations are a lot higher...You can tell once you cross that line...the county line to the reservation. You just know it. People have different attitudes and mentalities. They're just more, I won't say aggressive, but they just have more hatred towards people for some reason. I don't know why.

Researcher: Hatred towards who?

Police Officer #15: Towards each other. And other races. My opinion, Native Americans are more racist than any people in America...See, they're racist against...if you're Cherokee, you're racist against...I don't know if it'd be considered racist because you're from the same race... They don't like different Indians, they don't like black people, they don't like Mexicans, they don't like white people. I don't know why. Most Indians that I know think that they're above everybody, I don't know why. It's just my experience.

This officer suggests that American Indians are ethnocentric and detest other groups. It

could be argued that American Indians continue to feel threaten by "outsider"

encroachment and that this "invasion" jeopardizes the legacy of their Native ancestry and

culture. Another possible reason for these perceptions may rest on "historical unresolved

grief." Brave Horse and Debruyn (1999:60) suggest that "historical unresolved grief,

contributes to the current social pathology, originating from the loss of lives, land, and

vital aspects of Native culture promulgated by the European conquest." It could be that

this deep seated hatred towards "outsiders" by some American Indians rest on the idea that there is still unresolved grief due to the legacy of colonialism. As a result, some of the tribal members and officers demonize these "outsiders." While none of the participants revealed any hatred towards outside groups, some did appear apprehensive towards other racial groups on the reservation.

Within these conversations, the tribal officers discussed the appearance of a particular racial group - Latinos. Some of the participants appeared concerned about the influx of Latinos in the community (n=6). One officer suggested that Latinos were exploiting American Indian women to obtain U.S. citizenship. Indeed, Latinos represented as a threat to tribal members as they are increasingly entering into relationships with tribal women. At least two of the participants voiced their concerns about inter-racial marriages and the legacy of tribal familial lineage. Several of the tribal officers thought that Latinos were responsible for crimes such as prostitution and drug trafficking. In three interviews the participants inquired the ethnicity of the researcher, which is Latino. Indeed, other racial/ethnic groups represent a cultural and economic threat to tribal members. Perhaps this deep seated animosity towards other groups may impact negative perceptions about the upcoming rapper concert. For some officers, outside influences contribute to deviant behaviors. The participants also offered their predictions on the future crime and violence rates.

CRIME PREDICTIONS

Following the discussion on crime and violence, the police officers were asked to predict the future rate of crime for their tribal community. Several themes emerged from this question. First, a majority of police officers thought the crime rate on the reservation would gradually increase over time (n=17). An officer suggested "it's gonna out the roof" (Police Officer #22). The officers identified several factors that may contribute to increases in the crime rate: tourism (n=11), illegal drug use (n=5), and the economy (n=2).

Tourism

A majority of police officers believed that tourism would greatly impact the reservation's crime rate. Police Officer #28 suggested:

If we continue to develop and commercialize, [crime is] gonna go up. That's just a given, it's doesn't have anything to do with being an Indian reservation. Just like I said earlier; anytime there's more people, there's gonna be more crime.

Indeed, the tribal government has steadily improved the community's infrastructure by repaving roads and rebuilding bridges as well as constructing new buildings to attract more tourists. Moreover, the tribal government approved the construction of a nearly \$100 million adventure park which is expected to draw families to the reservation. The adventure park will include a hotel, restaurants, retail shops, and a water park. In addition, the Indian gaming casino underwent remodeling and now includes a hotel with over 1,000 rooms (making it largest in the state), a several thousand seat concert venue, and popular restaurant chains. The casino itself boasts hundreds of video slot machines and live card tables. The combination of gambling and Native heritage sites makes the reservation community a major tourist destination. With these expansions, many police officers thought the reservation's crime rate will increase over time as millions of

individuals will visit the tribal community to enjoy the recreational activities. One participant believed the casino will be a major source for criminality and victimization.

And of course now, our casino always seems to keep growing, which, that's a good thing. We just recently got approved and implemented the live dealers in our casino, so again, that's gonna bring a bigger group of people, different group of people. Because at one point people didn't like coming to our casino because we strictly had video gaming, now it's live so it's bringing a whole 'nother group of people and not only that, as far as marketing within the casino, they are bringing people to our casino to conduct their tournaments, you know, that, I'm saying there's people coming from all over. (Police Officer #4)

For this participant, the introduction of live dealers to the casino will cause a "different group of people" to the reservation community which may impact the crime rate. Once again, the narrative suggests that outsiders are responsible for crime and problems occurring on the reservation. Interestingly, the casino serves as a double-edged sword, on one hand the casino offers a source of income for tribal members; however, on the other, it also brings in "outsiders" with their problems and propensity to commit crimes. The police officers fear that the recent expansions to the casino will only draw more criminal activity to the area. Although the majority of tribal officers believed that the Indian gaming casino would significantly impact future rates of crime, the criminological research remains inconclusive about correlation between gaming casinos and crime rates (Hakim and Buck 1989; Chang 1996; Stitt, Nichols, and Giacopassi 2003). However, research does suggest that areas with higher levels of population density are more likely to experience high rates of crime (Shichor, Decker, and O'Brien 1979; Browning, Byron, Calder, Krivo, Kwan, Lee, and Peterson 2010). Simply put, the officers believed that more people visiting the reservation community would increase the rate of crime.

The reservation community is in short driving distance of a major national park and several major U.S. cities. As a result, the community experiences a heightened level of tourism during the summer months. Over three million people visit the reservation annually. Unfortunately, some of these individuals also bring "their problems" to the tribal community such as alcoholism, gambling, substance abuse problems just to name a few. Unfortunately, these problems also contribute to crime. Taken together, the officers believed that there is a nexus between tourism and crime. More importantly, the majority of officers suggested that the crime rates will increase as the reservation community becomes more commercialized and heterogeneous. Although the tribal officers offered their perceptions about crime, they also provided a discussion on achieving justice within their reservation community.

TRIBAL JUSTICE

The next line of questions explored the officer's perceptions about the tribal court system. To initiate a conversation focusing on tribal justice, the researcher asked the officers to describe their tribal court system. A group of officers described their community's court system as "unique" because it incorporated both Anglo and American Indian philosophies of justice (n=8). Many participants acknowledged that the tribal court system is improving in order to address crime and victimization. With this in mind, the participants were asked whether they thought the tribal court system was functional or dysfunctional. Interestingly, the responses from the police officers were nearly split: 12 police officers thought the court system was functional, 11 believed it was dysfunctional, and 3 suggested that it was both functional and dysfunctional. It is worth mentioning that the majority of non-enrolled police officers described the tribal court system as functional (n=4). The following provides a thematic analysis about the police officer's perceptions about the state of justice within their tribal community.

Functional Efficacy of the Tribal Court

While half of the police officers suggested the tribal court system was functional, they struggled to articulate their reasoning for this decision. One officer simply responded, "Well, I think it's functional. If it was dysfunctional we wouldn't have anybody in jail. (laughs) I mean, that's the way I look at it" (Police Officer #3). Another officer reasoned that the court was effective because of the high level of incarceration sentences.

[I] think that it works. I've seen more people be put in jail and stay in jail. I mean you... have your...everywhere you go you still have your occasional people who come back through the system. (Police Officer #20)

From this narrative, the officer believed that the tribal court system was fulfilling its duty by incarcerating offenders. Several police officers thought the tribal court was progressive than mainstream courts in addressing crime and victimization (n=4). An officer observed, "they're doing things that other, some other Indian reservation tribal courts are not doing, and I think that they're progressive in some ways" (Police Officer #28). Similarly, another participant felt the tribal court offered offenders a "second chance" which does not occur often in mainstream court systems. For this tribal officer, the tribal court relied on other methods of punishment rather than incarceration.

An innovative feature of the reservation's tribal court is its ability to have jurisdiction over non-enrolled members, which is uncommon for many tribal court systems in Indian Country. Through a legal procedure called the St. Cloud Test, nonenrolled tribal members have an opportunity to submit themselves to tribal jurisdiction, but only for misdemeanor offenses. A participant explained the process.

[The Tribal Court does] take non-enrolled members. They have the right to, we do what we call a "Saint Cloud Test" and that's one thing magistrates are responsible

for is that we have to read the questions to them and they have to answer yes or no to the questions. If we have determined that they're non-enrolled then at that point then they have the right to say whether or not they want to come to our court or go through a state court or a different court. (Tribal Judicial Personnel #3)

While this is an innovative feature, others admit that the tribal court system has the ability to improve.

At least two police officers claimed that the tribal court can be improved to effectively address crime and victimization within the community. One participant said, "it works... Yeah, it works. I mean, anywhere you go has got its own hiccups, I mean... It could come out to do better" (Police Officer #26). Similarly, another officer maintained, "our court system does the best that they can. I mean, I myself wouldn't have too much faith in our court system" (Police Officer #4). These officers raised some concerns about the tribal court system; however, they failed to explain the shortcomings of the court. The weaknesses and problems of the tribal court were discussed among participants who described the court system as being dysfunctional.

Eleven police officers thought that the tribal court system was dysfunctional. The basis of these beliefs rested on three primary themes: ineffective tribal judges (n=10), political influence (n=5), and federal restrictions (n=5). The failure of the tribal court system in addressing crime and victimization brought shame for one of the officers.

If I can prosecute outside of it, I will every single time...It's a failure. And I hate to say that about my own court system, it's in my jurisdiction, it should be a pride thing. I should be able to say, "it's the best, I would go there, I would prosecute somebody there," I would be in there and say if it works, but it doesn't. It doesn't. (Police Officer #22)

This statement captures the frustration that many tribal officers felt about the court system. Although the court system represents tribal sovereignty, its inability to effectively address crime within the reservation was seen as a failure. In fact, the ineffectiveness of the tribal court had some participants referring it as a "joke" (n=4).

Several police officers provided vignettes that explained their rationale for this

characterization.

Honest, it's a joke. It's like a... I don't know...well, you just have to go to court and you'll see what I mean. Court's supposed to start at 9:30, judge comes in at 11, ask you how you feel about being there, if you committed the crime, and... I saw a guy come in there one time. I'll give you an example. He didn't show up for court, so they issued a warrant for him. He comes into court and is sitting in there, they had to get a warrant for him, he posted bond, got out, came to court. Well, when they got to court the judge said, "Well, we're gonna continue it one more time." And he looked and he got up and he says, "This ain't nothing but a kangaroo court." And turns around and walks out. (Police Officer #10)

Everybody knows how the court system is. They call it, you know, a circus court. Because you go there and it's just...it's a circus. You go there and it's a joke. Everybody looks at it as a joke. Even the people that are going there. Rarely do you see them nervous or, you know, they just don't have no respect for the court system. They don't have any respect for the judge or anything because they know they're going home that evening...Even the repeat offenders, you know, they know. They know what to say and how to get out. (Police Officer #15)

The above narratives illustrate the leniency of the tribal court system, which they felt did not deter tribal member from committing crimes on the reservation. The statements revealed that tribal offenders have a nonchalant attitude about being tried in tribal court system. The participants revealed that tribal offenders anticipate a lenient sentence as the tribal court usually relies on probation, fines, and/or rehabilitation and offenders are usually allowed to return to their homes. The officers expressed that this persistent pattern of justice administered by the tribal court does little to address crime within their community and does little to deter individuals from committing crimes on the reservation. The participants thought the tribal court system is plagued with many shortcomings. Several officers suggested that the faults of the tribal court rest with the tribal judges.

Tribal Court Judges

The perceived ineptness of the tribal judges emerged as a theme among the police officers (n=11). The participants characterized the tribal judges as being "liberal," "too lenient," "too emotional," and "scared." An officer claimed "I think some of them need to be more stricter than what they are" (Police Officer #6). During a conversation, Police Officer #6 maintained that the judges afforded offenders too many "breaks" which proved to be ineffective in deterring them from committing future offenses. As a consequence of these lenient sentences, offenders repeatedly confront the tribal criminal justice system. For this police officer, the tribal judges contribute to the proverbial "revolving door" with respects to repeat offenders. In these discussions, the participants thought the judges just "slap people on the wrist" for crimes committed on the reservation.

Well, they're too lenient and I think that they, a lot of times, get too emotional and wrapped up with too many people of not wanting to be mean to them or send them to jail because they're afraid of the... what people in the community are going to think about sending someone to jail or... There's a lot of times that they'll be there during a trial and the judge will just stop and say, "I don't think it happened this way. I'm going to let him go." Or just, you know, just "don't do this no more." And slap them on the wrist for the fifth time. (Police Officer #15)

The descriptive narrative illustrated the frustration about the perceived leniency of the tribal judges in rendering punishments. Another participant felt that public opinion largely influences court decisions (Police Officer #13). That is, the judges do not want to experience negative tribal opinion and choose to rely on lenient sentences for tribal members. Likewise, another tribal officer voiced his frustration that the judges would sometimes arbitrarily dismiss cases or change the facts of a case which results in an acquittal or lenient sentence for the offender (Police Officer #15). Similarly, a police officer suggested that the tribal court system is inconsistent as the tribal judges "make

rules up as they go." Indeed, the inconsistency of the court could potentially impede the tribe's ability to administer justice. These reflections are consistent with the general consensus that the police have negative views on courtroom judges (Goldstein 1968). During this portion of the discussions, a couple of police officers appeared to be irritated towards one judge.

At least four participants centered their criticism of the tribal court system on one of the judges.

One's weak... He's not from here. He's a ... guy outta [nearby city]. They hired his daddy and his daddy got him the job here. Now his daddy's retired and we're stuck with him. And we need somebody with a set of balls, and he don't have them. (Police Officer #9)

This narrative centered on both nepotism and outside status in judicial appointments. Another interesting point about this statement is the negative connotation of being an "outsider." Indeed, the tribal judge was not an enrolled member, and some of the tribal police officers question his integrity and self-interests. For example, a participant believed the judge was more concerned about receiving a pay check rather than the impact of his decisions on the tribal community. The source of these negative attitudes about the tribal judge may rest on his Caucasian status. Perhaps being tried before a Caucasian in a tribal court invokes feelings of colonial domination and oppression for some of these participants. That is, the "white man" continues to dictate their lives even under their auspices of their own court system. Another factor that contributes to the ineffectiveness of the tribal court system is political influence.

Political Influence

Another theme that emerged in these discussions was the impact of political influence on the tribal court. At least five police officers thought that political influence jeopardizes the integrity of the tribal court. For these individuals, tribal leaders often interfere with the workings of the tribal court system to secure their position within the tribal government.

Police Officer #9: [It is] tribal politics. If you get old so-and-so's buddy, "oh hell, we'll make a phone call, we'll get it took care of." It gets took care of. I've saw a boy get a DUI and get it PJC'd – prior for judgment continuance- and turn right around and get another DUI and that one gets prior for judgment continuance. Nothing happens... Nothing. Fuck, you gotta be kidding me...It's politics, man. As soon as it goes tribal, the chief's over the judge; he hires and fires, so they're gonna do what he wants.

Researcher: There's a lot of politics involved?

Police Officer #9: Yeah. "If you want me to vote for you, help my boy out here." "Alright, we'll get him probation."

Tribal government leaders influence tribal court judges to dismiss cases, reduce charges,

or render lenient sentences for their constituents. According to this participant, tribal

leaders wield considerable power as they have the ability to dismiss people working

within the tribal government. In a non-recorded interview, a participant revealed that

tribal employees have been demoted because they disagreed with these tribal leaders.

Political influence extends among tribal leaders.

The biggest thing is take the politics out of the court, where the chief of the tribe has no influence over the courts, council has no influence over the court, now they always say, "Well, we don't have no influence." But they do. There's been numerous times that officers, including me, has wrote either a council member's son or somebody tickets, they never even... it never even gets to court. So, I mean, we know it happens, we can't never prove it...just because who they are don't make them no better than nobody else. (Police Officer #13) Within this narrative, the participant believed that the major problem confronting the tribal court is political influence that leads to corruption of the tribal justice system. The tribal officer suggested that traffic violation tickets are not upheld in tribal court because of tribal politics. A couple of the tribal officers also revealed how court proceedings are delayed months at a time due to political influences. Participants also recounted the impact of politics on the functions of the tribal police department.

A tribal officer thought that the tribal government distorted police department information to suit their political needs. That is, tribal leaders fabricate and circulate falsehoods about a particular crime or event for political purposes. Another police officer recounted that tribal leaders would make suggestions to him on how to charge tribal members for their crimes. Indeed, tribal leaders try to exert their influence on the tribal police department. In fact, due to political influence the tribal police department had to change some of their policies. For example, the tribal police department initiated a policy of banishment, which forces individuals off the reservation for engaging in destructive behaviors. Although the police department used banishment on enrolled tribal members for their destructive behaviors, the tribal government ended the policy due to political objectives. For tribal government leaders, the banishment of tribal members equates to the loss of votes in future tribal elections.

Perhaps the influence of politics was most tangible in the recent tribal elections. In another non-recorded interview, two individuals reported that one of the candidates provided tribal members with gifts such as washers and dryers and offered to pave driveways to secure votes. Given the close-knit nature of this tribal community, it is not surprising that political power is influential in the disposition of tribal offenders. Indeed, tribal leaders have monopolized the tribal justice system to maintain their political power within the reservation community. That is, tribal leaders circumvent tribal and state law to sustain their political power. Without a doubt, political influence interferes with the tribal court system, but federal restrictions also impede its ability to address crime and victimization.

Federal Restrictions

A small group of participants thought that federal restrictions obstructed the tribal court's ability to prosecute offenders for crimes committed on the reservation (n=5). As previously mentioned, tribal courts are restricted to trying misdemeanor offenses and tribal judges possess limited sentencing powers. Moreover, the race of the perpetrator also dictates which jurisdiction assumes control of the crime.

If the incident happened here (on the reservation), if it's a non-Indian involved, if an Indian's involved, it has to go federal. Because an Indian can't charge a nonenrolled member in our court system, they can't charge them in the state, so it has to go federal. And then as far as state, as long as it's non-Indian, non-Indian, we can charge them in state all day long. Now if its' Indian Indian, we can do tribal court all day long. That's the only problem we have is when it's a non-Indian Indian match up thing. And if the non-Indian is the aggressor in this matter, it has to go to federal court. (Police Officer #6)

The above narrative explains the levels of government that have jurisdiction in Indian Country. Within this statement, the tribal officer revealed that jurisdiction relies on race of the victim and offender. Indeed, it is a complex web of various jurisdictions that continue to puzzle many within the tribal community. While a majority of tribal courts do not have jurisdiction over non-American Indians for misdemeanor offenses, this tribal community has that power. As mentioned earlier, the tribal court has jurisdictional power over non-American Indians through the St. Cloud Test. While this appeared to be an innovative trait of the tribal court, a group of tribal police officers admitted the practice draws offenders to the reservation because the tribal court system is limited with respect to sentencing. Under the Indian Civil Rights Act of 1968, tribal courts were restricted from imprisoning an individual for no more than a year and a fine of \$5,000 (Pevar 2004). However, the recent Tribal Law and Order Act of 2010 increased these penalties to three years imprisonment and \$15,000 fines. The recent modifications still limit their sentencing options in comparison to mainstream courts. In fact, the restrictions of the tribal court are another reminder that American Indians continue to suffer from colonial domination. While the tribal court is seen as an avenue to enhance tribal sovereignty (Washburn 1971; Lester 1991), the federal government has severely restricted the powers of the court system (Pevar 2006). Unfortunately, non-American Indians take advantage of the limited powers of the tribal court.

A small group of police officers believed that non-enrolled tribal members realize the limited powers of the tribal court and view the reservation as an opportunity to engage in criminal activities (n=4). That is, non-enrolled individuals exploit the weaknesses of the tribal court. An officer remarked that "non-enrolled people, if they get caught on the reservation, they can submit to tribal jurisdiction, which they'd be a fool not to. Because nothing happens" (Police Officer #9). Although President Obama authorized the New Tribal Law and Order Act of 2010, the tribal court system has yet to take advantage of the new sentencing powers. That is to say, offenders confronting the tribal court system can be incarcerated up to a year in prison or a \$5,000 fine. Another police officer provided another incentive to commit crimes on the reservation.

We have a misdemeanor court which means no felonies can go there, however what you construe as a felony in regular country, in non-native country it would actually dwindle down to a misdemeanor here. So when they ask, "do you want to be under our jurisdiction" if it's somebody that's not enrolled or not a descendant, then it's good for them to say, "yeah I submit." Because basically that felony for possession of that meth that you have in your pocket won't be a felony on you, it goes to misdemeanor court, which our tribal court is. (Police Officer #9)

For some tribal officers, they perceived that the biggest problem for this court system is that felony crimes are reduced to misdemeanor offenses in the tribal court. The participants realize that drug traffickers realize the limited powers of the tribal court system and continue to distribute illegal drugs on the reservation. For some these individuals, they would rather be caught on tribal land than local or state jurisdiction because the penalties for trafficking narcotics are much lower in the tribal court system. Ultimately, the disjunction between tribal and federal law allows drug traffickers and serious offenders to evade justice by accepting tribal jurisdiction where the penalties are less severe. While the St. Cloud Test appears to enhance the sovereignty of the tribe by allowing their court to have jurisdiction over non-enrolled members, it is creating a safe haven for offenders to engage in criminal activities without fearing any real repercussions for their actions. Another problem with the administration justice in Indian Country is the federal government's refusal to prosecute serious crimes that occur on the reservation.

Two police officers suggested that the failure to achieve justice rests with federal prosecutors. Recall that serious tribal crimes (i.e. murder, rape, and aggravated assaults) are prosecuted by federal prosecutors in U.S. District Courts (with the exception of both parties being non-Indian which are tried in state court). Unfortunately, federal prosecutors do not always try these crimes in the federal system. The police officers believed that the refusal to try these crimes in court stem from racist beliefs.

In general, whenever an Indian gets... Two Indians shoot and kill each other, or let's say, two brothers or family members or... it doesn't even have to be family

members...But if you've got two enrolled members that got into a fight and one of killed the other one, you know, it's like the government don't look at that as being... I don't know what I'm trying to say here, but there was a term that people use around here and that would go like something like, "It's just an Indian killed another Indian, it's not like the federal government's gonna worry about that." And that's been pretty much true. I guess what I'm trying to say is that the government don't look at it like as it being a major issue for them, you know? The way Indians have been treated in the past...Historically. But, now if that Indian was to go and kill a white person, then they're gonna nail...the government's gonna nail that Indian person. I don't know what you could call that, but that's just an example I'm using, you know? (Police Officer #2)

The above narrative is consistent with previous research on the lack of federal

prosecution for serious crimes taking place in Indian Country (Riley 2007a, 2007b; Eid

and Doyle 2011). It also supports criminological research on the over-prosecution of

American Indians involved in serious crimes, especially if the victim was white (Baker

2007). For many of these participants, they felt frustrated that serious reservation crimes

are not treated with the same urgency as those occurring outside of Indian Country. In

fact, one of the tribal officers suggested that the federal government has adopted a casual

attitude about prosecuting serious crimes involving American Indians in the federal court

system. Another officer recounted his experience with a child abuse case.

Police Officer #22: I used to be a detective...We had a man that could have been sentenced to seventy-five years and he was allowed to plea to three [in federal court]. To me, you know, and that was an Indian victim and a non-Indian perpetrator. To me, that's not okay. If I would have prosecuted that man in state court... he would got the full blown seventy-five years, and would not have got to plead. There was forensic evidence, he confessed.

Researcher: But he went through the tribal court?

Police Officer #22: He went through federal court...Mm hmm. And the US DA allowed him to plead to three years when he should went for seventy-five. You see what I'm saying?

The above narrative illustrates the lack of justice that is achieved from American Indians when their cases are heard in federal courts. Although evidence suggests that the offender committed the crime, the federal court still rendered a lenient sentence for a serious offense. This police officer appeared frustrated that non-American Indians evaded punishment for abusing an enrolled member of the tribe. In addition, he revealed that the federal prosecutor is selective about cases to try in federal court in that he only takes "slam dunk" cases that will result in a conviction. For many American Indian victims, justice remains elusive, but for American Indian offenders who receive harsh treatment under the auspices of the federal court system

The ill treatment of American Indians in the federal court system was revealed by non-tribal police officers. In fact, two participants provided a discussion about American Indians who have been tried in the federal court system. They revealed a federal court system that infringes on the rights of American Indians. One of them suggested "when I was growing up they treated Indian bad in courts. Bad." (Tribal Judicial Personnel #1), but he maintained that the federal courts are still racist towards American Indians. According to the participant, people in the neighboring communities are resentful towards American Indians, because of the money that is generated by the Indian gaming casino and other entitlements that are given to them. These negative feelings towards American Indians impact their fate within the federal court system. Due to the inception of Indian gaming casinos, "outsiders" have the perception that American Indians are "rich Indians" meaning that they are wealthy from the income that is generated at these facilities (Churchill 2004; Lujan 2006). Unfortunately, this negative stereotype contributes to feelings of resentment towards American Indians which may be an influential factor in their disposition in the federal court system, especially for criminal trials with juries. For many tribal officers, the federal court system does not afford

American Indians equal rights. The combination of a restricted tribal court system and an ineffective federal prosecutor, the participants suggested that many tribal members are frustrated by the lack of justice in their community.

Frustration with Tribal Justice

A group of police officers revealed that tribal community members were frustrated and discouraged about the lack of justice for enrolled tribal members (n=7). As police officers, they often hear complaints about the ineffectiveness of the tribal and federal court systems. An officer noted that "I've heard a few people say stuff, wanting to know why it's not done, and then you have to sit down and explain to them" (Police Officer # 18). Another officer said, "Yeah, you hear it in the community... we hear the frustration...mothers and grandparents seeing sons and daughters, grandchildren who are repeatedly going through there and they're not doing nothing to them" (Police Officer #9). At community-police meetings, the officers are frequently asked about the lack of justice in their community. Criticism of the tribal criminal justice system also extends to the police department.

Some of the officers revealed that the tribal police department receives criticism about the lack of justice within the community (n=4). Police Officer #18 stated "they blame the PD for them not doing anything, and it's not the PD." Tribal police officers represent the tribal criminal justice system. Although they are responsible for enforcing the law and arresting perpetrators, they receive criticism over the failures of both the tribal and federal criminal justice system in addressing crime and victimization within their community. Upon being asked about whether the criticisms impacted his job, one officer replied:

It did at first. But now, I mean, that's...I do my job and I charge them and I do all my case files and do as much as I can. So I know I did my job. It is frustrating because it's a waste of my time a lot of the time. But as long as I can get them off the street for that night for whatever they're doing I know I've done my job. So I'm not frustrated with them. (Police Officer #15)

This police officer felt that he performs his policing duties to the best of his ability. Although the tribal court system is ineffective in reducing crime, he maintained that his tasked is to ensure the safety of the people in the community. This perception of the lack of justice within the tribal community has led tribal members to informally resolve conflicts.

INFORMAL RESOLUTIONS

The lack of faith in both the tribal and federal court systems has led to informal conflict resolutions, which sometimes result in bodily harm or the loss of property. Only three police officers reported that tribal members engage in vigilante justice to avenge the actions of a perpetrator. A police officer remarked about the use of vigilante justice in his community;

Police Officer #13: I think the response of the systems that are set up to assist fail here than they do in other areas. So, with that failure comes a response of a tidal wave, like the Hatfield and McCoy's sometimes. "You beat up my daughter, I'm gonna beat you up." So it's kind of a, it's a... you'd be amazed, you'd just be simply amazed... you can walk up on a DV situation between two sisters, you go to arrest one of the sisters, that sister turns on you for assisting her, who was previously the victim, but then becomes the perpetrator against law enforcement because you're taking her sister to jail. So, it's kind of... it's... it's a delicate situation and...Very more family oriented. Almost like the Hispanic race. They have that whole tight-knit family community thing? Researcher: So it's common to see somebody walking around with a black and blue eye because...

Police Officer #13: Oh yeah, that's common.

For this officer, there is a strong bond within Native families which facilitates the use of informal conflict resolutions. Another police officer noted that perpetrators not tried in court for their crimes will receive an "ass whipping" which is seen as a "little paycheck" (Police Officer #9). The officers also noted that the perpetrator who received the "ass whipping" declines to file a report about his victimization. The problem of the tribal court system perpetuates the use of this type of resolution to solve disputes among tribal members. Despite discussing crime and justice problems, the participants provided their ideas and thoughts about addressing and responding to crime and violence within the reservation community.

FACTORS PERCIEVED AS HINDERING POLICE EFFECTIVENESS

The tribal officers provided recommendations to improve the tribal criminal justice system and quality of life within the reservation community based on what they perceived to be a hindrance to their effectiveness and community success. To elicit suggestions from the tribal officers, the researcher asked the question; "if money was no object, how would you improve the lives of the people in the community." The participants offered several suggestions such as providing the tribal police department with more resources (n=7), establishing more substance abuse programs (n=4), restricting entitlements to tribal members (n=4), creating family oriented programs (n= 3), restructuring the tribal court system (n=3), and improving the tribal education system (n=1). The majority of tribal officers envisioned a tribal community with more police

officers and equipment. The participants suggested that the police department is understaffed and lack the resources needed to address crime and violence. A couple of participants revealed that there could be one or two tribal officers on patrol at any time on the reservation – this police department services a population of 15,000 across over 80 square miles of tribal land. As a result, they argued for more police department resources. Interestingly, a couple of tribal officer believed that the police department has a considerable amount of policing equipment compared to the neighboring municipality police departments. In fact, these neighboring police departments often call upon the tribal police department for assistance or the use of their policing equipment. During data collection, the researcher took note of the relatively new fleet of Dodge Charger police cars, a patrol Segway, and an armored S.W.A.T team vehicle. Several participants also noted that the majority of the tribal officers are trained to use taser weaponry. Indeed, the policing equipment illustrates the militarization of tribal police department to address crime and violence. Since the "war on drugs" and even the current "war on terrorism," domestic policing in the United States have become more militarized (Andreas and Price 2001; Kraska 2007). That is to say, the police rely on military equipment, strategies, and tactics to subdue real and perceived "enemies" (Kraska and Cubellis 1997; Kraska and Kappeler 1997; Falcone, Wells, and Weisheit 2002). Despite the militarization of this tribal police department, the quality of policing equipment was lauded among the participants. As one participant pointed out:

We have the best equipment, we have the best uniforms, we have the best patrol cars, we have the best of everything. And then, I mean, what our sheriff's department's got... some officers here have never had to work for a county office. And I mean, it's a whole different issue. (Police Officer #26)

Despite having many resources and equipment, several tribal officers thought that more resources and police personnel were needed to adequately address crime and violence. Other participants thought the tribal community should focus their energies in addressing substance abuse problems.

Given that a substantial amount of reservation crimes are drug-related, four of the tribal officers advocated for more substance abuse treatment programs. As previously mentioned, a majority of the tribal officers felt that illegal prescription drug use was largely responsible for property crimes. As a result, some of the participants suggested that substance abuse programs should be implemented to reduce the prevalence of crime in their community. Another participant revealed that the tribal school's Drug Awareness Resistance Education (D.A.R.E.) program is ineffective and does not adequately address substance abuse problems. While a couple of participants believed that the tribal government implement substance abuse programs in the community.

At least three participants felt that the tribal community did not offer enough family orientated programs or initiatives to offset the everyday mundane life of living in a rural community. Indeed, a geographically isolated community typically lacks activities that are found in large urban areas. As a result, tribal members are left to their own devices to address the monotony of living in a reservation community. Although the tribal community has an Indian gaming casino, a tribal police officer believed that it was not adequate for many tribal families. In fact, the individual advocated the implementation of culturally sensitive programs. Another participant suggested that the tribal community should establish parenting classes which may reduce delinquency and criminal activities. Although some participants thought that family orientated programs would improve the lives of tribal members, some argued that the tribal court system needs to be restructured to meet the needs of tribal members.

Recall that nearly half of the participants believed that the tribal court system was dysfunctional and did not adequately address crime and violence within the reservation. At least two tribal officers conveyed their thoughts about restructuring the tribal court system. An officer stated that "we [tribal police department] need more help from the court. I mean...the court is our weak link" (Police Officer #1). Another tribal officer suggested that the tribal court should be "put back to CFR Court [Courts of Federal Regulations], get federal judges coming back in here, federal prosecutors" (Police Officer #9). This participant felt that CFR courts are more reliable than the current tribal court system. The Department of Interior established CFR Courts in 1883 to resolve less serious crimes and disputes among tribal members. Typically, federal judges preside over CFR courts. CFR courts were criticized on two principles: 1) the lack of American Indian judges, and 2) the court was viewed as an extension of the federal government American Indian assimilation policies. As a result, many tribal governments established their own court system to represent the traditions and cultures of their tribe. Given the pervasiveness of tribal politics, CFR courts can be seen as a remedy to diminish the influence of politics. However, critics would argue that these courts fail to represent the interests of American Indians.

128

SUMMARY OF FINDINGS

In sum, the tribal officers suggested that the general rate of crime in their community was comparable to the neighboring communities. However, the participants revealed the tribal community suffered from high rates of illegal prescription drug use, which they thought was responsible for crime and violence, especially property crimes. The tribal officers also discussed violent crimes that occur in their community such as domestic violence and assault. The majority of the participants felt that domestic violence was the most common type of violence to take place on the reservation. Unfortunately, many of the participants thought that the reservation's crime rate will increase due to the recent expansion of the Indian gaming casino and tribal infrastructure because it will draw more tourists to the community.

Several major themes emerged within these conversations with tribal officers. First, the majority of tribal officers believed that their reservation community's crime rate was similar to neighboring communities. Indeed, the participants believed that their reservation community was similar to the small towns in rural America. More importantly, they felt that their community's problems were also similar to mainstream communities too. While they rejected any thoughts equating their reservation community to a "war zone," the participants suggested that the analogy was reflective of other reservation communities' experiences crime and violence.

Illegal prescription drug use emerged as a major theme among the tribal officers. Given the recent national illegal drug use trend in the U.S., it is not surprising that this form of substance abuse had filtered into the tribal community. Participants believed that all age groups were abusing this drug and that tribal members frequently attended "pill parties" where they consumed copious amounts of illegal prescription drugs. Also, tribal members obtain these drugs through drug traffickers and/or resort to property crimes to secure prescription drugs. In addition to illegal drug use, breaking and entering crimes emerged as a dominant theme among the tribal officers that they occur frequently in the community, and that these crimes were in response to individuals suffering from prescription drug addiction. The participants also appeared disheartened over the theft of Native crafts and ancestral heirlooms which are often sold to tourists to garner funds for illegal drugs. The tribal officers also acknowledged the many forms of violence that occur in their community.

Domestic violence, which is frequently triggered by alcohol and financial hardship, surfaced as the most common type of violence on the reservation. Because of the pervasiveness of domestic violence in the community, the tribal government established several programs to assist victims and educate tribal members about this type of violence. A small group of participants admitted that there is a high rate of male domestic violence victims because "[Native] women that will fight back" (Tribal Judicial Personnel #1). As a result, more male victims have recently filed domestic violence charges against female perpetrators in the tribal court system. Assaults also emerged as a common type of violence that often takes place within the community. According to the participants, these crimes are alcohol related and tend to be more violent than assaults occurring in non-tribal communities. For some participants, assaults are part of the "tribal norm," which causes reluctance among the victims to report assaults and the police department is unable to gauge the actual extent of these crimes. Although the participants provided a discussion about the characteristics of crime and violence, they also revealed their beliefs on crime causations.

The participants were split about the functional efficacy of the tribal court system. Although half of the participants felt that the tribal court system was functional, they were unable to articulate the reasons for their decision. On the other hand, tribal officers who thought that the tribal court system was dysfunctional provided several reasons for their assessment. These individuals cited issues with the tribal judges, political influence, and federal restrictions as hindrances in achieving justice within the tribal community. In addition, the tribal officers believed that tribal members had lost faith in the tribal court system and sometimes tribal members have resorted informal conflict resolutions which often result in bodily injury.

CHAPTER VII DISCUSSION

This study investigates the perceptions of crime and justice among tribal police officers. Similar to their mainstream counterparts, tribal police officers are on the "frontlines" when addressing crime and violence in Indian Country. They are the 'experts' with the ability to diagnose" all crime related issues (Loader 1997:3). Drawing on these concepts, the study explores the dynamics of crime and justice within a southeastern reservation community. In-depth interviews with 27 tribal police officers were conducted to answer the research question: What are the perceptions of crime and justice in Indian Country among tribal police officers? This chapter provides an analysis of the findings that is grounded within the theoretical frameworks of Loader's (1997) symbolic power of policing.

RELEVANCE OF THEORETICAL FRAMES

As Loader (1997:3) observed, the police have the "power to diagnose, classify, authorize, and represent both individuals and the world." Given their position on the reservation, tribal officers are official the "street level" experts on all crime and justice related matters within the community and have unique insight to and opinions about the dynamics of crime and justice on the reservation.

The conversations with the tribal police officers revealed a couple of overarching theoretical assumptions: colonialism and Loader's (1997) symbolic power of policing. First, colonialism remains intact for individuals living in Indian Country. That is, American Indians continue to have the misfortune of being socially, politically, and economically oppressed. Although this study did not explicitly investigate the role of colonialism in relation to crime and justice in Indian Country, it emerged as a significant factor for those interviewed. From these discussions, the narratives revealed that colonialism continues to impact every aspect of life on the reservation from substance abuse problems to the federal restrictions placed upon the tribal court.

COLONIALISM

Scholars argue that colonialism subjects American Indians to "internalized oppression" (Freire 1968), which may explain their high rates of substance abuse, homicide, child abuse, and domestic violence (May 1987; Bachman 1992). That is, the historical legacy of colonialism in the form of historical trauma and cultural oppression contributes to "historical unresolved grief" (Brave Heart and Debruyn 1998), which explains their destructive behaviors. Indeed, tribal police officers revealed the tribal community suffered from high rates of prescription drug use and domestic violence. With respects to domestic violence, Smith (2005) suggests that the historical marginalization of American Indians in the form of economic deprivation and substance are also the leading factors that perpetuate domestic violence among Native women. With respects to the current research, tribal police officers felt that domestic violence on their reservation stem from alcohol abuse and financial hardship. Colonialism also emerged in the context of tourism and the Indian gaming casino.

Tourism

The reservation community thrives on the tourism industry. Indeed, the tribal government has made great strides to lure tourists to the reservation by improving the tribal infrastructure, expanding the Indian gaming casino, and allocating funding for the construction of a water park. While these initiatives are seen as methods to generate funding for the reservation, it also proves to be a source of contemporary colonialism. During data collection, the researcher noted that many tribal members adhere to stereotypical American Indian depictions that are propagated through the media. In some cases, tribal members manipulate and reinforce Indian stereotypes for tourism and commercialism. For instance, some tribal members would perform ceremonial dances and rituals for the tourists with the hopes of making "donations" or tips for their performance. Upon being asked about these performances, a tribal officer noted that these rituals and dances were not part of the ancestral history of the tribe. Unbeknownst to the tourists, the performance they were observing was not part of the ancestral history or culture of the tribe. For these tourists, they believed that the event was authentic because it "looked Indian." Arguably, many Americans are ignorant about the American Indian culture. Given that American Indians are isolated – socially and geographically – from mainstream society, general knowledge about them is often derived from the media and popular culture. In several unrecorded conversations, the tribal police officers revealed that they often direct tourists to authentic Native artisan crafts shops on the reservation, because several other stores within the community sell fake crafts that are manufactured off the reservation, more specifically China. Unfortunately, many tourists are frequently "too ignorant to recognize real Indian culture when they stumble across it"

(Evans-Pritchard 1989:91). Although tourism is openly embraced within the tribal community, several of the tribal police officers felt that the future dynamics of crime on the reservation is contingent on the rate of tourism. For these participants, the rate of crime will increase over time due to the new tribal initiatives to boast the rate of tourism. Another contemporary source of colonialism that materialized in the findings centered on the Indian gaming casino and its problems.

Indian Gaming Casino

The Indian gaming casino emerged as an influential force for many people living on the reservation. During the interviews, the tribal officers revealed that the casino was a source of income for the tribal community from the bi-yearly "per cap" checks to funding the daily operations of the tribe. For some, the Indian gaming casino symbolically represents colonialism (Porter 2001; Cramer 2006). During the 1980s, the Reagan administration urged tribes to become more self-efficient, and suggested that they turn to the private sector to generate funds to address their social problems - poverty and health problems - that were occurring in Indian Country. As Oakley (2001:139) notes, "Reagan encouraged Native Americans to pull themselves up by their bootstraps, as the federal government started dismantling many Indian programs." Facing limited resources, many tribes turned to the gaming industry to generate revenue to address budget shortages. Although Indian gaming casinos are viewed as enhancing tribal sovereignty (Cornell and Kalt 1998), scholars argue that these enterprises are eroding tribal culture, customs and norms (Cramer 2006) and diminishing tribal sovereignty (Mezey 1996; Cramer 2006; Fenelon 2006). While the participants thought the Indian

gaming casino was affording several benefits to the tribal community, they did reveal some problems associated with it. For example, the tribal officers revealed that tribal members squander their "per cap" checks on illegal drugs and alcohol as well as on other materialistic goods, rather than improving their standards of living within the community. That is, tribal members rely on the stipends to fulfill their fetishes of consumerism and materialism, which is projected through the dominant society. Additionally, the tribal officers also believed that the casino was responsible for some crimes taking place in their community. Through their "police voice," the tribal officers suggested that the Indian gaming casino attracted individuals suffering from addiction or with the propensity to commit crimes to the reservation. With the recent expansion to the casino, the tribal officers felt that the crime rate will gradually increase over time. Another form of intrusion also emerged within the context of the tribal court system.

Tribal Court System

The tribal court system can be seen as symbolical form of tribal sovereignty; however, the federal restrictions placed upon the court are viewed as another form of colonial domination. Lester (1991:133) observes the agreement between tribal governments and the federal government:

Since American Indians have a special legal relationship with the American government, the federal government can prevent state jurisdictional encroachments. On the other hand, the federal government can also intervene and undermine tribal authority.

This agreement was captured during the conversations with the tribal police officers. They revealed that their tribal criminal justice system operates on different levels: tribal, state, and federal. Although the federal government protects tribes and their members from state jurisdiction, the government has the authority to dictate how tribes can operate their court system (Pevar 2006). Through a paternalistic approach, the federal government has limited the powers of the tribal courts. In fact, these courts are restricted to hear misdemeanor offense and limited sentencing powers. As a result, tribal communities are unable to adequately address crime and secure justice for American Indian victims in Indian Country. In this study, several tribal police officers reported that their tribal court was ineffective due to federal limitations. While tribal courts struggled to administer justice through these limitations, the participants also believed that the federal government also fails to address crime and victimization in Indian Country.

Federal Court System

In a broader context, the tribal officers also revealed that American Indians continue to suffer from colonial domination. The participants provided anecdotal accounts of injustice experienced by tribal members under the auspices of the federal court system. The tribal officers spoke about the failure of the federal government to investigate and prosecute serious crimes occurring on the reservation. On one hand the federal government has taken a nonchalant attitude towards assisting tribal victims, on the other, they aggressively prosecute crimes when it involves an American Indian offender. Thus, the treatment of tribal members in the federal court system illustrates the subjugation of American Indians. As Eid and Doyle (2011) note, the federal government has consistently administered a "separate but unequal" style of justice among American Indians. Unfortunately, the government's role in administering justice often infringes on Equal Protection rights of American Indians. For Eid and Doyle, the criminal justice system systematically discriminates both American Indian victims and offenders as they are not afforded their constitutional rights. In fact, the establishment of federal jurisdiction in Indian Country was critical in the government's effort to assimilate American Indians into the Anglo culture (Eid and Doyle 2011). The under and over of prosecution of American Indians in the federal court system suggests that American Indians continue to be socially marginalized and victims of colonial domination. Colonialism continues to impact American Indians in Indian Country. The results of this research also illustrates Loader's (1997) concept of the symbolic power of policing by positioning themselves within the discourse pertaining to crime and justice.

SYMBOLIC POWER OF POLICING

The tribal officers provided in-depth discussions about the dynamics of crime and justice within their reservation community. During these conversations, the participants "legitimately named" individuals as being criminal and also demonstrated how policing on the reservation reveals "stories" about the tribal community. Loader (1997) in his attempt to "illustrate the symbolic power" of policing, asserts that the police enjoy coercive and symbolic powers, that allow them to shape ideas and perceptions about crime related issues. It is within their position, that police have the "power to diagnose, classify, authorize, and represent both individuals and the world, and to have this power of 'legitimate naming' not just taken seriously, but taken-for-granted" (Loader 1997:3). More importantly, he contends that the police serve as cultural producers who are able to produce, reinforce, and circulate societal norms and expectations. With respects to crime, the police are "experts" on all matters related to crime and justice. In conjunction with

coercive tactics, the police also rely on the "police voice," which grants the privilege to "in constructing representations of crime events" (Loader 1997:10). For Loader, the police wield considerable influential power as they are able to control the discourse on crime. The tribal officers' comments illustrated several aspects of Loader's symbolic power of policing. Indeed, the tribal police officers revealed themselves as cultural producers in their quest to address crime and violence in their community.

Cultural Producers

Drawing on the work of Turner (1974), Loader (1997) argues that the police are condensation symbols. That is, they are "cultural producers" whose position allows them to produce, reinforce, and circulate social meanings within society. In the current study, the tribal police officers manufactured and distributed their narratives about their struggle to win the "war on drugs." The participants revealed that their community was suffering from substance abuse in the form of prescription drug use. Indeed, the tribal police officers provided "war stories" about intercepting trafficked drugs, arresting illegal drug users, and reiterating the harmful effects of substance abuse. To transform the term "street-level bureaucrats" within the present study, the findings suggest that tribal police officers are "rez-level bureaucrats" with the mission to produce, reinforce, and circulate the anti-drug message to individuals living in Indian Country. For many of the participants, their conversations focused exclusively on the frequency and intensity of illegal drug use within the tribal community. Unfortunately, many of the individuals overlooked the structural or social factors that may contribute to substance abuse. In fact, some of the participants suggested that substance abuse was conducive to individual-level

factors such being unemployed. Loader would argue that the tribal police officers used the interviews as a platform to discuss the emerging threat of illegal drugs among American Indians. That is, the participants provided a discourse about the current state of the war on drugs in Indian Country. More importantly, they used the occasion to galvanize the need for tribal police and more policing equipment to fight drugs and crime on the reservation. It is within these discussions that the tribal police officers were able to frame individuals as criminal or the "other." The tribal police officers' perceptions on illegal drug in their community is consistent with previous American Indian substance abuse research. As previously mentioned, the AI/NA suffer from extraordinary rates of illegal drug use and alcoholism (Snipp 1992; National Institute on Drug Abuse 2003; Substance Abuse and Mental Health Services Administration 2010). Recent figures also suggest that the AI/NA population has experienced a gradual increase in the consumption of non-prescription drugs (National Congress of American Indians 2006). Accordingly, the majority of tribal police officers believed that the reservation community suffers high rates of non-prescription drug use.

Legitimate Namers

Loader (1997:5) asserts that the police have the power to legitimately name individuals or groups of people as criminal or deviant. The tribal officers in this research identified two groups that were responsible for the majority of crimes within the tribal community: prescription drug addicts/youth and "outsiders." A majority of tribal officers believed that illegal prescription drug users were responsible for a substantial amount of crimes on the reservation. The participants strongly suggested there is a direct correlation between illegal drug addiction and criminality. For these participants, young tribal drug addicts commit predatory crimes (i.e. breaking and entering, shoplifting, and larceny) to fulfill their addiction needs. Some of the officers framed these individuals as a serious threat to the reservation.

The officers' narratives are reminiscent of Bennet et al's (1996) concept of the "superpredator." Bennet and colleagues (1996: 43) assert that "America is now home to thickening ranks of juvenile 'superpredators' -- radically impulsive, brutally remorseless youngsters, including ever more preteenage boys, who murder, assault, rape, rob, burglarize, deal deadly drugs, join gun-toting gangs and create serious communal disorders." Several participants identify the tribal community as "home to thickening ranks of juvenile superpredators" (Bennet et al 1996: 43), whose destructive behaviors are contributing to crime and violence. Indeed, tribal officers framed younger members and drug addicts as "troublemakers" who engage in crimes. These perceptions are consistent with research focusing on youth involvement in Indian gangs in Indian country. Indeed, Indian gang related activities has emerged as pressing problem on many reservations. Within the last twenty years, many Indian reservation communities witnessed an increasing rate of Native youth joining Indian gangs (Hailer and Hart 1999; Joseph and Taylor 2003; Pridemore 2004). There are nearly 400 gangs with over 4,500 members operating in or in close proximity to reservation communities (Pridemore 2004). Statistics also suggest that Native youth are more heavily involved in gang involvement than their non-AI/NA peers; 15% of all Native young individuals are involved in gangs while only 6% of African-American and 8% of Latino youth participate in such organizations (Donnermeyer, Edwards, Chavez, and Beauvais 2000;

Glesmann, Krisberg, and Marchionna 2009). Native adolescent males between the ages of 11 and 18 comprise most Indian gang members (Hailer and Hart 1999). While many non-Indian gangs are motivated by economic interests, Indian gangs are less likely to have economic goals (Joseph and Taylor 2003). Rather, boredom and monotony fuels their involvement in crimes that range from aggravated assault to drive-by shootings. The high level of Native youth gang involvement has also impacted crime rates on Indian reservations. According to the Department of Justice, "anecdotal reports and official records from juvenile justice officials (i.e. tribal courts and probation and/or law enforcement officers) in a number of Indian country communities indicate increased levels of crime associated with youth gangs" (Major, Egley, Howell, Mendenhall, and Armstrong 2004). In the same study, over 20% of American Indians reported that their community had experienced youth gang activity although much of it centered on property offenses such as vandalism and graffiti. Aside from Native youth delinquency, the tribal police officers provided their perceptions about non-American Indians and their involvement in criminal activities within the reservation.

Many tribal officers also believed that "outsiders" were responsible for crimes and acts of violence within the community. In several conversations, the participants appeared to be apprehensive towards Blacks and Latinos. Perhaps this observation can best be explained under racial threat theory. As Breunig and Rose Ernst (2011:235) note, "this theory posits that the presence of a racialized 'other' in a population increases fear and/or hostility among White Americans toward this other group which, in turn, provokes support for social control policies." The presence of non-American Indians within or in the area surrounding the reservation caused some participants to feel panic and/or anger. It could be possible that these feelings contributed to the tribal officers' framing of "outsiders" as being criminal or "troublemakers."

Although tribal officers felt that outsider influences such as cultural influences and the media did not impact the behaviors of community members, they were critical about the upcoming rap concert on the reservation and expressed disappointment that tribal members adopting elements of urban culture. These participants demonized urban culture in their discussion about outside influences and suggested that urban culture greatly impacts the appearance, attitudes, and behaviors of tribal members. The tribal officers equated urban culture to criminal behaviors such as gang membership and abusing illegal drugs. As a result, the participants framed tribal members as being criminal and/or deviant who adopted elements of urban culture. Perhaps this framing is in response to a perceived threat among tribal officers. That is, the participants felt that a substantial number of tribal members were adopting outside cultural values over Native ancestral cultural traits and failing to preserve and maintain the ancestral heritage of the tribe.

Many tribal officers also appeared apprehensive towards the presence of Latinos on the reservation. The tribal community underwent a massive infrastructure improvement project, which ushered in new reservation roads, the construction of new tribal government buildings and expanded the Indian gaming casino. These improvements used outside Latino labor and occurred during the recent economic recession, which negatively impacted the economic status of tribal members. Several of the tribal officers conveyed their negative perceptions about the influx of Latinos working on the reservation. Like many non-American Indians, they blamed migrant workers or immigrants for periods of economic recessions. Americans who suffered from financial burdens often looked for scapegoats, and blame immigrants (Sanchez 1993), for the lack of employment opportunities and the degradation of wages. Anger and distrust of immigrants increased as limited resources was available for White citizens. As Massey (2007:135) suggests "[Americans] have found it convenient to demonize immigrants during periods of social upheaval and economic insecurity." Tribal officers believed Latinos were responsible for economic problems within the tribal community. Indeed, the participants demonized and shared their dismay about the presence of Latinos in the community and a couple of the participants framed these individuals as being criminal or "troublemakers." In fact, some of the tribal officers suggested that Latinos were responsible for drug trafficking, assaults, and property crimes on the reservation. These tribal officers felt that Latinos were exploiting tribal members for their "per cap" check. Some of the participants suggested that Latinos were taking advantage of tribal women to obtain citizenship and voiced concerns about the impact of these inter-racial relationships on tribal families' lineage. Similar to their assessment of the urban culture, the tribal officers viewed Latinos as a threat to the community and their Native culture. The fear of Latinos was so extensive that a couple of the participants inquired about the ethnicity of the researcher, who is Latino. Once again, the tribal officers framed Latinos as the "other" who frequently engage in criminal behaviors or as a threat to the tribal community.

Policing and Stories

In addition to legitimately naming criminals, policing on the reservation also revealed stories about the tribal community. Loader (1997:2) suggests that policing can reveal "stories" about society. As he states:

Every abuse of due process, every failure to respond to call or complaint, every racist ... sexist ...homophobic, every diagnosis of the crime problem, every depiction of criminals – all these send small, routine, authoritative signals about societies conflicts, cleavages and hierarchies about whose claims are considered legitimate within it, about whose status identity is to be affirmed or denied as part of it. (Loader 2006:211)

The conversations with tribal officers revealed "stories" about the tribal community. Indeed, the nature of policing on the reservation provided a glimpse into the daily life in the community. Unfortunately, that glimpse was of a tribal community gripped by intertribal politics. From disappearing traffic violation tickets to the demotion of ranking police officers, the ubiquitous nature of tribal politics greatly impacts the tribal police department and reservation community. As a result, many participants believed that their tribal criminal justice is broken. Indeed, the tribal officers felt that politics greatly diminishes their capacity to perform their policing duties. More importantly, they felt that tribal politics jeopardizes the integrity of the tribal court system as offenders are not held accountable for their crimes. The tribal officers' frustrations about their broken criminal justice system were evident in several discussions. Many tribal officers stated that they carried out their policing duties as best as they can despite the overwhelming influence of tribal politics. The power of tribal politics transcends throughout the community. In a non-recorded interview, a participant revealed that the tribal government refused to financially assist his family to help pay a medical debt. The participant felt that politics was a factor in the tribal government's decision, because he did not vote for one

of the tribal leaders in a previous election. Similarly, other participants revealed that tribal members were given gifts (e.g. home appliances and home repairs) for their political support. Through their "police voice" (Loader and Mulcahy 2001), the participants revealed a tribal community plagued with inter-tribal politics. The tribal officers believed that political influence hinders their ability to effectively police the reservation. Drawing on the proverbial "all bark and no bite," the police found themselves as symbols of law and order; but with little political power at times to enforce the law.

The tribal police officers also experienced challenges to their legitimacy as being enforcers of the law. A couple of participants felt that "outsiders" do not recognize the legitimacy of the tribal police department. In essence, their role as tribal officers is not respected. Simple procedures such as enforcing traffic codes are met with skepticism among non-enrolled tribal members. From questioning their policing powers to allegations of racism, tribal officers confront many barriers in enforcing the law on the reservation. The combination of a racially biased federal court system and a broken tribal justice system may lead some to find that justice is virtually non-existent in this tribal community. Although the research found overarching themes pertaining to colonialism and Loader's (1997) symbolic power of policing, the findings were also consistent with past criminological research on police perceptions.

POLICE PERCEPTIONS

The findings from this research are consistent with previous criminological research on police perceptions. First, the police were reluctant to discuss domestic

violence occurring within the reservation. Given the hegemonic masculine nature of the police, scholars argue that they have a negative attitudes or views about domestic violence (Martin 1989; Fielding 1994; McNamara 1999), which may explain the reluctance among the participants to discuss these crimes. Additionally, some of the participants revealed that there are more male than female victims of domestic violence. This too might explain the lack of discourse among the participants. Past research found that male domestic violence victims are not seen as victims (Gilmartin-Zena 1988; Mitchell et al. 1999). That is, some tribal police officers may perceive that reservation domestic violence crimes with male victims are not actual crimes. Aside from the role of gender influencing the perceptions of tribal police officers, race and ethnicity also proved to a consistent finding in prior research.

Tribal police perceptions of crime with respects to race/ethnicity appeared to be consistent with past criminological research. As previously mentioned, Skolnick (1966) argued that the police rely on their image of the "symbolic assailant" to determine whether an individual has engaged in criminal behavior. The concept of the "symbolic assailant continues to be a topic of discussion within criminology. As Jones-Brown (2007:103) note:

Forty years and many policing innovations later, evidence continues to mount that, in the minds of police agents and civilians, being black and male is inextricably coupled with beliefs about criminality and dangerousness producing a conditioned reflex of suspicion and fear that then leads to over-policing.

With respects to the current study, the tribal police officers provided their description of the "symbolic assailant." For some tribal police officers, the reservation's "symbolic assailant" are those tribal members who embrace elements of the urban culture. Indeed, the participants thought that the urban culture promoted illegal drug use and property

crimes. More importantly, these tribal members also disrespected the Native culture by adopting these outside values. Previous research suggests that violations of Indian cultural values are viewed as a more serious issue than street-level crimes among American Indians (Abril 2007). The participants in this study identified individuals with specific clothes styles and music as people who have the propensity to commit crimes and engage in illegal drug use. In fact, scholars would argue that the use of an individual's appearance constitutes a form of racial profiling (Meehan and Ponder 2002; Weitzer and Tuch 2002; Bjerk 2007; Welch 2007). Aside their perception on nexus between race and crime among the participants, tribal police officers also held views about key actors in the tribal court system.

The tribal police views of the tribal court judges were also consistent with the literature on police perceptions. In the current study, the tribal police officers held negative attitudes about the tribal court judges. This finding is supportive of past research showing that police officers generally have negative attitudes and perceptions towards courtroom judges (Goldstein 1968; LaFave and Remington 1968). For this study, the tribal officers felt that the tribal judges were "too lenient" or "too emotional." The participants also believed that tribal judges rendered lenient sentences which contributed to high rates of recidivism in the tribal criminal justice system. The participants' perceptions about crime and justice are consistent with previous criminological research. It appears from the data and theoretical frame that there is a dialectic process occurring between police symbolic use of power, the carceral state, and the historical legacies of colonialism. Based on this, the following section addresses potential policy implications.

POLICY IMPLICATIONS

The perceptions among the tribal police officers reveal social problems that extend beyond the reservation community. As Jackson and Bradford (2009:514) note:

Some crimes are signals not only of criminality, but also provide messages about a broader set of social problems, it is possible that these are driving perceptions of disorder and lack of informal social control and in turn affecting confidence in the police.

Indeed, the participants' perceptions exposed a wide array of social problems that plague their reservation community. These problems could be alleviated in a collaborative partnership with both the federal and tribal government. First, given the magnitude of substance abuse within the community, the tribal government should establish substance abuse programs to educate and rehabilitate tribal members suffering from illegal drug and alcohol addictions. As revealed by the participants, substance abuse problems were strongly correlated to crime and violence within their reservation community. To overcome substance abuse problems, the tribal government and social service providers should draw on evidence-based treatment programs which are culturally sensitive to American Indian cultural values. The implementation of these treatment programs could substantially reduce illegal drug related crimes.

Family and parenting programs emerge as another possibility to address tribal crime and violence. Indeed, the participants recounted the impact of poor parenting on the tribal community. Due to the effects of colonialism, the structure and dynamics of Native families shifted dramatically. As a result, Native families experience an unprecedented rate of familial disruption, which is correlated to crime and violence. To overcome familial problems, the tribal government should establish family orientated and parenting programs directed towards tribal members. These programs ought to teach tribal members positive parenting and conflict resolution techniques, which may reduce the prevalence of anti-social behaviors among tribal youth. Once again, these programs should also be sensitive to the cultural needs of tribal members. With respects to childcare, tribal child-welfare services should proactively respond to the needs of troubled tribal children. As revealed in the discussions, tribal victims of child abuse and neglect seldom receive inadequate treatment from social services for their victimization.

Next, the tribal government should expand on existing preventative domestic violence to include male victims of domestic violence. A couple of participants revealed that the reservation community has more male victims of domestic violence than outside neighboring communities. To overcome gender stigmas associated with male victims, preventive domestic programs should raise awareness that domestic violence is not restricted to any gender and males can also find themselves victims of this crime. In addition to these programs, the tribe should establish a shelter for domestic violence victims.

Aside from establishing programs to alleviate some of the social problems within the reservation community. The tribal community should enact policies and organizations to address the ubiquitous nature of inter-tribal politics on the reservation. The tribal police officers revealed that tribal leaders greatly influence the criminal justice system to appease their constituents which jeopardizes the integrity of the tribal court system. There are several ideas to diminish the influence of tribal politics within the tribal court system. First, tribal judges should be elected by the tribal community, which could reduce the political influence of tribal leaders who appoint these individuals to the tribal bench. Also, policies should be enacted that hold individuals accountable for abusing their position or tribal government office which may also diminish political influence within the tribal criminal justice system.

The federal government has considerable power in addressing crime and justice problem in Indian Country. For instance, the government can substantially increase funding for reservation communities. Nielsen (1996:299) argues that many reservation communities suffer from "developmental needs."

Developmental needs" refers to the lack of resources that exist in Native communities and hinder the development of criminal justice services. Developmental needs are closely related to the lack of sovereignty. Because Native Peoples have been prevented from governing their own affairs and operating their own services until very recently, they have often not done well when experimental programs were thrust on them "from above," that is, by the federal government.

While the federal government promotes the idea of tribal sovereignty, they continue to decrease monetary appropriations for American Indian initiatives (Gould and Ross 2006). Indeed, for tribal communities to be self-efficient they must be economically stable. Given the geographical isolation of a substantial number of reservation communities, many Americans Indian in Indian Country are unable to secure employment as one might find in larger urban areas, which contributes to high rates of poverty and unemployment within these areas. With assistance from the federal government, policies could be created to promote economic success within these communities. As Bachman (1992:124) suggests, "creating policies that permit all American Indian communities to have the opportunity to become economically viable seems to be an urgent priority." Aside for a need of federally sponsored economic initiatives, the government should provide resources to address crime and justice in Indian Country.

In light of the war on terrorism, crime and violence in Indian Country remains a non-essential priority of the federal government. In his reporting about the lack of federal assistance, Riley (2007b:1) found that these problems remain at the bottom of the government's list:

The FBI would always say that Indian Country was the agency's second-highest priority, and our running joke was that there is everything else and then there is Indian Country, said James Kilbourne, tribal prosecutor for the Eastern Band of Cherokee in the mountains of North Carolina. Then, after 9/11 and terrorism, we became the third- highest priority - terrorism, everything else and then Indian Country.

Indeed, following the 9/11 attacks, the federal government dramatically directed its attention to funding national security initiatives. As Gould and Ross (2006:240) point out:

In post-9/11 America, resources are tight, and the focus is on new external problems rather than on long-term, internal difficulties. There will be a tendency to spend money on such things as the border patrol.

With the federal government's focus on enhancing national security measures, tribal governments receive fewer resources than are needed to address crime. For example, the U.S. Department of Justice reduced the number of U.S. Attorneys responsible for prosecuting serious crimes occurring on tribal lands (Riley 2007b). The lack of resources for tribal communities greatly hinders their ability to address crime and violence as well as administer justice in Indian Country.

More importantly, the government should increase the tribal sovereignty and selfdetermination for all reservation communities. By granting American Indians and tribal governments more powers they are in position to control their fate within society. For instance, the federal government should lessen the restrictions placed on tribal courts so that these institutions can adequately respond to crime and violence within reservation communities. As previously mentioned, tribal courts are restricted to hearing misdemeanor cases and can impose no more than a three-year imprisonment sentence and up to \$15,000 in fines. The participants revealed that perpetrators realize the court's limitations and often exploit the weaknesses of the court and habitually commit crimes on the reservation because they do not fear serious repercussions. To remedy this situation, the federal government should grant tribal courts similar powers to those found in traditional court systems. In addition, the federal government should grant tribal courts full jurisdiction to handle all crimes and parties (irrespective of enrollment status of either party), such powers will inevitably enhance tribal sovereignty and heighten selfdetermination of the American Indian population.

Another idea to improve the lives of American Indians requires a social and cultural changes with respects to public attitudes towards Native populations. For centuries, indigenous populations have suffered from negative labels and stigmas to marginalize them. Unfortunately, these stereotypes continue to haunt American Indians to the extent that many individuals believe these negative attributes to be true. In doing so:

We humiliate someone – either individually or as a member of a larger collective – we make subsequent cruelty to that person easier, for ourselves and for others. If a person or a group of persons is routinely referred to, thought of, and treated as demons, objects, machines, animals, or otherwise subhuman, physical cruelty is a short leap away. (Levy 2000:25)

With respects to the present study, it could be argued that negative public perceptions about American Indians are responsible for the high levels of illegal drug use and violence among these individuals. For these problems to be addressed, requires a transformation about attitudes, opinions, and perceptions of American Indians in a positive light. However, this change should also take place among tribal police officers.

The tribal police officers' perceptions reinforce negative stereotypes and beliefs about criminality. The participants recognized individual-level factors, rather than structural factors that contribute to crime and delinquency on the reservation. In doing so, they reinforced popular stereotypes and fallacies about crime and violence. As a result, individuals suffer from insecurity and clamor for police protection. As Loader (2006:212) notes:

Policing can enact or support forms of cultural domination, nonrecognition, and disrespect in ways that conduce to both objective and subjective insecurity. They demonstrate that certain 'tough' styles of ambient policing can appear to reduce crime or produce order while contributing to public insecurity through unfair practices.

Indeed, the tribal police officers otherized and demonized drug addicts and fellow tribal members. This in itself contributes to cultural domination through the criminalization of certain groups of individuals. As a result, it facilitates insecurity among tribal members in the community and breeds fear among individuals, which can impact the quality of life on the reservation. For many tribal police officers, they spoke about controlling and preventing crime through more policing personnel and equipment rather than social programs designed to address the root causes of reservation crimes. As a result, they reified the need for policing within the reservation community. The tribal police officers should change their negative perceptions and beliefs about crime, instead of producing, reproducing, and circulating inaccuracies about criminality.

CONTRIBUTIONS AND FUTURE DIRECTIONS

This study contributed to the criminological literature by providing an overview about crime and justice as perceived by a group of tribal police officers. Indeed, the study offered a glimpse into the dynamics of crime and justice within a tribal community. The participants perceptions of a tribal community suffering from extraordinary rates of illegal prescription drug use that is strongly correlated with crime and violence as well as a tribal justice system that is plagued with federal restrictions and inter-tribal politics has indeed become a reality within the community. Loader's (1997) symbolic power of policing provides an adequate explanation about the role of tribal police in the discourse about crime and justice in Indian Country. That is, tribal police officers were able to discuss all crime-related matters given their "expertise" about these problems.

Upon data collection, the researcher noted the level of sophistication with respects to the policing equipment that was found within the tribal police department. Indeed, the police department possessed a wide array of policing equipment that may be found in typical mid-size U.S. cities. For example, the tribal police department had a SWAT vehicle, Segway, taser weaponry, and other policing equipment. Taken together, this suggests that there might be a militarization effect occurring in Indian Country which may them be manifested into a reality, based on Loader's (1997) ideas on the symbolic power of policing. Given the high rates of crime and violence, tribal police departments could be utilizing federal funds to purchase policing equipment to win the war on crime. Future research should explore the militarization dynamics that could be occurring within reservation communities. Other research should also compare crime rates between reservation and non-reservation communities. The findings from this study may have implications for future research to further understand the dialectics of crime and justice in Indian Country and police perceptions. For instance, a majority of the participants believed that the reservation's crime rate was similar to neighboring communities. Future research should examine compare and contrast the crime rates between tribal and non-tribal communities to determine whether these rates are comparable. Although past studies have examined homicide rates between reservation and non-reservation communities (see Lanier & Huff-Corzine 2006; Bachman 1991), future research should examine other crimes (e.g. domestic violence, property crimes, and illegal drug consumption) between the two distinct communities. Another area to examine is the use of informal resolutions among tribal members living in reservation communities.

Given the inadequacies of the tribal and federal court systems in responding to American Indian victims and offenders, the participants revealed that tribal members rely on informal methods to resolve disputes between themselves. Many tribal members do not have faith in either the tribal or federal court system and resort to informal conflict resolutions to avenge their victimization. Future research should draw on Girling, Loader, and Sparks' (1998) work on "local talk" and individuals' perceptions on the fear of crime as it is related to the use of vigilante justice. American Indians living in Indian Country reside in close-knit communities that allows them to be fully aware of victimizations occurring in these areas. As a result, scholars should investigate the use of "local talk" within tribal communities in relation to the use of informal conflict resolutions.

While American Indians have made great strides to maintain their identity and culture, they, like everyone in the U.S., experience significant exposure to mass media and the promulgation of mainstream American values of consumerism and materialism. Their predicament illustrates Jock Young's (2007,2003) "bulimic society" perspective, which explains the transmission process of dominant cultural norms and values to lower class individuals at the same time that structural conditions prevent them from achieving those same cultural goals. American Indians, of whatever social class, exemplify this experience. Many American Indians who live in reservation communities are unable to participate in the "American Dream" cultural ethos of consumerism and materialism. As a result, they are forced to rely on illegitimate means not only to obtain these cultural goals, but also to meet basic survival needs, which ultimately lead to their involvement in the criminal justice system. More importantly, reservation communities can be characterized as an "epicenter" for culture conflict. While American Indians are attempting to maintain their cultural identity and ancestral traditions, they are simultaneously enticed to adopt mainstream American values. This theoretical framework can serve as the foundation for future research investigating the perceptions about crime, violence, and justice, and their connection to the tribal culture among American Indian adults and youth. Future research on American Indian criminality can also overcome some of the limitations of the current study.

Future research should expand on the number of participants to fully understand the dynamics of crime, violence, and justice in Indian country. For example, the current study relies on the perceptions of 27 police officers from one tribal police department, which means that the results are not representative to the general American Indian population. As such, the perceptions of crime, violence, and justice among the participants could be considerably different from individuals in other tribal communities. In addition, the participants could have withheld or exaggerated their thoughts about crime, violence, and justice within their community, which could subject the findings to biases. Simply put, it is unknown whether the accounts provided by the participants are accurate although the fact that similar stories were frequently repeated suggests that they are accurate, at least in terms of "local talk." Additionally, data depended on the participant to volunteer information about crime, violence, and justice within their community. It could be possible that tribal officers may exclude experiences that may be embarrassing or that they have difficulty remembering events. Foregoing these limitations, the study does provide valuable insight about the dynamics of crime, violence, and justice in a reservation community.

CONCLUSION

This research offers a window into the problems experienced among American Indians living in Indian Country. Through their "police voice," the participants suggested that problems experienced in their reservation were similar to other rural communities outside of Indian Country. Although the tribal officers revealed that the tribal community suffers from illegal prescription drug use, property crimes, and domestic violence, they cautioned that these crimes were also occurring in non-tribal communities. Overwhelmingly, the tribal officers rebuked the "war zone" analogy in describing the rate of crime and violence. In fact, to equate an Indian reservation as a "war zone" further propagates negative American Indian stereotypes that longed have been used to "otherize" this group of individuals. Although the tribal officers believed that their community is no different than other communities, they revealed sources of colonial domination that continue to subjugate American Indians to the lower fringes of society.

The tribal officers perceived that American Indians continue to suffer from the social ills of colonialism which has been substantiated by the data. Although American Indians have made great strides to achieve tribal sovereignty, the federal government has limited their tribal autonomy. The tribal officers felt that federal restrictions on the tribal court system obstruct their ability to achieve justice in their community. In addition, they also believed that inter-tribal politics negatively affects the triba's ability to hold tribal members accountable for their transgressions. Additionally, the tribal officers also provided narratives of miscarriages of justice at the hands of the government through under-and-over prosecuting of American Indians in the federal court system. Achieving justice is further complicated by the disjunction between federal and tribal law. Indeed, the results indicated that justice remains elusive for American Indians living in Indian Country.

After 400 years of historical trauma and cultural oppressions, American Indians continue to cling to their traditions and ancestral culture. That is to say, in the face of adversity, American Indians continue to be resilient. In decades to come, we can only hope that American Indians continue to be tenacious and persistent in achieving greater tribal autonomy and social equality, which will only enhance their standing in society as well as improve the lives of people living in Indian Country.

/

APPENDIX

INTERVIEW SCHEDULE

Introduction: Hi, my name is Favian Martin and I am a Ph.D. student at Old Dominion University in Norfolk, VA. Thank you so much for coming today to help me with this research. Today we will be talking about your thoughts about crime, violence, and justice in your tribal community. Your participation is strictly confidential. That means that I will never tell anyone what any individual said here today. I may use what you say, but I will not identify that you were even here or that you said any specific thing.

I am recording our conversation so that I can focus on what you are saying instead of trying to take notes. I will listen to the recording to take notes later. I will also destroy the recording as soon as the notes are complete – your real names won't appear in my notes or any of my files. Again, I am recording so I can be really focused on you.

Our conversation is just one part of the research I am doing for my dissertation. When I am done, the tribe will get a copy. But again, no one here will be mentioned by name in the dissertation or in any way that you can be identified.

Does anyone have any questions? May we begin?

- What does it mean to be an American Indian?
 - What do you mean?
- Is crime or violence as a problem here on the reservation?
 - What makes you think that?
 - Could you give me some examples?
- Do you think that crime and violence in your community is related to the tribal culture in anyway?
 - Why do you think this is the case?
- What do you think about the tribal police and the tribal court?
 - Why do you say that? (Straightforward reply)
 - What do you mean by that? (If answer seems unclear)
- What type of media do most of the people here pay attention to the most?
 - Do many people here have access to the internet? T.V.? (If these are not brought up)
- What messages do you think that people take from the media
- Do you think that the media has any influence on crime and violence here in the reservation?
 - Why do you say that?
 - Do you think that the youth's behavior is influenced by the media? (If they bring up the youth)
- Does the media have any influence on the tribal police or tribal court system?
 - Could you tell me more about your thinking on that?

- Are there any stereotypes about American Indians, generally or about your tribe specifically?
 - How do they impact you?
- How do outside influences impact your connection to the American Indian culture?
- Do these stereotypes have any connection to crime and violence here?
 - Do you think that these stereotypes influence the tribal police or tribal court?

Conclusion of the interview:

Do you have questions for me?

Closing: I want to thank you for your time. I hope that you've found this time useful. You have really helped me and I appreciate it. As I mentioned at the beginning, this is for my dissertation research which the tribe will get a copy of. Remember that your names will not be included and I will never tell anyone what any of you as individuals said. Thanks again and have a great day.

REFERENCES

- Aarons, Natalie M., Martine B. Powell, and Jan Browne. 2004. "Police Perceptions of Interviews Involving Children with Intellectual Disabilities: A Qualitative Inquiry." *Policing and Society* 14(3):269-278.
- Abril, Julie C. 2007. "Cultural Conflict and Crime: Violations of Native American Indian Cultural Values." International Journal of Criminal Justice Sciences 2(1):44-62.
- Ackerson, Lynn M., Rhonda Wiegman, Spero M. Manson, and Anna E. Baron. 1990. "Properties of the Inventory to Diagnose Depression in American Indian Adolescents." Journal of the American Academy of Child & Adolescent Psychiatry 29(4):601-607.
- Adams, David Wallace. 1995. Education for Extinction: American Indians and the Boarding School, 1875-1928. Lawrence, KS: University Press of Kansas.
- Albrecht, Stan L., & Miles Green. 1977. "Attitudes Toward the Police and the Larger Attitude Complex." Criminology 15(1):67-86.
- Alvarez, Alexander and Ronet D. Bachman. 1996. "American Indians and Sentencing Disparity: An Arizona Test." *Journal of Criminal Justice* 24(6):549-61.
- American Bar Association. 2001. *The Directory of Minority Judges of the United States*. American Bar Association, Judicial Division, Standing Committee on Minorities in the Judiciary. New York, NY: CQ Press.
- American Indian and Alaska Native Resource Manual (NAMI). 2003. "American Indian and Alaska Natives Communities Mental Health Facts." Arlington, VA:
 American Indian and Alaska Native Resource Manual. Retrieved on January 15, 2013.
 (http://www.nami.org/ContentManagement/ContentDisplay.cfm?ContentFileID= 32231).
- Amnesty International. 2004. Threat and Humiliation: Racial Profiling, Domestic Security, and Human Rights in the United States. New York, NY: Amnesty International.
- Andreas, Peter and Richard Price. 2001. From War Fighting to Crime Fighting: Transforming the American National Security State. *International Studies Review*. 3(3): 31-52.
- Angell, J. E. 1978. Alaska Village Police Training: An Assessment and Recommendations. Anchorage, AK: University of Alaska Anchorage, Criminal Justice Center.

- Armstrong, Troy L., Michael H. Guifoyle, and Ada Pecos Melton. 1992. "Native American Delinquency: An Overview of Prevalence, Causes, and Correlates, and Promising Tradition-Based Approached to Sentencing." Office of Juvenile Justice and Delinquency Prevention. Washington DC: National Institute of Justice.
- Arrick, L. Jackson and John E. Wade. 2005. "Police Perceptions of Social Capital and Sense of Responsibility: An Explanation of Proactive Policing." *Policing* 28(1): 49-68.
- Arya, Neelum and Addie Rolnick. 2008. "A Tangled Web of Justice: American Indian and Alaska Native Youth in Federal, State, and Tribal Justice Systems." Campaign for Youth Justice Policy Brief. Washington, DC: Campaign for youth Justice. Retrieved October 12, 2013 (http://www.campaignforyouthjustice.org/documents/CFYJPB_TangledJustice.pd f).
- Bachman, Ronet. 1992. Death and Violence on the Reservation: Homicide, Family Violence, and Suicide in American Indian Population. New York, NY: Auburn House.
 - ——. 1991a. "An Analysis of American Indian Homicide: A Test of Social Disorganization and Economic Deprivation at the Reservation County Level." Journal of Research in Crime and Delinquency 28(4):456-71.
- -----. 1991b. "The Social Causes of American Indian Homicide as Revealed by the Life Experiences of Thirty Offenders." *American Indian Quarterly* 15(4):469-92.
- Bachman, Ronet, Heather Zaykowski, Christina Lanier, Margarita Poteyeva, and Rachel Kallmyer. 2010. "Estimating the Magnitude of Rape and Sexual Assault against American Indian and Alaska Native Women." *Australian & New Zealand Journal* of Criminology 43(2):199-222.
- Baker, David V. 2007. "American Indian Executions in Historical Context." Criminal Justice Studies 20(4):315-373
- Banner, Stuart. 2005. *How the Indians Lost Their Land*. Cambridge, MA: Harvard University Press.
- Banton, Michael. 1964. Policeman in the Community. New York: Basic Books.
- Barak, Greg, Paul Leighton, and Jeanne Flavin. 2007. Class, Race, Gender, and Crime. Lanham, MD: Rowman and Littlefield.
- Bard, Morton and Joseph Zacker. 1974. "Assaultiveness and Alcohol Use in Family Disputes: Police Perceptions." *Criminology* 12(3):281-292.

- Bargh, J. A. 1996. Automaticity in Social Psychology. Pp. 169-183 in Social psychology: Handbook of Basic Principles edited by E. T. Higgins & A. W. Kruglanski. New York City, NY: Guilford.
- Barlow, D. E., and M. H Barlow. 2000. Police in a Multicultural Society: An American Story. Prospect Heights, IL: Waveland Press.
- Barlow, David E., and Melissa Hickman Barlow. 2002. "Racial Profiling: A Survey of African American Police Officers." *Police Quarterly* 5(3):334-358.
- Bayley, David H. 1994. Police for the Future. New York, NY: Oxford University Press.
- Bazemore, Gordon, and Mark Umbreit. 1995. "Rethinking the Sanctioning Function in Juvenile Court: Retributive or Restorative Responses to Youth Crime." Crime & Delinquency 41(3):296-316.
- Beals, Janette, Joan Piasecki, Scott Nelson, Monica Jones, Ellen Keane, Paul Dauphinais, Roy Red Shirt, William H. Sack, and Spero M. Manson. 1997. "Psychiatric Disorder among American Indian Adolescents: Prevalence in Northern Plains Youth." Journal of the American Academy of Child & Adolescent Psychiatry 36(9):1252-59.
- Beauvais, Fred and Steve LaBoueff. 1985."Drug and Alcohol Abuse Intervention in American Indian Communities." *International Journal of the Addictions* 20(1): 139-71.
- Beauvais, Fred., Eugene R. Oetting, W. Wolf, and Ruth W. Edwards. 1989. "American Indian Youth and Drugs, 1976-87: A Continuing Problem." American Journal of Public Health 79(5):634-36.
- Beauvais, Fred., Eugene R. Oetting, and Ruth W. Edwards. 1985. "Trends in the Use of Inhalants among American Indian Adolescents." *White Cloud Journal of American Indian Mental Health* 3(4):3-11.
- Becker, T. M., J.M. Samet, C.L. Wiggins, and C.R. Key. 1990. "Violent Death in the West: Suicide and Homicide in New Mexico, 1958-1987." Suicide and Life-Threatening Behavior 20(1):324-34.
- Belknap, Joanne. 2007. The Invisible Woman: Gender, Crime, and Justice. Belmont, CA: Thomson Wadsworth.
- -----. (1995). "Law Enforcement Officers' Attitudes About the Appropriate Responses to Woman Battering." *International Review of Victimology*, 4(1):47-62.
- Belknap, Joanne and Hillary Potter. 2005. "The Trials of Measuring The 'Success' Of Domestic Violence Policies." *Criminology and Public Policy* 4(3):559-66.

- Belur, Jyoti. 2009. "Police Use of Deadly Force: Police Perceptions of a Culture of Approval." Journal of Contemporary Criminal Justice 25(2):237-252.
- Bennett, Andy. 1999. "Hip hop Am Main: The Localization of Rap Music and Hip Hop Culture." *Media, Culture & Society* 21(1):77-91.
- Bennett, William J., Dilulio, Jr., John J., & Walters, John P. 1996. Body Count: Moral Poverty -- and How to Win America's War Against Crime and Drugs. New York, NY: Simon and Schuster.
- Berg, Bruce L. 2007. *Qualitative Research Methods for the Social Sciences*. Boston, MA: Pearson.
- Berger, Peter L. and Thomas Luckmann. 1967. *The Social Construction of Reality: A Treatise in the Sociology of Knowledge*. New York, NY: Anchor Books.
- Berk, S.F. and D.R. Loseke. 1980-1981. "Handling Family Violence: Situational Determinants of Police Arrests in Domestic Disturbances." *Law and Society Review* 15(1):317-46.
- Bernard, Thomas J., Jeffrey B. Snipes, and Alexander L. Gerould. 2010. Vold's Theoretical Criminology. New York, NY: Oxford University Press.
- Binder, Amy. 1993. "Constructing Racial Rhetoric: Media Depictions of Harm in Heavy Metal and Rap Music." *American Sociological Review* 58(6):753-767.
- Bissett, Daryl, Jennifer Bissett and Snell, Clete. 2012. "Physical agility tests and fitness standards: perceptions of law enforcement officers." *Police Practice & Research* 13(3):208-223.
- Bittner, Egon. 1970. *The Functions of Police in Modern Society*. Chevy Chase, MD: National Clearinghouse for Mental Health.
- Bjerk, David. 2007. "Racial Profiling, Statistical Discrimination, and the Effect of a Colorblind Policy on the Crime Rate." *Journal of Public Economic Theory* 9(3): 521-545.
- Blauner, Robert. 1994. "Talking Past Each Other: Black and White Languages of Race." Pp. 27-34 in *Race and Ethnic Conflict: Contending Views on Prejudice, Discrimination, and Ethnoviolence,*, edited by F. Pincus and H. Ehrlich. Boulder, CO: Westview Press.
- ——. 1972. Racial Oppression in America. New York, NY: Harper and Row.
- Blumer, Herbert. 1969. Symbolic Interactionism: Perspectives and Methods. New York, NY: Prentice Hall.

- Bohm, Robert M. 2001. *A Primer on Crime and Delinquency Theory*. Belmont, CA: Wadsworth.
- Bohn, Diane K. 2003. "Lifetime Physical and Sexual Abuse, Substance Abuse, Depression, and Suicide Attempts among Native American Women." *Issues in Mental Health Nursing* 24(3):333-352.
- Bond-Maupin, Lisa, Taka X. GoodTracks, and James R. Maupin. 2006. "Research on Juvenile Delinquency in Indian Communities: Resisting Generalization." Pp. 187-196 in Native Americans in the Criminal Justice System edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- Boni, N., and J. Packer. 1998. The Police Role Survey: A Tool for Comparing Public and Police Perceptions of the Police Role: Report Series No 126. Canberra, AUS: Australasian Centre for Policing Research.
- Bonta, James. 1989. "Native Inmates: Institutional Response, Risk, and Needs." Canadian Journal of Criminology 18(1):49-62.
- Bordua, David J. and Larry L. Tifft. 1971. "Citizen Interviews, Organizational Feedback, and Police-Community Relations Decisions." *Law & Society Review* 6(2):155-182.
- Borum, Randy, Deane, Martha Williams, Steadman, Henry J., & Morrissey, Joseph. 1998. "Police Perspectives on Responding to Mentally Ill People in Crisis: Perceptions of Program Effectiveness." *Behavioral Sciences & the Law* 16(4):393-405.
- Bourdieu, Pierre. 1991. Language and Symbolic Power. Cambridge: Polity Press.
- Boutwell, Brian B. and Kevin M. Beaver. 2010. "The Role of Broken Homes in the Development of Self-Control: A Propensity Score Matching Approach." *Journal* of Criminal Justice 38(4):489-495.
- Bowker, L. 1982. "Police Services to Battered Women: Bad or Not So Bad?" Criminal Justice and Behavior 9(4):476-94.
- Bracey, Dorothy H. 2006. "Criminalizing Culture: An Anthropologist Looks at Native Americans and the U.S. Legal System." Pp. 35-51 in *Native Americans and the Criminal Justice System*, edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- Brave Heart, Maria Yellow Horse. 2003. "The Historical Trauma Response among Natives and Its Relationship with Substance Abuse: A Lakota Illustration." *Journal of Psychoactive Drugs* 35(1):7-13.

- ———. 1999. "Oyate Ptayela: Rebuilding the Lakota Nation through Addressing Historical Trauma among Lakota Parents." Journal of Human Behavior in the Social Environment 2(2):109-26.
- Brave Heart-Jordan, Maria Yellow Horse. 1995. "The Return to the Sacred Path: Healing from Historical Trauma in the Treatment of Native American Indian Women."
 Pp. 345-368 in *Racism in the Lives of Women: Testimony, Theory and Guides to Anti-Racist Practice*, edited by J. Adelman and G. Enguidanos. New York: Haworth Press.
- Britz, MarjieT. 1997. "The Police Subculture and Occupational Socialization: Exploring Individual and Demographic Characteristics." *American Journal of Criminal Justice* 21(2):127-146.
- Brown, Lester B. 1997. "Women and Men, Not-Men and Not-Women, Lesbians and Gays." Journal of Gay & Lesbian Social Services 6(2):5-20.
- Brown, M. K. 1988. Working the Street: Police Discretion and the Dilemmas of Reform. Beverly Hills, CA Sage.
- Brown, Stephen E. 1984. "Police Responses to Wife Beating: Neglect of a Crime of Violence." Journal of Criminal Justice 12(3):277-88.
- Browning, Christopher R., Reginald A. Byron, Catherine A. Calder, Lauren J. Krivo, Mei-Po Kwan, Jae-Yong Lee, and Ruth D. Peterson. 2010. "Commercial Density, Residential Concentration, and Crime: Land Use Patterns and Violence in Neighborhood Context." *Journal of Research in Crime and Delinquency* 47(3):329-357.
- Burnett, S. 2006. "Ute Homicide Rate Soars." *The Rocky-Mountain News* May 15. Retrieved February 15, 2013. www.rockymountainnews.com/com/news/2006/Nov/25/ute-homicide.
- Burrus, J. H., Jr. 2006. "Remarks by James H. Burrus, Jr. Assistant Directory, Criminal Investigative Division," National Congress of American Indians Annual Convention, Sacramento, California. Retrieved February 4, 2013 (http://www.fbi.gov/pressrel/speeches/burrus100506.htm).
- Bush-Baskette, Stephanie. 2000. "The War on Drugs and the Incarceration of Mothers." *Journal of Drug Issues* 30(1):919-28.
- Buzawa, E.S., & C.G. Buzawa. 2003. Domestic Violence: The Criminal Justice Response. Newbury Park, CA: Sage

Cain, Maureen. 1973. Society and the Policeman's Role. London: Routledge.

- Calloway, Colin G. 2006. The Scratch of a Pen: 1763 and the Transformation of North America. New York, NY: Oxford University Press.
- Campbell, Rebecca, & Johnson, Camille R. 1997. "Police Officers' Perceptions of Rape: Is There Consistency Between State Law and Individual Beliefs?" *Journal of Interpersonal Violence* 12(2):255-274.
- Carlson, Leonard A. 1978. "The Dawes Act and the Decline of Indian Farming." The Journal of Economic History 38(1):274-76.
- Carter, Samuel. 1976. Cherokee Sunset : A Nation Betrayed : A Narrative of Travail and Triumph, Persecution and Exile. Garden City, N.Y: Doubleday.
- Chambliss, William J. 1964. A Sociological Analysis of the Law of Vagrancy. Social Problems 12(1):67-77.
- Chang, Semoon. 1996. "Impact of Casinos on Crime: The Case of Biloxi, Mississippi." Journal of Criminal Justice 24(5):431-436.
- Centers for Disease Control and Prevention. 2011. "National Center for Injury Prevention and Control. Web-based Injury Statistics Query and Reporting System (WISQARS)." Atlanta, GA: Centers for Disease Control and Prevention.
- Centers for Disease Control and Prevention. 2008. "National Center for Injury Prevention and Control. Suicide Prevention: Youth Suicide." Atlanta, GA: Centers for Disease Control and Prevention.
- Centers for Disease Control and Prevention. 2007. "National Center for Health Statistics and Office of Injury Mortality, Division of Violence Prevention. Atlas of Injury Mortality." Atlanta, GA: Centers for Disease Control and Prevention.
- Centers for Disease Control and Prevention. 2005. "National Center for Injury Prevention and Control. Web-based Injury Statistics Query and Reporting System." Atlanta, GA: Centers for Disease Control and Prevention.
- Centers for Disease Control and Prevention. 2004. "National Center for Injury Prevention and Control. Web-based Injury Statistics Query and Reporting System (WISQARS)." Atlanta, GA: Centers for Disease Control and Prevention.
- Centers for Disease Control and Prevention. 2003. "Injury Mortality among American Indians & Alaska Native Children and Youth – United States, 1989-1998." Atlanta, GA: Centers for Disease Control and Prevention.
- Chaiken, Jan M., Thomas B. Crabill, L. P. Holliday, David L. Jaquette, M. Lawless, and E. S. Quade. 1977. "Summary of Operational Criminal Justice Models." Pp. 29-

40 in *Modeling the Criminal Justice System*, edited by Stuart S. Nagel. Beverly Hills, CA: Sage.

Chamlin, Mitchell B. 1991. "A Longitudinal Analysis of the Arrest-Crime Relationship: A Further Examination of the Tipping Effect." Justice Quarterly 8(2):187-99.

------. 1988. "Crime and Arrests: An Autoregressive Integrated Moving Average Approach." Journal of Quantitative Criminology 4(1):247-58.

- Chesney-Lind, Meda. 2002. "Criminalizing Victimization: The Unintended Consequences of Pro-Arrest Policies for Girls and Women." *Criminology and Public Policy* 2(1):81-90.
- Chester, B., R., W. Robin, M. P. Koss, J. Lopez, and D. Goldman. 1994. "Grandmother Dishonored: Violence against Women by Male Partners in American Indian Communities." Violence & Victims 9(3):249-58.
- Chief Elk, Lauren. "Hidden Epidemic: Why One Group of Americans Is Raped without Justice." *Women Under Siege Project* December 9. Retrieved on January 25, 2013 (http://www.womenundersiegeproject.org/blog/entry/Hidden-epidemic-why-onegroup-of-Americans-is-raped-without-justice).
- Child, Brenda. 1993. "Homesickness, Illness, and Death: Native-American Girls in Government Boarding Schools." Pp. 169-179 in Wings of Gauze: Women of Color and the Experience of Health and Illness, edited by B. Bair and S. E. Cayleff. Detroit, MI: Wayne State University Press.
- Churchill, Ward. 2004. Kill the Indian, Save the Man: The Genocidal Impact of American Indian Residential Schools. San Francisco, CA: City Lights Books.

———. 1997. A Little Matter of Genocide: Holocaust and Denial in the Americas 1492 to the Present. San Francisco, CA: City Lights Books.

- Clawson, C. and S.K. Chang. 1977. "Relationship of Response Delay and Arrest Rates." Journal of Police Science and Administration 5(1):53-68.
- Cohen, David S. 1996. "Official Oppression: A Historical Analysis of Low-Level Police Abuse and Modern Attempt at Reform." *Columbia Human Rights Law Review* 1(28):32-42.
- Coker, Donna. 1999. "Enhancing Autonomy for Battered Women: Lessons from Navajo Peacemaking." UCLA Law Review 47(1):1-111.
- Colman, A., & Gorman, L. 1982. "Conservatism, Dogmatism and Authoritarianism in British Police Officers." Sociology 16(1):1-11.

- Conley, Robert J. 2005. *The Cherokee Nation: A History*. Albuquerque, NM: University of New Mexico Press.
- Connell, R. W. 2002. Gender (Short Introductions). Cambridge, UK: Polity Press.
- -----. 1987. Gender and Power. Cambridge, UK: Polity.
- ------. 1995. Masculinities. Los Angeles, CA: University of California Press.
- Cooper, Virginia G., Alix M., Mclearen, and Patricia A. Zapf. 2004. "Dispositional Decisions with the Mentally Ill: Police Perceptions and Characteristics." *Police Quarterly* 7(3):295-310.
- Cornell, Stephen, & Kalt, Joseph P. 1998. "Sovereignty and Nation-Building: The Development Challenge in Indian Country Today." *American Indian Culture and Research Journal* 22(3):187-214.
- Correia, Mark E., Michael D. Reisig and Nicholas P. Lovrich. 1996. "Public Perceptions of State Police: An Analysis of Individual-Level and Contextual Variables." *Journal of Criminal Justice* 24(1):17-28.
- Cosmides, Leda, John Tooby, and Robert Kurzban. 2003. "Perceptions of Race." *TRENDS in Cognitive Sciences*, 7(4):173-179.
- Costello, E. J., E. M. Farmer, A. Angold, B. J. Burns, and A. Erkanli. 1997. "Psychiatric Disorders among American Indian and White Youth in Appalachia: The Great Smoky Mountains Study." *American Journal of Public Health* 87(5):827-32.
- Couture, Joe, Ted Parker, Ruth Couture, and Patti Laboucane. 2001."Community Holistic Circle Healing: A Cost-Benefit Analysis, Hollow Water." Manitoba: Public Safety and Emergency Preparedness Canada.
- Craglia, M., R. Haining and P. Signoretta. 2005. "Modeling High-Intensity Crime Areas: Comparing Police Perceptions With Offence/Offender Data in Sheffield." *Environment and Planning* 1(37):503-524.
- Cramer, Renee Ann. 2006. "The Common Sense of Anti-Indian Racism: Reactions to Mashantucket Pequot Success in Gaming and Acknowledgment." *Law & Social Inquiry* 31(2):313-341.
- Creswell, John W. 2007. *Qualitative Inquiry and Research Design*. Thousand Oaks, CA: Sage.

——. 1998. *Qualitative Inquiry and Research Design: Choosing among Five Traditions*. Thousand Oaks, CA: Sage.

- Culverson, Donald R. 1998. "Stereotyping by Politicians: The Welfare Queen and Willie Horton." Pp. 51-76 in *Images of Color: Images of Crime*, edited by C. R. Mann and M. S. Zatz. Los Angeles, CA: Roxbury Publishing.
- Daly, Kathleen. 2002. "Restorative Justice." Punishment & Society 4(1):55-79.
- Daly, Kathleen and Julie Stubbs. 2006. "Feminist Engagement with Restorative Justice." *Theoretical Criminology* 10(1):9-28.
- Dantzker, M. L., and Betsy Kubin. 1998. "Job Satisfaction: The Gender Perspective among Police Officers." American Journal of Criminal Justice 23(1):19-31.
- Das, Suranjan, & Chattopadhyay, Basudeb. 1991. "Rural Crime in Police Perception: A Study of Village Crime Note Books." *Economic and Political Weekly* 26(3):129-135.
- Davies, Michelle, Rhiannon Smith, & Paul Rogers. 2009. "Police Perceptions of Rape as a Function of Victim Gender and Sexuality." *Police Journal* 82(1):4-12.
- Deer, Ada E. 1993. "Statement of Ada E. Deer before the Senate Committee of Indian Affairs July 15, 1993." *Wicazo Sa Review* 9(2):105-108.
- Deer, Sarah. 2006. "Native American women, domestic violence and Congress." *Color Lines: News for Action* March 5. Retrieved on January 25, 2013 (http://colorlines.com/archives/2007/07/native american women domestic.html).
- DeJong, Christina, Amanda Burgess-Proctor, and Lori Elis. 2008. "Police Officer Perceptions of Intimate Partner Violence: An Analysis of Observational Data." *Violence and Victims* 23(6):683-696.
- DiGregory, K. V. and H. A. Manuel. 1997. "Final Report of the Executive Committee for Indian Country Law Enforcement Improvements." U. S. Department of Justice. Washington DC: U.S. Government Printing Office.
- Dinges, Norman G. and & Quang Duong-Tran. 1992. "Stressful Life Events and Co-Occurring Depression, Substance Abuse and Suicidality among American Indian and Alaska Native Adolescents." *Culture, Medicine and Psychiatry* 16(4):487-502.
- Dissel, Amanda, and Kindiza Ngubeni. 2003."Giving Women Their Voice: Domestic Violence and Restorative Justice in South Africa." edited by XIth International Symposium on Victimology. Stellenbosch.
- Donnermeyer, J. F., R. W. Edwards, E. L. Chavez, and F. Beauvais. 1996. "Involvement of American Indian Youth in Gangs." *Free Inquiry in Creative Sociology* 24(2): 167-174.

- Dowler, Kenneth. 2005. "Job Satisfaction, Burnout, and Perception of Unfair Treatment: The Relationship Between Race and Police Work." *Police Quarterly* 8(4):476-489.
- Dotter, Daniel. 2002. "Creating Deviance: Scenarios of Stigmatization in Postmodern Media Culture." *Deviant Behavior 23*(5):419-448.
- Duclos, Christine Wilson, Janette Beals, Douglas K. Novins, Cheryl Martin, Chastity S. Jewett, and Spero M. Manson. 1998. "Prevalence of Common Psychiatric Disorders among American Indian Adolescent Detainees." Journal of the American Academy of Child & Adolescent Psychiatry 37(8): 866-73.
- Durkhiem, Emile. [1897] 1970. "La Conception Materialiste de Historie." in The Presentation of Self in Everyday Life, edited by J.-F. Filloux. New York: Pelican."
- Duran, B., J. Oetzel, J. Lucero, Y. Jiang, D. K. Novins, S. Manson, and J. Beals. 2005. "Obstacles for Rural American Indians Seeking Alcohol, Drug, or Mental Health Treatment." *Journal of Consulting and Clinical Psychology* 73(5):819-829.
- Duro v. Reina, 495 U.S. 676 (1990)
- Eid, Troy A., & Doyle, Carrie Covington. 2011. "Separate But Unequal: The Federal Criminal Justice System In Indian Country." University of Colorado Law Review 81(4): 1067-1117.
- Erez, Edna and Joanne Belknap. 1998. "In Their Own Words: Battered Women's Assessment of the Criminal Processing System's Responses." *Violence and Victims* 13(3):251-168.
- Ericson, Richard V. 1994. "The Division of Expert Knowledge in Policing and Security." *The British Journal of Sociology* 45(2):149-175.
- Ericson, R., and Haggerty, K. 1997. Policing the Risk Society. London: Clarendon.
- Evans-Pritchard, Deirdre. 1989. How "they" See "us": Native American Images of Tourists. Annals of Tourism Research 16(1):89-105.
- Faber, W.O., P.A. Odeen, and R.A. Tschetter. 1957. Indian, Law Enforcement and Local Government. Brookings, SD: University of South Dakota.
- Fairchild, D.G., M.W. Fairchild, and S. Stoner. 1998. "Prevalence of Adult Domestic Violence among Women Seeking Routine Care in a Native American Health Care Facility." *American Journal of Public Health* 88:1515-17.

Fanon, Frantz. 1967. Black Skin, White Masks. New York: Grove Press.

-----. 1963. The Wretched of the Earth. Grove, NY: Grove Press.

- Feagin, Joe R., and Clairece Booher Feagin. 2008. Racial and Ethnic Relations. Upper Saddle River, NJ: Pearson.
- Felson, Richard B. and Paul-Philippe Paré. 2005. "The Reporting of Domestic Violence and Sexual Assault by Nonstrangers to the Police." *Journal of Marriage and Family* 67(3):597-610.
- Fenelon, James V. 2006. "Indian Gaming: Traditional Perspectives and Cultural Sovereignty." *American Behavioral Scientist* 50(3), 381-409.
- Fernandez, Marilyn, Kichi Iwamoto, and Bernadette Muscat. 1997. "Dependency and Severity of Abuse." *Women & Criminal Justice* 9(1):39-63.
- Ferraro, K. 1995. "Cops, Courts, and Woman Battering in the United States." Pp. 155-184 in Women, Policing, and Male Violence, edited by J. Hanmer, J. Radford and E. Stanko. London: Routledge.
- ------. 1989. "The Legal Response to Women Battering in the United States." Pp. 155-184 in *Women, Policing, and Male Violence*, edited by J. Hanmer, J. Radford and E. Stanko. London: Routledge.
- Fielding, Nigel G. 1994. "Cop Canteen Culture." Pp. 46-63 in Just Boys Doing Business? Men, Masculinities and Crime edited by T. A. Newburn and E. A. Stanko. London, UK: Routledge.
- Fielding, Nigel G. and Jane Fielding. 1991. "Police Attitudes to Crime and Punishment: Certainties and Dilemmas." *British Journal of Criminology* 31(1):39-53.
- Finger, John R. 1981. "The Abortive Second Cherokee Removal, 1841-1844." *The Journal of Southern History* 47(2):207-26.
- Finn, Mary A., and Pamela Bettis. 2006. "Punitive Action or Gentle Persuasion: Exploring Police Officers' Justifications for Using Dual Arrest in Domestic Violence Cases." Violence Against Women 12(3):268-87.
- Fiske, S. 2004. Social Beings: A Core Motives Approach to Social Psychology. Jersey City, New Jersey: John Wiley & Sons.
- Fitzgerald, Robin T., and Peter J. Carrington. 2008. "The Neighborhood Context of Urban Aboriginal Crime." Canadian Journal of Criminology & Criminal Justice 50(5):523-57.

Flint, E.H. 1971. Indian Wars of the West. Cincinnati, OH: N&G Guilford and Company.

Flowers, R. B. 1990. Minorities and Criminality. Westport, CT: Greenwood.

- Fortier, L. 1975. Women, Sex and Patriarchy. Family Planning Perspectives, 7(6):278-281.
- Frank, Michael L., and David Lester. 2002. "Self-Destructive Behaviors in American Indian and Alaska Native High School Youth." *American Indian and Alaska Native Mental Health Research* 10(1):24-32.
- Franken, Al. 2010. Examining Bureau of Indian Affairs and Tribal Police Recruitment, Training, Hiring, and Retention: Before the Committee on Indian Affairs - United States Senate One Hundred Eleventh Congress. Washington, DC: U.S. Government Printing Office.
- Freedenthal, Stacey, and Arlene Rubin Stiffman. 2007. "They Might Think I Was Crazy": Young American Indians' Reasons for Not Seeking Help When Suicidal." *Journal of Adolescent Research* 22(1):58-77.
- Freiberg, Arie. 2001. "Affective Versus Effective Justice: Instrumentalism and Emotionalism in Criminal Justice." *Punishment & Society* 3(2):265-78.
- Freire, P. 1968. Pedagogy of the Oppressed. New York City: NY: The Seabury Press.
- French, Laurence A. 1980. "Anomie and Violence among Native Americans." International Journal of Comparative and Applied Criminal Justice 4(1):75-84.
 - . 1979. "The Minority Perspective on Violence." International Journal of Comparative and Applied Criminal Justice 3(1):43-49.
- French, Laurence, and Jim Hornbuckle. 1977. "An Analysis of Indian Violence: The Cherokee Example." *American Indian Quarterly* 3(4):335-56.
- Friday, Paul C., Scott Metzgar and David Walters. 1991. "Policing Domestic Violence: Perceptions, Experience, and Reality." *Criminal Justice Review* 16(2):198-213.
- Fry, Louis W. and Sue Greenfield. 1980. "An Examination of Attitudinal Differences Between Policewomen and Policemen." *Journal of Applied Psychology* 65(1):123-126.
- Gaarder, Emily and Lois Presser. 2006. "A Feminist Vision of Justice? The Problems and Posibilities of Restorative Justice for Girls and Women." Pp. 483-492 in *Handbook of Restorative Justice" A Global Perspective*, edited by Dennis Sullivan and Larry Tifft. New York, NY: Routledge.

- Gabbidon, Shaun L. 2003. "Racial Profiling by Store Clerks and Security Personnel in Retail Establishments: An Exploration of "Shopping While Black"." Journal of Contemporary Criminal Justice 19(3):345-64.
- Gabbidon, Shaun L., and George E. Higgins. 2007. "Consumer Racial Profiling and Perceived Victimization: A Phone Survey of Philadelphia Area Residents." *American Journal of Criminal Justice* 32(1-2):1-11.
- Gabbidon, Shaun L., and Helen Taylor Greene. 2012. Race and Crime. Thousand Oaks, CA: Sage. 2005.
- Gaines, L., V. Kappeler, & J. Vaughn. 1994. *Policing in America*. Cincinnati, OH: Anderson Publishing.
- Gardiner, Christine. 2012. "An Absolute Revolving Door: An Evaluation of Police Perception and Response to Proposition 36." *Criminal Justice Policy Review* 23(3): 275-303.
- Garland, David. 2001. Mass Imprisonment: Social Causes and Consequences. Thousand Oaks, CA: Sage.
- ———. 1990. Punishment and Modern Society: A Study in Social Theory. Chicago, IL: University of Chicago Press.
- Garroutte, Eva Marie, Jack Goldberg, Janette Beals, Richard Herrell, and Spero M. Manson. 2003. "Spirituality and Attempted Suicide among American Indians." Social Science & Medicine 56(7):1571-79.
- Gelles, R.J. 1993. "Through a Sociological Lens: Social Structure and Family Violence."
 Pp. 1383-1404 in *Current Controversies on Family Violence* edited by R Gelles and D Loseke. Newbury Park, CA: Sage.
- Gibson, H. B. 1969. "Early Delinquency in Relation to Broken Homes." Journal of Child Psychology and Psychiatry 10(3):195-204.
- Gillum, Tameka L. 2008. "Community Response and Needs of African American Female Survivors of Domestic Violence." *Journal of Interpersonal Violence* 23(1):39-57.
- Gilmartin-Zena, P. 1988. "Gender Differences in Students' Attitudes Toward Rape." Sociological Focus 21(4):279-292.
- Girling, Evi, Loader, Ian, and Sparks, Richard. (1998). "A Telling Tale: A Case of Vigilantism and Its Aftermath in an English Town." *The British Journal of Sociology*, 49(3),474-490.

- Glesmann, Caroline, Barry Krisberg, and Susan Marchionna. 2009. Focus: Views from the National Council on Crime and Delinquency - Youth in Gangs: Who Is at Risk?" Oakland, CA: National Council on Crime and Delinquency.
- Godfredson, Joel W., Stuart D. M. Thomas, James R. P. Ogloff, and Stefan Luebbers. 2011. "Police Perceptions of Their Encounters with Individuals Experiencing Mental Illness: A Victorian Survey." *Australian & New Zealand Journal of Criminology* 44(2):180-195.
- Goldberg, Carole, and Duane Champagne. 2007. Final Report Law Enforcement and Criminal Justice under Public Law 280. Washington DC: National Criminal Justice Reference Service..
- Goldkamp, John S. 1976. "Minorities as Victims of Police Shootings: Interpretation of Racial Disproportionality and Police Use of Deadly Force." *The Justice System Journal* 2(2):169-183.
- Goldsmith, Andrew. 2005. "Police Reform and the Problem of Trust." *Theoretical Criminology* 9(4):443-70.
- Goldstein, Herman. 1968. "Trial Judges and the Police: Their Relationships in the Administration of Criminal Justice." Crime & Delinquency 14(1):14-25.
- Gone, Joseph. 2004. "Mental Health Services for Native Americans in the 21st Century United States." *Professional Psychology: Research and Practice* 35(1):10-18.
- Gone, Joseph P., and Carmela Alcántara. 2007. "Identifying Effective Mental Health Interventions for American Indians and Alaska Natives: A Review of the Literature." *Cultural Diversity and Ethnic Minority Psychology* 13(4):356-63.
- Gould, Larry and Jeffrey Ian Ross. 2006. "Integrating the Past, Present, and Future." Pp. 237-242 in *Native Americans in the Criminal Justice System* edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- Gracia, Enrique, García, Fernando, & Lila, Marisol. 2011. "Police Attitudes Toward Policing Partner Violence Against Women: Do They Correspond to Different Psychosocial Profiles?" *Journal of Interpersonal Violence* 26(1):189-207.
- Greenberg, David F. 2012. "Assessments of Crime Seriousness on an American Indian Reservation." *Race and Justice* 2(1):29-41.
- Greenberg, David F., Ronald C. Kessler, and Charles H. Logan. 1979. "A Panel Model of Crime Rates and Arrest Rates." *American Sociological Review* 44(1):843-50.

- Greene, Jack R. 1989. "Police Officer Job Satisfaction and Community Perceptions: Implications for Community-Oriented Policing." *Journal of Research in Crime* and Delinquency 26(2):168-183.
- Greenfeld, Lawrence A., and Steven K. Smith. 1999. "American Indians and Crime." United States Department of Justice: Bureau of Justice Statistics. Washington DC: U.S. Government Printing Office.
- Greenwood, Peter W., Jan M. Chaiken, and Joan Petersilia. 1977. *The Criminal Investigation Process*. Santa Monica CA: The RAND Corporation.
- Griffiths, Curt Taylor, and Ron Hamilton. 1996."Sanctioning and Healing: Restorative Justice in Canadian Aboriginal Communities." Pp. 175-191 in *Restorative Justice: International Perspectives*, edited by Burt Galaway and Joe Hudson. Monsey, NY: Criminal Justice Press.
- Grobsmith, Elizabeth. 1994. Indians in Prison: Incarcerated Native Americans in Nebraska. Lincoln, NE: University of Nebraska Press.
- Grossman, David C., James W. Krieger, Jonathan R. Sugarman, and Ralph A. Forquera. 1994. "Health Status of Urban American Indians and Alaska Natives." *JAMA: The Journal of the American Medical Association* 271(11):845-50.
- Gruber, Aya. 2007. "The Feminist War on Crime." Iowa Law Review 92(741):803-09.
- Gruenewald, Paul J., Bridget Freisthler, Lillian Remer, Elizabeth A. LaScala, and Andrew Treno. 2006. "Ecological Models of Alcohol Outlets and Violent Assaults: Crime Potentials and Geospatial Analysis." *Addiction* 101(5):665-677.
- Haas, Theodore H. 1957. "The Legal Aspects of Indian Affairs from 1887 to 1957." Annals of the American Academy of Political and Social Science 311:12-22.
- Habermas, Jurgen. 1979. Communication and the Evolution of Society. Boston, MA: Beacon Press.
- Hagan, John, A. R. Gillis, and Janet Chan. 1978. "Explaining Official Delinquency: a Spatial Study of Class, Conflict and Control." *Sociological Quarterly* 19(3):386-398.
- Hagan, William T., 1966. Indian Police and Judges: Experiments in Acculturation and Control. New Haven CT: Yale University Press.
- Hailer, J. A., and C. B. Hart. 1999. "A New Breed of Warrior: The Emergence of American Indian Youth Gangs." *Journal of Gang Research* 7(1):23-33.

- Haining, Robert and Jane Law. 2007. "Combining Police Perceptions with Police Records of Serious Crime Areas: A Modeling Approach." Journal of the Royal Statistical Society 170(4):1019-1034.
- Hall, Edwin L., and Albert A. Simkus. 1975. "Inequality in the types of sentences received by Native Americans and Whites." *Criminology* 13(2), 199-222.
- Hakim, Simon and Andrew J. Buck. 1989. "Do casinos enhance crime?" Journal of Criminal Justice 17(5):409-416.
- Hamby, S., and M. Skupien. 1998. "Domestic Violence on the San Carlos Apache Reservation: Rates, Associated Psychological Symptoms, and Current Beliefs." *Indian Health Service Provider* 103-06.
- Harring, Sidney. 1994. Crow Dog's Case: American Indian Sovereignty, Tribal Law, and United States Law in the Nineteenth Century. Cambridge MA: Cambridge University Press.
- ———. 1983. Policing a Class Society: The Experience of American Cities, 1865-1915. New Brunswick, NJ: Rutgers University Press.
- ———. 1976. "Class Conflict and the Suppression of Tramps in Buffalo 1892-1894." Law & Society Review 11(1):873-879
- Harwell, Todd S., Kelly R. Moore, and Michael R. Spence. 2003. "Physical Violence, Intimate Partner Violence, and Emotional Abuse among Adult American Indian Men and Women in Montana." *Preventive Medicine* 37(4):297-303.
- Hawkins, D. F. 1999. "What Can We Learn from Data Disaggregation? The Case of Homicide and African Americans." Pp. 45-76 in *Homicide: A sourcebook for social research* edited by M. D. Smith and M. A. Zahn. Thousand Oaks, CA: Sage.
- Hayner, N. 1942. "Variability in the Criminal Behavior of American Indians." American Journal of Sociology 47:602-13.
- Healey, J.F. 2003. Race, Ethnicity, Gender, and Class: The Sociology of Group Conflict and Change. Thousand Oaks, CA: Pine Forge Press.
- Henderson, E., S.J., Kunitz, K.R., Gabriel, McCright, A., and J.E., Levy. 1998. "Boarding and Public Schools: Navajo Educational Attainment, Conduct

Disorder, and Alcohol Dependency." American Indian and Alaska Native Mental Health Research 8(2):28-50.

- Henson Eric C., Jonathan B. Taylor, Catherine E.A. Curtis, Stephen Cornell, Kenneth W. Grant, Miriam R. Jorgensen, Joseph P. Kalt, and Andrew J. Lee. 2007. The Harvard Project on American Indian Economic Development. The State of the Native Nations Conditions under U.S. Policies of Self-Determination. New York, NY: Oxford University Press.
- Higgins, George E., Shaun L. Gabbidon, and Kareem L. Jordan. 2008. "Examining the Generality of Citizens' Views on Racial Profiling in Diverse Situational Contexts." *Criminal Justice and Behavior* 35(12):1527-41.
- Hill, C. E., B. J. Thompson, and E. N. Williams. 1997. "A Guide to Conducting Consensual Qualitative Research." *The Counseling Psychologist* 1(25):517–572.
- Hirschel, J.D., and I.W. Hutchinson. 1992. "Female Spouse Abuse and the Police Response: The Charlotte, North Carolina, Experiment." *Journal of Criminal Law and Criminology* 83:73-119.
- Hobfoll, Stevan E., Anita Jackson, Ivonne Hobfoll, Charles A. Pierce, and Sara Young.
 2002. "The Impact of Communal-Mastery Versus Self-Mastery on Emotional Outcomes During Stressful Conditions: A Prospective Study of Native American Women." American Journal of Community Psychology 30(6):853-71.
- Hollinshead, Keith. 1992. "White' gaze, 'red' people Shadow Visions: The Disidentification of 'Indians' in Cultural Tourism." *Leisure Studies* 11(1):43-64.
- Holt, Thomas and Adam Bossler. 2012. "Police Perceptions of Computer Crimes in Two Southeastern Cities: An Examination from the Viewpoint of Patrol Officers." *American Journal of Criminal Justice* 37(3):396-412.
- Homant, Robert J. and Daniel B. Kennedy. 1985. "Police Perceptions of Spouse Abuse: A Comparison of Male and Female Officers." *Journal of Criminal Justice* 13(1):29-47.
- Hough, Mike. 2007. "Policing, New Public Management and Legitimacy in Britain." Pp. 59-89 in *In Legitimacy, Criminal Justice, and the Law*, edited by Jeffrey Fagan and Tom R. Tyler. New York, NY: Russell Sage.
- -----. 2003. "Modernization and Public Opinion: Some Criminal Justice Paradoxes." Contemporary Politics 9(2):143-155.
- ———. 2004. "Modernization, Scientific Rationalization and the Crime Reduction Program." *Criminal Justice* 4(3):439-454.

- Howard, Matthew Owen, R. Dale Walker, Patricia Silk Walker, Linda B. Cottler, and Wilson M. Compton. 1999. "Inhalant Use among Urban American Indian Youth." Addiction 94(1):83-95.
- Huang, Wilson S. and Michael S. Vaughn. 1996. "Support and Confidence: Public Attitudes Toward the Police." Edited by T. Flanagan & D. Longmire in Americans View Crime and Justice: A National Public Opinion Survey. Thousand Oaks, CA Sage Publications.
- Huff-Corzine, Lin, Jay Corzine, and David C. Moore. 1986. "Southern Exposure: Deciphering the South's Influence on Homicide Rates." *Social Forces* 64(4):906-24.
- Humphrey, John A., and Harriet J. Kupferer. 1982. "Homicide and Suicide among the Cherokee and Lubee Indians of North Carolina." *International Journal of Social Psychiatry* 28(2):121-28.
- Humphries, Drew. 2005. "No Easy Answers: Public Policy, Criminal Justice, and Domestic Violence." Criminology and Public Policy 4(3):91-6.
- Indian Health Service. 2008. *Indian Health Service Introduction*. U.S. Department of Health and Human Services. Rockville, MD: Department of Health and Human Services.
- Indian Health Service. 2007. *IHS Fact Sheet. 2000-2007*. U.S. Department of Health and Human Services. Rockville, MD: Department of Health and Human Services. Retrieved April 14, 2012 (http://info.ihs.gov/Disparities.asp).
- Indian Health Service. 2005. Facts on Indian Health Disparities U.S. Department of Health and Human Services. Rockville, MD: Department of Health and Human Services. Retrieved April 14, 2012 (http://info.ihs.gov/Health/11_DisplaritiesFacts-Jan2005.doc).
- Indian Health Service 1996. *Trends in Indian Health*. U.S. Department of Health and Human Services. Rockville, MD: Department of Health and Human Services.
- Ikner, Michael A., Janice Ahmad, and Alejandro del Carmen. 2005. "Vehicle Cues and Racial Profiling: Police Officers' Perceptions of Vehicles and Drivers." *The Southwest Journal of Criminal Justice* 2(2):82-100.
- Ioimo, Ralph, Rachel S. Tears, Leslie A. Meadows, J. Bret Becton, and Michael T. Charles. 2007. "The Police View of Bias-Based Policing." *Police Quarterly* 10(3): 270-287.

Ingrama, Jason R., and Robert R. Weidner. 2011. "Front-line Supervisors' Perceptions of Less-Lethal Force Policies: Examining the 'Transmission Belts' of Police Departments." *Journal of Crime & Justice* 34(3):221-233.

Innes, Martin. 2004. "Signal Crimes and Signal Disorders: Notes on Deviance as Communicative Action." *The British Journal of Sociology* 55(3):335-355.
International Association of Chiefs of Police. 2001. *Improving Safety in Indian Country*. Alexandria, VA: International Association of Chiefs of Police.

- Isaacs, H. 1967. A Study of Communications, Crimes, and Arrests in a Metropolitan Police Department - Task Force Report: Science and Technology: A Report to the President's Commission on Law Enforcement and Administration of Justice. Washington, DC: U.S. Government Printing Office.
- Ivan, Y. Sun and Ruth A. Triplett. 2008. "Differential Perceptions of Neighborhood Problems by Police and Residents: The Impact of Neighborhood-Level Characteristics." *Policing* 31(3):435-455.
- Jacob, Herbert. 1971. "Black and White Perceptions of Justice in the City." Law & Society Review 6(1):69-89.
- Jackson, Jonathan, and Ben Bradford. 2009. "Crime, Policing and Social Order: On the Expressive Nature of Public Confidence in Policing." *The British Journal of Sociology* 60(3):493-521.
- Jemelka, Ron, Eric W. Trupin, and John A. Chiles. 1989. "The Mentally Ill in Prisons: A Review." *Hospital and Community Psychiatry* 40(5):481-491.
- Jensen, G. F., J.H. Strauss, and V.W. Harris. 1977. "Crime, Delinquency and the American Indian." *Human Organization* 36:252-57.
- Johnson, A. G. 1997. *The Gender Knot: Unraveling Our Patriarchal Legacy*. Philadelphia, PA: Temple University Press.
- Johnson, Jeannette L. and Mark C. Cameron. 2001. "Barriers to Providing Effective Mental Health Services to American Indians." *Mental Health Services Research* 3 (4):215-23.
- Johnson, Troy, and Holly Tomren. 1999. "Helplessness, Hopelessness, and Despair: Identifying the Precursors to Indian Youth Suicide." *American Indian Culture and Research Journal* 23(3):287-301.
- Jones-Brown, Delores. 2007. "Forever the Symbolic Assailant: The More Things Change, The More They Remain The Same." *Criminology & Public Policy* 6(1):103-121.

- Joseph, Janice, and Dorothy Taylor. 2003. "Native American Youths and Gangs." Journal of Gang Research 10(2):45-54.
- Kakar, Suman. 2002. "Gender and Police Officers' Perceptions of Their Job Performance: An Analysis of the Relationship Between Gender and Perceptions of Job Performance." Criminal Justice Policy Review 13(3):238-257.
- Kakar, Suman. 2003. "Race and Police Officers' Perceptions of their Job Performance: An Analysis of the Relationship Between Police Officers' Race, Education Level, and Job Performance." *Journal of Police and Criminal Psychology* 18(1):45-56.
- Kappeler, Victor E., Richard D. Sluder, and Geoffrey P. Alpert. 1998. Forces of Deviance: Understanding the Dark Side of Policing. Prospect Heights, Illinois: Waveland Press.
- Karp, David R., and Beau Breslin. 2001. "Restorative Justice in School Communities." Youth & Society 33(2):249-72.
- Kebbell Mark R. and M.E. O'Kelly Caitriona. 2007. "Police Detectives' Perceptions of Giving Evidence in Court." *Policing: An International Journal of Police Strategies & Management* 30(1):8–20.
- Keeling, Deborah, and Tom Hughes. 2011. "Police Officer Attitudes Toward Muslims and Islam: 'Worlds Apart'?" American Journal of Criminal Justice 36(4):307-318.
- Kelling, George L., Tony Pate, Duane Dieckman, and Charles E. Brown. 1974. *The Kansas City Preventive Patrol Experiment*. Washington, DC: The Police Foundation.
- Kelly, Lawrence C. 1975. "The Indian Reorganization Act: The Dream and the Reality." *Pacific Historical Review* 44(3):291-312.
- Khoury, Laura J. 2009. "Racial profiling as dressage: a social control regime!" African Identities 7(1), 55-70.
- Kidd-Hewitt, D. 1995. "Crime and the Media: A Criminological Perspective." Pp. 9-24 edited by D. Kidd-Hewitt & R. Osborne in *Crime and the Media: The Post-Modern Spectacle* (pp. 9-24). London, UK: Pluto.
- King, J. C. 1999. First People, First Contacts: Native Peoples of North America. Cambridge, MA: Harvard University Press.
- Klinger, David A. 1997. "Negotiating Order in Patrol Work: An Ecological Theory of Police Response to Deviance." *Criminology* 35(2):277-306.

- Klockars, Carl B., and Stephen Mastrofski. 1991. *Thinking About Police: Contemporary Readings*. Highstown, NJ: McGraw-Hill.
- Kort-Butler, Lisa A. 2012. "Rotten, Vile, and Depraved! Depictions of Criminality in Superhero Cartoons." *Deviant Behavior* 33(7):566-581.
- Knepper, P., and M. Puckett. 1995. "The Historicity of Hillerman's Indian Police." Journal of the West 33:13-18.
- Kraska, Peter B. 2007. "Militarization and Policing—Its Relevance to 21st Century Police." *Policing* 1(4):501-513
- Kraska, Peter B. and Louis J. Cubellis. 1997. "Militarizing Mayberry and Beyond: Making Sense of American Paramilitary Policing." Justice Quarterly 14(4):607-629.
- Kraska, Peter B. and Victor E. Kappeler. 1997. "Militarizing American Police: The Rise and Normalization of Paramilitary Units." *Social Problems* 44(1):1-18.
- Kubrin, Charis E. 2005. "Gangstas, Thugs, and Hustlas: Identity and the Code of the Street in Rap Music." *Social Problems* 1(52):360-378.
- Kunitz, S. J., K. R. Gabriel, J. E. Levy, E. Henderson, K. Lampert, J. McCloskey, G. Quintero, S. Russell, and A. Vince. 1999. "Risk Factors for Conduct Disorder among Navajo Indian Men and Women." Social Psychiatry and Psychiatric Epidemiology 34(4):180-89.
- LaFave, Wayne R. and Frank J. Remington. 1965. "Controlling the Police: The Judge's Role in Making and Reviewing Law Enforcement Decisions." *Michigan Law Review* 63(6):987-1012.
- Lanier, Christina, and Lin Huff-Corzine. 2006. "American Indian Homicide." *Homicide Studies* 10(3):181-94.
- LaPrairie, Carol. 2002. "Aboriginal over-Representation in the Criminal Justice System: A Tale of Nine Cities." *Canadian Journal of Criminology* 44(2):181-208.
- -----. 1999. "The Impact of Aboriginal Justice Research on Policy: A Marginal Past and an Even More Uncertain Future." *Canadian Journal of Criminology* 41(2): 249-60.
- ------. 1997. "Reconstructing Theory: Explaining Aboriginal over-Representation in the Criminal Justice System in Canada." *Australian & New Zealand Journal of Criminology* 30(1):39-54.

------. 1995. "Community Justice or Just Communities? Aboriginal Communities in Search of Justice." *Canadian Journal of Criminology* 37(4):521-45.

- Lauriello, John, Juan Bustillo, and Samuel J. Keith. 2000. "Can Intensive Psychosocial Treatments Make a Difference in a Time of Atypical Antipsychotics and Managed Care?" *Schizophrenia Bulletin* 26(1):141-144.
- Lee, Nella. 1995. "Culture Conflict and Crime in Alaskan Native Villages." Journal of Criminal Justice 23(2):177.
- -----. 1993. "Native American Crime: The Invisible Tragedy." *Wicazo Sa Review* 9 (1):38-46.
- Lester, David. 1999. Crime and the Native American. Springfield, IL: Charles C. Thomas Publisher.
- Levi, M. and S. Jones. 1985. "Public and Police Perceptions of Crime Seriousness in England and Wales." *British Journal of Criminology* 25(3):234-250.
- Levy, J. 2000. The Multiculturalism of Fear. Oxford, UK: Oxford University Press.
- Levy, Jerrold E., and Stephen J. Kunitz. 1987. "A Suicide Prevention Program for Hopi Youth." Social Science & Medicine 25(8):931-940.
- Like, Toya Z. and Jody Miller. 2006. "Race, Inequality and Gender Violence: A Contextual Examination." Pp. 157-177 in *The Many Colors of Crime: Inequalities of Race, Ethnicity and Crime in America*, edited by Ruth Peterson, Lauren Krivo and John Hagan. New York, NY: New York University Press.
- Liu, Peter, & Thomas Cook. 2005. "Speeding Violation Dispositions in Relation to Police Officers' Perception of the Offenders." *Policing & Society* 15(1):83-88.
- Livingston, Michael. 2008. "Alcohol Outlet Density and Assault: A Spatial Analysis." *Addiction* 103(4):619-628.
- Loader, Ian. 2006. "Policing, Recognition, and Belonging." The Annals of the American Academy of Political and Social Science 605(1):201-21.
 - —. 1997. "Policing and the Social: Questions of Symbolic Power." The British Journal of Sociology 48(1):1-18.
- Loader, Ian, and Aogan Mulcahy. 2003. Policing and the Condition of England: Memory, Politics and Culture. Oxford: Oxford University Press.
- Logan, Charles H. 1975. "Arrest Rates and Deterrence." Social Science Quarterly 55: 376-89.

- Logan, T. K., Lisa Shannon, and Robert Walker. 2006. "Police Attitudes Toward Domestic Violence Offenders Journal of Interpersonal Violence." *Journal of Interpersonal Violence* 21(10):1365-1374.
- Logan, T.K., Lisa Shannon, and Robert Walker. 2005. "Protective Orders in Rural and Urban Areas: A Multiple Perspective Study." Violence Against Women 11(7): 876-911.
- Lopez, Natalie and Chris Lukinbeal. 2010. "Comparing Police and Residents Perceptions of Crime in a Phoenix Neighborhood using Mental Maps in GIS." Yearbook of the Association of Pacific Coast Geographers 1(72):33-55.
- Lujan, Carol Chiago. 2006. U.S. Politicians and the Prevalent of Stereotyping of American Indian and Alaska Native Nations and People. Pp. 130-140 edited by Coramae Richey Mann, Majorie S. Zatz & Nancy Rodriguez in *Images of Color*, *Images of Crime* (pp. 130-140). New York, NY: Oxford University Press.
- Lujan, Carol Chiago. 1993. "A Sociological View of Tourism in an American Indian Community: Maintaining Cultural Integrity at Taos Pueblo." *American Indian Culture and Research Journal* 17(3):101-120.
- Luken, Jennifer N., Rhonda R. Dobbs, and Alejandro del Carmen. 2006. "Vehicle Cues and Perceptions of Driver Characteristics: A Comparative Analysis of Police Officers and College Students." Southwest Journal of Criminal Justice 3(2):127-147.
- Luna, Eileen, 1998. "The Growth and Development of Tribal Police: Challenges and Issues for Tribal Sovereignty." *Journal of Contemporary Criminal Justice* 14 (1):75-86.
- Luna-Firebaugh, Eileen. 2002. "Women in Tribal Policing: An Examination of Their Status and Experiences." *The Social Science Journal* 39(4):583-92.
- Luna-Firebaugh, Eileen, and Samuel Walker. 2006. "Law Enforcement and the American Indian: Challenges and Obstacles to Effective Law Enforcement." Pp. 117-135 in Native Americans and the Criminal Justice System, edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- Lung-Teng, Hu. 2010. "Same Bed, But Different Dreams? Comparing Retired and Incumbent Police Officers' Perceptions of Lost Knowledge and Transfer Mechanisms." Crime, Law & Social Change 53(4):413-435.
- Lusane, C. 1991. *Pipe Dream Blues: Racism and the War on Drugs*. Boston, MA: South End Press.

- Lynch, Michael J., and Raymond Michalowski. 2006. Primer in Radical Criminology: Critical Perspectives on Crime, Power, and Identity. Boulder, CO: Lynne Rienner Publishers.
- Maanen, John Van. 1974. "Working the Street." Pp. 83-130 edited by Herbert Jacob in *The Potential for the Reform of Criminal Justice* (pp. 83-130). Beverly Hills, CA: Sage Criminal Justice System Annuals.
- MacMeekin, Daniel H. 1969. "Red, White, and Gray: Equal Protection and the American Indian." *Stanford Law Review* 21(5):1236-48.
- Magen, Randy H., and Darryl S. Wood. 2006. "Intimate Partner Violence against Ahtna (Alaska Native) Women in the Copper River Basin." National Institute of Justice U.S. Department of Justice. Washington DC: U.S. Government Printing Office.
- Mahan, Sue. 1996. Crack Cocaine, Crime, and Women: Legal, Social, and Treatment Issues. Thousand Oaks, CA: Sage.
- Maher, Timothy M. 2003. "Police Sexual Minconduct: Officer's Perceptions of it Extent and Causality." Criminal Justice Review 28(2):355-381.
- Major, Aline K., Jr. Arlen Egley, James C. Howell, Barbara Mendenhall, and Troy Armstrong. 2004. "Youth Gangs in Indian Country." U.S. Department of Justice Office of Juvenile Justice and Delinquency Prevention, Office of Justice Programs. Washington DC: U.S. Government Printing Office.
- Malcoe, Lorraine, Bonnie Duran, and Juliann Montgomery. 2004. "Socioeconomic Disparities in Intimate Partner Violence against Native American Women: A Cross-Sectional Study." *BMC Medicine* 2(1):1-20.
- Malley-Morrison, Kathleen, and Denise A. Hines. 2004. Family Violence in a Cultural Perspective: Defining, Understanding, and Combating Abuse. Thousand Oaks, CA: Sage.
- Manning, Peter K. 2001. "Theorizing Policing: The Drama and Myth of Crime Control in the NYPD." *Theoretical Criminology* 5(3):315-344.
- -----. 1997. Police Work: The Social Organization of Policing (3rd). Long Grove, IL: Waveland Press.
- . 1995. "The Police Occupational Culture in Anglo-American Societies." Pp. 472-475 edited by W. Bailey in *The Encyclopedia of Police Science*. New York City, NY: Garland Publishing.
- ------. 1977. Police Work: The Social Organization of Policing. Cambridge, MA: MIT Press.

- Manson, Spero M., Lynn M. Ackerson, Rhonda Wiegman Dick, Anna E. Baron, and Candace M. Fleming. 1990. "Depressive Symptoms Among American Indian Adolescents: Psychometric Characteristics of the Center for Epidemiologic Studies Depression Scale (CES-D)." Psychological Assessment: A Journal of Consulting and Clinical Psychology 2(3):231-237.
- Maris, Ronald W., Alan Lee Berman, Morton M. Silverman, and Bruce Michael Bongar. 2000. *Comprehensive Textbook of Suicidology*. New York City, NY: Guilford Publications.
- Marshall, Tony. 1996. "The Evolution of Restorative Justice in Britain." European Journal on Criminal Policy and Research 4(4):21-43.
- Martin, S. 1989. "Women in Policing: The Eighties and Beyond." Pp. 53-75 edited by D. Kenney in *Police and Policing Contemporary Issues*. New York: Praeger.
- Marvell, Thomas, and Carlisle Moody. 1996. "Specification Problems, Police Levels, and Crime Rates." Criminology 34(4):609-46.
- Massey, Douglas S. 2007. Categorically Unequal: The American Stratification System. New York, NY: Russell Sage Foundation.
- May, Philip A. 1987. Suicide and Self-destruction among American Indian Youth. American Indian and Alaska Native Mental Health Research 1(1):58-69.
- May, Philip A. 1982. "Contemporary Crime and the American Indian: A Survey and Analysis of the Literature." *The Plains Anthropologist* 27(97):225-38.
- McCaskill, Joseph C. 1938. "Juvenile Delinquency among Indians." *Federal Probation* 2(1):31-40.
- McEwan, Bonnie G. 2000. Indians of the Greater Southeast: Historical Archaeology and Ethnohistory. Gainsville, FL: University Press of Florida.
- McLaughlin, Lindsay M., Shane D. Johnson, Kate J. Bowers, Dan J. Birks and Ken Pease. 2007. Police Perceptions of the Long- and Short-term Spatial Distribution of Residential Burglary. *International Journal of Police Science & Management*, 9(2):99-111.
- McNamara, John. 1967. "Uncertainties in Police Work: The Relevance of Police Recruits' Background and Training." In David J. Bordua, *The Police: Six* Sociological Essays. New York City. NY: John Wiley and Sons.

- Meehan, Albert J., and Michael C. Ponder. 2002. "Race and Place: The Ecology of Racial Profiling African American Motorists." *Justice Quarterly* 19(3):399-430.
- Memmott, Paul. 2010. "On Regional and Cultural Approaches to Australian Indigenous Violence." Australian & New Zealand Journal of Criminology 43(2):333-355.
- Menard, Anne E., and Anthony J. Salius. 1990. "Judicial Response to Family Violence: The Importance of Message." *Mediation Quarterly* 7(1):293-302.
- Meriam Report. 1928. The Problem of Indian Administration. Institute for Government Research. Washington, DC: U. S. Government Printing Office.
- Meyer, Jessica R., and N. Dickon Reppucci. 2007. "Police Practices and Perceptions Regarding Juvenile Interrogation and Interrogative Suggestibility." *Behavioral Sciences & the Law* 25(6):757-780.
- Mezey, Naomi. 1996. "The Distribution of Wealth, Sovereignty, and Culture Through Indian Gaming." *Stanford Law Review* 48(3):711-737.
- Michalowski, Raymond. 1985. Order, Law, and Crime an Introduction to Criminology. Westminister, MD: Random House.
- Miller, J. and M. D. Schwartz. 1995. "Rape Myths and Violence Against Street Prostitutes." *Deviant Behavior* 16(1):1-23.
- Million, D. 2000. "Policing the Rez: Keeping No Peace in Indian Country." Social Justice 27(3):101-119.
- Minnis, Mhyra S. 1963. "The Relationship of the Social Structure of an Indian Community to Adult and Juvenile Delinquency." *Social Forces* 41(4):395-403.

____. "Selected Social Problems of Fort Hall Reservation." Sociology and Social Research 46(1):436-45.

- Mitchell, D., R. Hirschman, and G. C. Nagayama Hall. 1999. "Attributions of Victim Responsibility, Pleasure and Trauma in Male Rape." *The Journal of Sex Research*, 36(4):369-373.
- Mitchell, Christina M., Douglas K. Novins, and Tamara Holmes. 1999. "Marijuana Use among American Indian Adolescents: A Growth Curve Analysis from Ages 14 through 20 Years." Journal of the American Academy of Child & Adolescent Psychiatry 38(1):72-78.
- Moore, Evan. 1990. "Gambling, Murder Split Reservation/ It's Nothing but a Damn War Zone." *Houston Chronicle*, May 6, 1990. Section1, Page 15.

- Moran, Rebecca, and Bobby D. Rampey. 2007. "National Indian Education Study Part Ii: The Educational Experiences of American Indian and Alaska Native Students in Grades 4 and 8." Institute of Education Sciences National Center for Education Statistics, U.S. Department of Education. Washington, DC: U.S. Government Printing Office.
- Morris, Gregory D., and Peter B. Wood. 2010. "Delinquency and Non-Reservation American Indians: The Role of Unique and General Predictors of Interpersonal and Property Offending." *Journal of Ethnicity in Criminal Justice* 8(4):248-65.
- Morris, Gregory D., Peter B. Wood, and R. Gregory Dunaway. 2006. "Self-Control, Native Traditionalism, and Native American Substance Use: Testing the Cultural Invariance of a General Theory of Crime." *Crime & Delinquency* 52(4):572-98.
- National Congress of American Indians. 2006. Methamphetamine in Indian Country: An American Problem Uniquely Affecting Indian Country. Arlington, VA: National Congress of American Indians
- National Institute on Drug Use. 2003. "Drug Use among Racial and Ethnic Minorities." U.S. Department of Health and Human Services: National Institute of Health. Washington, DC: U.S. Government Printing Office.
- National Indian Gaming Commission. "National Indian Gaming Commission Annual Report, 2004." Washington, DC: U. S. Government Printing Office.
- National Institute of Justice. 2005. "American Indian Suicides in Jail: Can Risk Screening Be Culturally Sensitive?" United States Department of Justice, Office of Justice Programs. Washington, DC: U.S. Government Printing Office.
- National Survey on Drug Use and Health Report. 2011. "Results from the 2011 National Survey on Drug Use and Health: Summary of National Findings." Washington, DC: U.S. Government Printing Office.
- National Victim Assistance Academy. 1999. "Tribal Justice. Specific Justice Systems and Victim's Rights." Office of Justice Programs. U.S. Department of Justice, National Institute of Justice. Washington, DC: U.S. Government Printing Office.
- Newburn, Tim and Trevor Jones. 2007. "Symbolizing Crime Control: Reflections on Zero Tolerance." *Theoretical Criminology* 11(2):221-243.
- Nielsen, Marianne O. 1996. "Major Issues in Native American Involvement in the Criminal Justice System." Pp. 293-302 in *Native Americans, Crime, and Justice* edited by Marianne O. Nielsen & Robert A. Silverman. Boulder, CO: Westview Press.

- Nielsen, Marianne O., and Robert A. Silverman. 1996. Native Americans, Crime, and Justice. Boulder, CO: Westview Press.
- Niethammer, C. 1997. Daughters of the Earth: The Lives and Legends of American Indian Women. New York, NY: Macmillan.
- Nilson, Chad and Willard Oliver. 2006. "Changes in Latitude, Changes in Attitude: An Examination of Police Officer Perceptions of Police Effectiveness among Select Police Agencies in Canada, Venezuela, and the USA." *Police Practice & Research* 7(3):231-247.
- Noblet, A. J., J. J. Rodwell, and A. F. Allisey. 2009. "Police Stress: The Role Of the Psychological Contract and Perceptions of Fairness." *Policing* 32(4):613-630.
- Noriega, J. 1992. "American Indian Education in United States: Indoctrination for Subordination to Colonialism." Pp. 371-402 in *The State of Native America: Genocide, Colonization, and Resistance* edited by M.A. Jaimes. Boston, MA: South End Press.
- Norris, Tina, Paula L. Vines, and Elizabeth M. Hoeffel. 2012. "The American Indian and Alaska Native Population: 2010." U.S. Census Bureau. Washington, DC: U.S. Government Printing Office.
- Norton, Ilena M., and Spero M. Manson. 1995. "A Silent Minority: Battered American Indian Women." *Journal of Family Violence* 10(3):307-18.
- Novins, Douglas K., Janette Beals, James H. Shore, and Spero M. Manson. 1996.
 "Substance Abuse Treatment of American Indian Adolescents: Comorbid Symptomatology, Gender Differences, and Treatment Patterns." *Journal of the American Academy of Child & Adolescent Psychiatry* 35(12):1593-601.
- Nye, F. Ivan. 1958. Family Relationships and Delinquent Behavior. New York, NY: Wiley.
- Oakley, Christopher Arris. 2001. "Indian Gaming and the Eastern Band of Cherokee Indians." *The North Carolina Historical Review* 1(2):133-155.
- O'Brien, Greg. 2011. Chronology of Native American. London: Amber Books.
- O'Brien, Michael J. 1977. "Children: Indian Juveniles in the State and Tribal Courts of Oregon." American Indian Law Review 5(2):343-67.
- Oetting, E. R., R. C. Swaim, R. W. Edwards, and F. Beauvais. 1989. "Indian and Anglo Adolescent Alcohol Use and Emotional Distress: Path Models." *The American Journal of Drug and Alcohol Abuse* 15(2):153-72.

- Ogunwole, Stella U. 2006. "We the People: American Indians and Alaska Natives in the United States." U.S. Census Bureau. Washington, DC: U.S. Government Printing Office.
- Okwumabua, Jebose O., and Elias J. Duryea. 1987. "Age of Onset, Periods of Risk, and Patterns of Progression in Drug Use among American Indian High School Students." Substance Use & Misuse 22(12):1269-76.
- Onwudiwe, Ihekwoaba. 2005. "Defining Terrorism, Racial Profiling and the Demonization of Arabs and Muslims in the USA." *Safer Communities* 4(2):4-11.
- Oppenlander, N. 1982. "Coping or Copping Out? Police Service Delivery in Domestic Disputes." 1982. Criminology 20(3-4):449-65.
- Orfield, Gary, and Chungmei Lee. 2005. "Why Segregation Matters: Poverty and Educational Inequality." The United States Civil Rights Department. Cambridge, MA.
- Pagano, Marc R. and Stuart P. Taylor. 1980. "Police Perceptions of Alcohol Intoxication." Journal of Applied Social Psychology 10(2):166-174.
- Page, Amy Dellinger. 2008. "Judging Women and Defining Crime: Police Officers' Attitudes Toward Women and Rape." *Sociological Spectrum* 28(4):389-411.
- Pagelow, M. D. 1981. Woman Battering: Victims and Their Experiences Beverly Hills, CA: Sage.
- Painter-Davis, Noah. 2012. "Structural Disadvantage and American Indian Homicide and Robbery Offending." *Homicide Studies* 16(3):219-37.
- Paoline, Eugene A. 2004. "Shedding Light on Police Culture: An Examination of Officers' Occupational Attitudes." *Police Quarterly* 7(2):205-236.
- Paoline, Eugene A., Stephanie M. Myers, and Robert E. Worden. 2000. "Police Culture, Individualism, and Community Policing: Evidence from Two Police Departments." Justice Quarterly 17(3):575-605.
- Parker, E.S. 1969. Annual Report to the Commissioner of Indian Affairs. Bureau of Indian Affairs. Washington DC: U.S. Government Printing Office.
- Parker, Karen F. 2001. "A Move toward Specificity: Examining Urban Disadvantage and Race-and Relationship-Specific Homicide Rates." Journal of Quantitative Criminology 17(1):89-110.
- Parmentier, Stephan and Elmar G.M. Weitekamp. 2005. "The Truth and Reconciliation Commission in South Africa" Pp. 151-159 in: *Introduction to International Criminal Justice* edited by Mangai Natarajan. New York, NY: McGraw Hill.

- Parmentier, Stephan and Elmar G.M. Weitekamp. 2007. "Political Crimes and Serious Violations of Human Rights: Towards Criminology of International Crime." Pp. 109-146 in Crime and Human Rights, A Volume in the Sociology of Crime, Law and Deviance Series edited by Stephan Parmentier and Elmar G.M. Weitekamp Oxford: Elsevier Books.
- Patel, Reshma N., L. J. David Wallace, and Len Paulozzi. 2005. "Atlas of Injury Mortality among American Indian and Alaska Native Children and Youth 1989-1998." Center for Disease Control and Prevention (CDC) and National Center for Injury Prevention and Control. Washington, DC: U.S. Department of Health and Human Services.
- Paulsen, D. 2004. "To Map Or Not To Map: Assessing the Impact of Crime Maps on Police Officer Perceptions of Crime." International Journal of Police Science & Management 6(4):234-246.
- Payne, Brian K. and Victoria M. Time. 2005. "Perceptions About Changing the Miranda Warnings Among Police Chiefs The Role of Race." Journal of Ethnicity in Criminal Justice 3(4):69-80.
- Peak, Ken. 1989. "Criminal Justice, Law, and Policy in Indian Country: A Historical Perspective." *Journal of Criminal Justice* 17(5):393-407.
- Peak, K., and J. Spencer. 1987. "Crime in Indian Country: Another Trail of Tears." Journal of Criminal Justice 17(1):485-94.
- Pearson, Noel. 2001. "On the Human Right to Misery, Mass Incarceration and Early Death." In Dr Charles Perkins Memorial Oration. Sydney, Australia: University of Sydney. Retrieved March 9, 2013 (http://sydney.edu.au/koori/news/pearson.pdf).
- Pelikan, Christa. 2000. "Victim Offender Mediation in Domestic Violence Cases A Research Report." United Nations Crime Congress. Vienna, Austria: Ancillary Meeting.
- Peltier, Leonard. 2000. Prison Writings: My Life Is My Sun Dance. New York, NY: St. Martin's Griffin.
- Pencak, William A., and Daniel K. Richter. 2004. Friends and Enemies in Penn's Woods: Indians, Colonists, and the Racial Construction of Pennsylvania. University Park, PA: The Pennsylvania State University Press.
- Pennell, Joan, and Gale Burford. 1997. "Family Group Decision Making: After the Conference-Progress in Resolving Violence and Promoting Well-Being." St.

John, Newfoundland: Memorial University of Newfoundland School of Social Work.

Perdue, Theda. 1989. The Cherokee. New York, NY: Chelsea House Publishers.
——. 1985. Native Carolinians: The Indians of North Carolina. Division of Archives and History. Raleigh, NC: North Carolina Department of Cultural Resources.

- Perdue, Theda, and Michael D. Green. 2005. The Cherokee Removal: A Brief History with Documents. Boston, MA: Bedford.
- Perlmutter, D. 2000. Policing the Media: Street Cops and Public Perceptions of Law Enforcement. Thousand Oaks, CA: Sage.
- Perry, Barbara. 2009. Policing and Place in Indian Country. Lanham, MD: Lexington Books.
- ------. 2008. Silent Victims: Hate Crimes against Native Americans: Tucson, AZ: University of Arizona Press.
- -----. 2006. "Nobody Trusts Them! Under- and over-Policing Native American Communities." *Critical Criminology* 14(4):411-44.
- ———. 2002. "From Ethnocide to Ethnocide: Layers of Native American Victimization." *Contemporary Justice Review* 5(3):231–247.
- Perry, B. D., and R. Pollard. 1998. "Homeostasis, Stress, Trauma, and Adaptation: A Neurodevelopmental View of Childhood Trauma." *Child and Adolescent Psychiatric Clinics of North America* 7(1):33-51.
- Perry, Steven W. 2004. "American Indians and Crime." U.S. Department of Justice, Bureau of Justice Statistics. Washington, DC: U.S. Government Printing Office.
- Peterson, Ruth D. 2012. "The Central Place of Race in Crime and Justice the American Society of Criminology's 2011 Sutherland Address." *Criminology* 50(2):303-28.
- Peterson, Richard R., and Jo Dixon. 2005. "Court Oversight and Conviction under Mandatory and Non-mandatory Domestic Violence Case Filing Policies." *Criminology and Public Policy* 4(3):535-58.
- Pevar, Stephen. 2004. The Rights of Indians and Tribes: The American Civil Liberties Union. Carbondale, IL: Southern Illinois University Press.
- Pfohl, Stephen. 2009. Images of Deviance and Social Control: A Sociological History. Long Grove, IL: Waveland Press Inc.

------. 1994. Images of Deviance and Social Control: A Sociological History. Long Grove, IL: Waveland Press Inc.

- Pharris, Margaret Dexheimer, Michael D. Resnick, and Robert W. Blum. 1997.
 "Protecting against Hopelessness and Suicidality in Sexually Abused American Indian Adolescents." *Journal of Adolescent Health* 21(6):400-406.
- Pickering, Kathleen. 2000. "Alternative Economic Strategies in Low-Income Rural Communities: TANF, Labor Migration, and the Case of the Pine Ridge Indian Reservation." *Rural Sociology* 65(1):148-167.
- Pogrebin, Mark R., and Eric D. Poole. 1990. "Culture Conflict and Crime in the Korean-American Community." Criminal Justice Policy Review 4(1):69-78.
- Pommersheim, Frank. 1995. Braid of Feather: American Indian Law and Contemporary Tribal Life. Berkeley, CA: University of California Press.
- Porter, Robert B. 2001. "Indian Gaming Regulation: A Case Study in Neo-Colonialism." Gaming Law Review 5(4):299-309.
- Powell, William S. 2006. *Encyclopedia of North Carolina*. Chapel Hill, NC: University of North Carolina Press.
- Pratt, Richard H. 1964. Battlefield and Classroom. New Haven: Yale University Press.
- _____. 1910. "The Indian: No Problem." Missionary Review of the World 33(1):6.
- Presser, Lois, and Emily Gaarder. 2000. "Can Restorative Justice Reduce Battering? Some Preliminary Considerations." *Social Justice* 27(79):175-95.
- Pridemore, William Alex. 2004. "Review of the Literature on Risk and Protective Factors of Offending among Native Americans." *Journal of Ethnicity in Criminal Justice* 2(4):45-63.
- Prucha, Francis Paul. 1986. The Great Father: The United States Government and the American Indian. Lincoln, NB: University of Nebraska Press.
- Quinn, James F. and William Downs. 1993. "Police Perceptions of the Severity of Local Gang Problems: An Analysis of Noncriminal Predictors." *Sociological Spectrum* 13(2):209-226.
- Quintero, Gilbert. 2001. "Making the Indian: Colonial Knowledge, Alcohol, and Native Americans." American Indian Culture and Research Journal 25(4):57-71.
- Rahav, Giora. 1981. "Culture Conflict, Urbanism, and Delinquency." Criminology 18(4): 174-91.

- Ray, Ross. 2008. "Issues in Restorative Justice." In Speech given to the Queensland Law Society Forum - Law and Order Issues in the 21st Century. Queensland, Australia: Queensland Law Society Forum 1(2)1-10.
- Reasons, C. 1972. "Crime and the American Indian." Pp. 78-93 in *Native Americans Today*, edited by H.M. Bahr, B.A. Chadwick and R.C. Day. New York, NY: Harper and Row.
- Reaves, Brian A. 2008. "Tribal Law Enforcement" Rockville, MD: Bureau of Justice Statistics, US Dept of Justice, Office of Justice Programs. Washington, DC: U.S. Government Printing Office.
- Reckless, Walter. 1961. "A New Theory of Delinquency and Crime." *Federal Probation* 25(1):42-46.
- Reiner, Robert. 2000. The Politics of the Police Oxford: Oxford University Press.
- Reisig, Michael D. 2000. The Politics of the Police. Oxford: Oxford University Press.
- Reiss, Albert J. 1951. "Delinquency as the Failure of Personal and Social Controls." American Sociological Review 16(2):196-207.
- Reyhner, Jon, and Jeanne Eder. 2004. American Indian Education: A History. Norman, OK: University of Oklahoma Press.
- Rigakos, G. S. 1997. "Situational Determinants of Police Responses to Civil and Criminal Injunctions for Battered Women." *Violence Against Women* 1(3):204 -216.
- Riley, Michael. 2007a. "Promises, Justice Broken: A Dysfunctional System Lets Serious Reservation Crimes Go Unpunished and Puts Indians at Risk." *The Denver Post*. Retrieved on February 4, 2013 (http://www.denverpost.com/ci_7429560).
- 2007b. "Justice: Inaction's Fatal Price: Delays and Missteps in Indian Country Criminal Cases Can Give Offenders a Free Pass. The Consequences Can Be Tragic." *The Denver Post*. Retrieved January 15, 2013 (http://www.denverpost.com/ci_7437278?source=pkg).
- Ritzer, G. 1992. Metatheorizing. Thousand Oaks, CA: Sage.
- Roberg, R. 1978. "An Analysis of the Relationships Among Higher Education, Belief Systems, and Job Performance of Patrol Officers." *Journal of Police Science and Administration* 6(1):336-344.

- Robinson, Amanda L. and Meghan S. Chandek. 2000. "The Domestic Violence Arrest Decision: Examining Demographic, Attitudinal, and Situational Variables." *Crime & Delinquency* 46(1):18-37.
- Robinson, Amanda L. and Meghan S. Chandek. 2000. "Differential Police Response to Black Battered Women." *Women & Criminal Justice* 12(2-3):29-61.
- Ross, Jeffrey Ian, and Larry Gould. 2006. "Native Americans, Criminal Justice, Criminological Theory, and Policy Development." Pp. 3-14 in *Native Americans in the Criminal Justice System* edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- Ross, Luana. 1998. Inventing the Savage: The Social Construction of Native American Criminality. Austin, TX: University of Texas Press.
- Rowe, Kelly. 1985. "The Limits of the Neighborhood Justice Center: Why Domestic Violence Cases Should Not Be Mediated." *Emory Law Journal* 34(1):855-910.
- Rubenstein, Jonathan. 1973. City Police. New York City, NY: Farrar, Straus and Giroux.
- Ruiz, Jim and Chad Miller. 2004. "An Exploratory Study of Pennsylvania Police Officer's Perceptions of Dangerousness and their Ability to Manage Persons with Mental Illness." *Police Quarterly* 7(3):359-371.
- Russell-Brown, Katheryn. 1998. The Color of Crime: Racial Hoaxes, White Fear, Black Protectionism, Police Harassment and Other Macroaggressions. New York: New York University Press.
- Sale, Kirkpatrick. 1990. The Conquest of Paradise: Christopher Columbus and the Columbian Legacy New York, NY: Knopf Publishing.
- Salisbury, Neal. 1996. "The Indians' Old World: Native Americans and the Coming of Europeans." *The William and Mary Quarterly* 53(3):435-58.
- Sanchez, G. J. 1993. Becoming Mexican American: Ethnicity, Culture, and Identity in Chicano Los Angeles, 1900-1945. New York, NY: Oxford University Press.
- Sandefur, G. D. 1989. "American Indian Reservations: The First Underclass Areas?" *Focus* 12(1):37-41.
- Satcher, David. 2001. Testimony of David Satcher, Assistant Secretary for Health and Surgeon General, U.S. Public Health and U.S. Department of Health and Social Services: Before the House Commerce Committee, Subcommittee on Health and Environment Retrieved March 10, 2013 (http://www.hhs.gov/asl/testify/t000511a.html).

- Savitz, L. 1971. Socialization of the Police." Report submitted to Pennsylvania Criminal Justice Planning Board. Harrisburg, PA: Pennsylvania State Press.
- Saunooke, Robert O. 2009. "Native Americans and the Federal Bench the Time Has Come." *The Judge's Journal* 48(1):25-29.
- Saylors, K., and N. Daliparthy. 2006. "Violence against Native Women in Substance Abuse Treatment: American Indian and Alaska Native Mental Health Research." *The Journal of the National Center* 13(1):32-51.
- Schafer, Joseph A., and Matthew J. Giblin. 2010. "Policing Intimate Partner Violence in Rural Areas and Small Towns: Policies, Practices, and Perceptions." Women & Criminal Justice 20(4):283-301.
- Sherman, Lawrence. 1995. "The Police." Pp. 327-348 in *Crime*, edited by James Q. Wilson and Joan Petersilia. San Francisco CA: ICS Press.
- Sherman, Lawrence W., and Richard A. Berk. 1984. "The Specific Deterrent Effects of Arrest for Domestic Assault." *American Sociological Review* 49(2):261-72.
- Sherman, Lawrence W., and David Weisburd. 1995. "General Deterrent Effects of Police Patrol in Crime "Hot Spots": A Randomized, Controlled Trial." *Justice Quarterly* 12(4):625-48.
- Shichor, David L. Decker, and Robert M. O'Brien. 1979. "Population Density and Criminal Victimization: Some Unexpected Findings in Central Cities." *Criminology* 17(2):184-193.
- Shoemaker, Donald J., and J. Sherwood Williams. 1987. "The Subculture of Violence and Ethnicity." *Journal of Criminal Justice* 15(6):461-72.
- Shoham, Shlomo. 1962. "The Application of The "Culture-Conflict" Hypothesis to the Criminality of Immigrants in Israel." *The Journal of Criminal Law, Criminology, and Police Science* 53(2):207-14.
- Skolnick, Jerome H. 1966. Justice without Trial: Law Enforcement in Democratic Society. London: Wiley.
- Sievers, M. L. 1968. Cigarette and Alcohol Usage by Southwestern American Indians. American Journal of Public Health and the Nations Health 58(1):71-82.

- Simmons, David. 2011. "Testimony of David Simmons, Government Affairs Director for the National Indian Child Welfare Association before the Tribal Law and Order Commission." Tunalip, WA: U.S. Government Printing Office.
- Simoni, Jane M., Shalini Sehgal, and Karina L. Walters. 2004. "Triangle of Risk: Urban American Indian Women's Sexual Trauma, Injection Drug Use, and HIV Sexual Risk Behaviors." AIDS and Behavior 8(1):33-45.
- Sims, Barbara, Jim Ruiz, Ginger Weaver, and William L. Harvey. 2005. "Police Perceptions of Their Working Environment: Surveying the Small Department." International Journal of Police Science & Management 7(4):245-263.
- Sinden, Peter G. and B. Joyce Stephens. 1999. "Police Perceptions of Domestic Violence: The Nexus of Victim, Perpetrator, Event, Self and Law." Policing: An International Journal of Police Strategies & Management 22(3):313-327.
- Skelton, Ann. 2002. "Restorative Justice as a Framework for Juvenile Justice Reform: A South African Perspective." *British Journal of Criminology* 42(3):496-513.
- Skolnick, Jerome H. 1994. Justice Without Trial: Law Enforcement in Democratic Society (3rd edition). New York City, NY: Wiley.

-----. 1975. Justice Without Trial: Law Enforcement in Democratic Society (2nd edition). New York City, NY: Wiley.

- ------. 1966. Justice Without Trial: Law Enforcement in a Democratic Society. New York City, NY: John Wiley and Sons.
- Skolnick, Jerome H. and T.C. Gray. 1975. *Police in America*. Boston, MA: Little, Brown and Company.
- Skoog, Douglas M. 1996. "Taking Control: Native Self-Government and Native Policing." Pp. 118-131 in *Native Americans, Crime, and Justice* edited by Marianne O. Nielsen & Robert A. Silverman. Boulder, CO: Westview Press.
- Sliverman, Robert A. 1996. "Patterns of Native American Crime." Pp. 58-74 in *Native Americans, Crime, and Justice*, edited by Marianne O. Nielsen and Robert A. Silverman. Boulder, CO: Westview Press.
- Smith, A. B., B. Locke, and A. Fenster. 1970. "Authoritarianism in Policemen Who Are College Graduates and Non-College Graduates." *Journal of Criminal Law*, *Criminology and Police Science* 1(61):313-315.
- Smith, A. B., B. Locke, and W. F. Walker. 1968. "Authoritarianism in Police College Students and Non-Police College Students." *Journal of Criminal Law*, *Criminology, and Police Science* 1(59):440-443.

- Smith, Andrea. 2007. "Soul Wound: The Legacy of Native American Schools." *Amnesty International Magazine*. Retrieved on February 4, 2013 (http://www.amnestyusa.org/node/87342).
- -----. 2005. Conquest: Sexual Violence and American Indian Genocide. Cambridge, MA: South End Press.
- ------. 2004. "Boarding School Abuses, Human Rights, and Reparations." *Social Justice* 31(4):89-103.
- Smith, D., and J. Klein. 1984. "Police Control of Interpersonal Disputes." Social Problems 31(4):468-81.
- Smith, Douglas. 1986. "The Neighborhood Context of Police Behavior." Edited by Albert Reiss & Michael Tonry in *Crime and Justice* (Vol. 8). Chicago, IL: University of Chicago Press.
- Smith, Douglas A., Christy A. Visher, and L. Davidson. 1984. "Equity and Discretionary Justice: The Influence of Race on Police Arrest Decisions." *Journal of Criminal Law and Criminology* 1(75):234-249.
- Smith, Paul E. and Richard O. Hawkins. 1973. "Victimization, Types of Citizen-Police Contacts, and Attitudes toward the Police." *Law & Society Review* 8(1):135-152.
- Snipp, C. Matthew. 1992. "Sociological Perspectives on American Indians." Annual Review of Sociology 18(1):351-71.
- Snow, Dean R., and Kim M. Lanphear. 1988. "European Contact and Indian Depopulation in the Northeast: The Timing of the First Epidemics." *Ethnohistory* 35(1):15-33.
- Snowball, Lucy, and Don Weatherburn. 2008. "Theories of Indigenous Violence: A Preliminary Empirical Assessment." Australian & New Zealand Journal of Criminology 41 (2): 216-35.
- Snyder, H., and M. Sickmund. 2006. "Juvenile Offenders and Victims: 2006 National Report." Office of Justice Programs, U.S. Department of Justice, Office of Juvenile Justice and Delinquency Prevention. Washington, DC: U.S. Government Printing Office.
- Sparrow, M. K., M. H. Moore, and D. M. Kennedy. 1990. Beyond 911: A New Era for Policing. New York City, NY: Basic Books.
- Stalans, Loretta J. and Mary A. Finn. 2000. "Gender Differences in Officers' Perceptions and Decisions About Domestic Violence Cases." Women & Criminal Justice 11(3):1-24.

- Stangor, Charles, Laure Lynch, Changming Duan, and Beth Glas. 1992. "Categorization of Individuals on the Basis of Multiple Social Features." *Journal of Personality and Social Psychology* 62(2):207-218.
- Stanko, Elizabeth. 1985. Intimate Intrusions: Women's Experience of Male Violence. London: Routledge.
- Stannard, D. 1992. American Holocaust. New York, NY: Oxford.
- Staples, Robert. 1987. The Urban Plantation: Racism and Colonialism in the Post-Civil Rights Era. Los Angeles, CA: The Black Scholar Press.
- Steele, Paul, Nell Damon, and Kristene Denman. 2004. "Crime and the New Mexico Reservation: An Analysis of Crime on Native American Land (1996-2002)." National Institute of Justice. Rockville, MD: U.S. Government Printing Office.
- Steinert, Heinz. 2003. The Indispensable Metaphor of War: On Populist Politics and the Contradictions of the State's Monopoly of Force. *Theoretical Criminology* 7(3):265-291.
- Steinwachs, Donald M., Judith D. Kasper, and Elizabeth A. Skinner. 1992. *Final Report: NAMI Family Survey*. Arlington, VA: MacArthur Foundation, National Alliance for the Mentally Ill.
- Stephan, James J. 2004. State Prison Expenditures, 2001 Bureau of Justice Statistics Special Report. Washington D.C.: U.S. Department of Justice, Office of Justice Programs.
- Stewart, Daniel M. 2011. "Collaboration Between Federal and Local Law Enforcement: An Examination of Texas Police Chiefs' Perceptions." *Police Quarterly* 14(4): 407-430.
- Stewart, Daniel M., & Robert G. Morris. 2009. "A New Era of Policing?: An Examination of Texas Police Chiefs' Perceptions of Homeland Security." *Criminal Justice Policy Review* 20(3):290-309.
- Stewert, O. 1964. "Questions Regarding American Indian Criminality." *Human* Organization 23(1):61-66.
- Stitt, B. Grant, Mark Nichols, and David Giacopassi. 2003. "Does the Presence of Casinos Increase Crime? An Examination of Casino and Control Communities." *Crime & Delinquency* 49(2):253-284.
- Strauss, A. & Corbin, J. 1990. Basics of Qualitative Research: Grounded Theory Procedures and Techniques. Newbury Park, CA: Sage Publications.

- Strickland, C. June, and Michelle Cooper. 2011. "Getting into Trouble." Journal of Transcultural Nursing 22 (3): 240-47.
- Stubbs, Julie, and Julia Tolmie. 2008. "Battered Women Charged with Homicide: Advancing the Interests of Indigenous Women." *Australian & New Zealand Journal of Criminology* 41 (1): 138-61.
- Substance Abuse and Mental Health Services Administration, Center for Behavioral Health Statistics and Quality. 2011. "The NSDUH Report: Substance Use among American Indian or Alaska Native Adolescents." Rockville, MD. Substance Abuse and Mental Health Services Administration Retrieved on March 12, 2013 (http://www.samhsa.gov/data/2k11/WEB_SR_005/WEB_SR_005.htm).
- Substance Abuse and Mental Health Services Administration, Office of Applied Studies. 2010. "The NSDUH Report: Substance Use among American Indian or Alaska Native Adults." Rockville, MD. Substance Abuse and Mental Health Services Administration. Retrieved on March 12, 2013 (http://www.samhsa.gov/data/2k10/182/AmericanIndian.htm).
- Sun, Ivan Y., Brian K. Payne, and Yuning Wu. 2008. "The Impact of Situational Factors, Officer Characteristics, and Neighborhood Context on Police Behavior: A Multilevel Analysis." Journal of Criminal Justice 36(1):22-32.
- Sun, Ivan Y., and Ruth A. Triplett. 2008. "Differential Perceptions of Neighborhood Problems by Police and Residents: The Impact of Neighborhood-Level Characteristics." Policing: An International Journal of Police Strategies & Management 31(3):435-455.
- Sung, H. 2002. The Fragmentation of Policing in American Cities: Toward an Ecological Theory of Police-Citizen Relations Westport, CT: Praeger.
- Swem, E. G. 1928. "Some Notes on the Four Forms of the Oldest Building of William and Mary College." *The William and Mary Quarterly* 8(4):217-307.
- Talarico, Susette M. and Charles R. Swanson. 1982. "Police Perceptions and Job Satisfaction." Work and Occupations 9(1):59-78.
- Tatum, Becky. 2000. Crime, Violence and Minority Youths. Aldershot, NY: Ashgate.
- Taylor, Ian. 1999. Crime in Context. Cambridge, MA: Polity Press.
- Taylor, Shelley E., Susan T. Fiske, Nancy L. Etcoff, and Audrey J. Ruderman. 1978. "Categorical and Contextual Bases of Person Memory and Stereotyping." *Journal of Personality and Social Psychology* 36(7):778-793.

- Terrill, William, and Michael D. Reisig, 2003. "Neighborhood Context and Police Use of Force." Journal of Research in Crime and Delinquency 40(3):291-321.
- Theriot, Matthew T., and Barbara "Sunshine" Parke. 2008. "Native American Youth Gangs." Journal of Ethnicity in Criminal Justice 5(4):83-97.
- Theobald, Delphine, David P. Farrington, and Alex R. Piquero. 2013. "Childhood Broken Homes and Adult Violence: An Analysis of Moderators and Mediators." *Journal* of Criminal Justice, 41(1):44-52.
- Tittle, Charles R., and Alan R. Rowe. 1974. "Certainty of Arrest and Crime Rates: A Further Test of the Deterrence Hypothesis." *Social Forces* 52(4):455-62.
- Tjaden, P., and N. Thoennes. 2006. "Extent, Nature and Consequences of Rape Victimization: Findings from the National Violence against Women Survey." National Institute of Justice, U.S. Department of Justice. Washington, DC: U.S. Government Printing Office.
- 2000. "Full Report of the Prevalence, Incidence, and Consequences of Violence against Women: Findings from the National Violence against Women Survey." National Institute of Justice, U.S. Department of Justice. Washington, DC: U.S. Government Printing Office.
- Tong, R. 1984. Women, Sex and Law. Totowa, NJ: Romwan & Allanheld.
- Tonry, Michael. 1994. "Racial Politics, Racial Disparities, and the War on Crime." Crime & Delinquency, 40(4):475-494.
- Toth, Reid C., Gordon A. Crews, and Catherine E. Burton. 2007. In the Margins: Special Populations and American Justice. Saddle Creek, NJ: Prentice Hall.
- Truxillo, D. M., S. R. Bennett, and M. L. Collins. 1998. "College Education and Police Job Performance: A Ten Year Study." *Public Personnel Management* 27(2): 269– 280.
- Turner, Victor. 1974. Dramas, Fields and Metaphor: Symbolic Action in Human Society. Ithaca, NY: Cornell University Press.
- Tyre, M., and S. Braunstein. 1992. "Higher Education and Ethical Policing." FBI Law Enforcement Bulletin June: 6-10.
- United States Bureau of Indian Affairs. 2006. "BIA Funded School Adequate Yearly Progress 2004-2005." Washington, DC: U.S. Government Printing Office Retrieved on February 15, 2013 (http://www.oiep.bia.edu/).

- United States Bureau of Indian Affairs. 2007. "U.S. Bureau of Indian Affairs, 2007." Washington, DC: U.S. Government Printing Office Retrieved on February 15, 2013 (http://www.doi.gov/bureau-indian-affairs.html).
- United States Census Bureau. 2009. "American Indian and Alaska Native Heritage." Washington, DC: U.S. Government Printing Office. Retrieved on February 15, 2013.<http://www.census.gov/PressRelease/www/releases/archives/facts_for_feat ures_special_editions/007489.html>.
- United States Census Bureau. 2010. "American Community Survey for the American Indian and Alaska Native alone population." Washington, DC: U.S. Government Printing Office. Retrieved on February 15, 2013 (http://factfinder2.census.gov).
- United States Department of Commerce. 2009. "Census Bureau, Decennial Census, 1990 and 2000." Washington, DC: U.S. Government Printing Office.
- United States Department of Justice and U.S. Department of Interior. 2011."Tribal Law and Order Act (TLOA) Long Term Plan to Build and Enhance Tribal Justice Systems." Washington, DC: U.S. Department of Justice & U.S. Department of Interior.
- United States General Accounting Office. 1998. Casino Gaming Regulation: Roles of Five States and the National Indian Gaming Commission. Washington, DC: U.S. Government Printing Office.
- United States Indian Office. 1869. "Annual Report of the Commissioner of Indian Affairs (Commissioner's Report)." Washington, DC: U. S. Government Printing Office.
- United States Renal Data System. 2003. "Annual Data Report: Atlas of End-Stage Renal Disease in the United States." Government Printing Office, Washington DC: National Institutes of Health.
- United States Senate Committee on Indian Affairs. 2002. "Hearing on Contemporary Tribal Governments: Challenges in Law Enforcement Related to the Rulings of the United States Supreme Court: Testimony of Tracy Toulou, Director of Tribal Justice." United States Senate Committee on Indian Affairs. Washington D.C.: U.S. Government Printing Office.
- Velez, Maria B., Lauren J. Krivo, and Ruth D. Peterson. 2003. "Structural Inequality and Homicide: An Assessment of the Black-White Gap in Killings." Criminology 41(3):645-72.
- Vera Sanchez, Claudio G. and Dennis P. Rosenbaum. 2011. "Racialized Policing: Officers' Voices on Policing Latino and African American Neighborhoods." Journal of Ethnicity in Criminal Justice 9(2):152-178.

- Vernon, Irene S., and Roe Bubar. 2001. "Child Sexual Abuse and HIV/AIDS in Indian Country." Wicazo Sa Review 16(1):47-63.
- Vigil, Donna. 2006. "Division of Health Programs, White Mountain Apache Tribe. Oral Testimony Offered to the U.S. Senate Committee on Indian Affairs. Hearing: Indian Youth Suicide." Washington, DC: U.S. Government Printing Office.
- Vila, Bryan. 2006. "Impact of Long Work Hours on Police Officers and the Communities They Serve." American Journal of Industrial Medicine, 49(11):972-980.
- Von Hentig, Hans. 1945. "The Delinquency of the American Indian." Journal of Criminal Law and Criminology (1931-1951) 36 (2): 75-84.
- Waddington P. A. J. 1999. *Policing Citizens: Authority and Rights*. London: University College Press.
- Wahab, Stéphanie, and Lenora Olson. 2004. "Intimate Partner Violence and Sexual Assault in Native American Communities." *Trauma, Violence, & Abuse* 5(4): 353-66.
- Wakeling, Stewart, and Miriam Jorgensen. 2008. "Strengthening and Rebuilding Tribal Justice Systems: A Participatory Outcomes Evaluation of the U.S. Department of Justice Comprehensive Indian Resources for Community and Law Enforcement (Circle) Project." Office of Justice Programs U.S. Department of Justice, National Institute of Justice. Washington, DC: U.S. Government Printing Office.
- Wakeling, Stewart, Miriam Jorgensen, Susan Michaelson, and Manley Begay.
 2001."Policing on American Indian Reservations." National Institute of Justice.
 Washington DC: U.S. Government Printing Office.
- Walch, Michael C. 1983. "Terminating the Indian Termination Policy." Stanford Law Review 35(6)1181-1215.
- Walker, R. Dale, M. Dow Lambert, Patricia Silk Walker, Daniel R. Kivlahan, Dennis M. Donovan, and Matthew O. Howard. 1993. "Alcohol Abuse in Urban Indian Adolescents and Women: A Longitudinal Study for Assessment and Risk Evaluation." American Indian and Alaska Native Mental Health Research 7(1):1-47.
- Walker, Samuel. 2011. Sense and Nonsense About Crime and Drugs: About Crime, Drugs, and Communities. Belmont, CA: Wadsworth.
- Wallace J.D., A.D. Calhoun, K.E. Powell, J. O'Neil and S.P. James (1996), Homicide and Suicide Among Native Americans, 1979–1992. Atlanta, GA: Centers for Disease Control and Prevention, National Center for Injury Prevention and Control. Violence Surveillance Series, No. 2.

- Wallbank, Derek. 2010."Senators Demand National Investigation into Indian Health Service Mismanagement." *MinnPost* December 5. Retrieved on February 4, 2013 (http://www.minnpost.com/dc-dispatches/2010/09/senators-demand-nationalinvestigation-indian-health-service-mismanagement).
- Wardman, D., N. El-Guebaly, and D. Hodgins. 2001."Problem and Pathological Gambling in North American Aboriginal Populations: A Review of the Empirical Literature." Journal of Gambling Studies 17(2):81-100.
- Washburn, Wilcomb E. 1971. Red Man's Land/White Man's Law: The Past and Present Status of the American Indian. Norman, OK: University of Oklahoma.
- Weatherburn, Don. 2010. "Indigenous Violence." Australian & New Zealand Journal of Criminology 43(2):197-198.
- Weatherburn, Don, Lucy Snowball, and Boyd Hunter. 2008. "Predictors of Indigenous Arrest: An Exploratory Study." Australian & New Zealand Journal of Criminology 41(2):307-322.
- Websdale, Neil. 1999. Understanding Domestic Homicide. Boston, MA: Northeastern University Press.

-----. 1995. "An Ethnographic Assessment of the Policing of Domestic Violence in Rural Eastern Kentucky." *Social Justice* 22(1):102-122.

- Websdale, Neil, and Byron Johnson. 1997. "The Policing of Domestic Violence in Rural and Urban Areas: The Voices of Battered Women in Kentucky." *Policing and Society* 6(4):297-317.
- Weiner, N. 1976. "The Educated Policeman." Journal of Police Science and Administration 1(1):450-457.
- Weitekamp, Elmar. 1993. "Reparative Justice." European Journal on Criminal Policy and Research 1(1):70-93.
- Weitekamp, Elmar G.M., Stephan Parmentier, Kris Vanspauwen, Marta valinas and Roel Gerits. 2006. "How to Deal with mass Victimization and Gross Human Rights Violations: A Restorative Approach." Pp. 217-241 in Large-Scale Victimization as a Potential Source of Terrorist Activities edited by Uwe Ewald and Ksenija Turkovic. Amsterdam: IOS Press.
- Weitzer, Ronald and Steven A. Tuch. 2002. "Perceptions of Racial Profiling: Race, Class, and Personal Experience." Criminology 40(2):435-456.
- Welch, Kelly. 2007. "Black Criminal Stereotypes and Racial Profiling." Journal of Contemporary Criminal Justice 23(3):276-88.

- Weller, Michelle, Lorraine Hope, and Lorraine Sheridan. 2013. "Police and Public Perceptions of Stalking: The Role of Prior Victim–Offender Relationship." *Journal of Interpersonal Violence*, 28(2):320-339.
- Wellman, Paul I. 1934. The Indian Wars of the West. New York City, NY: Doubleday & Company.
- Wells, Edward L. and Joseph H. Rankin. 1986. "The Broken Homes Model of Delinquency: Analytic Issues." Journal of Research in Crime and Delinquency 23(1):68-93.
- Wells, William and Joseph A. Schafer. 2006. "Officer perceptions of police responses to persons with a mental illness." *Policing* 29(4):578-601.
- Wemmers, Jo-Anne, and Marisa Canuto. 2002. "Victims' Experiences with Expectations and Perceptions of Restorative Justice: A Critical Review of the Literature." edited by Policy Centre for Victim Issues Research and Statistics Division, Department of Justice. Ottawa, Ontario: Canada.
- Wentz, Ericka and Carol A. Archbold. 2012. "Police Perceptions of Sexual Assault Victims: Exploring the Intra-Female Gender Hostility Thesis." *Police Quarterly* 15(1):25-44.
- Westley, W.A. 1970. Violence and the Police: A Sociological Study of Law, Custom, and Morality. Cambridge, MA: MIT Press.
- Worden, R. E. 1995. "Police Officers' Belief Systems: A Framework for Analysis." American Journal of Police 14(1):49-81.
- Whitbeck, Les B., Dan R. Hoyt, Barbara J. McMorris, Xiaojin Chen, and Jerry D. Stubben. 2001. "Perceived Discrimination and Early Substance Abuse among American Indian Children." *Journal of Health and Social Behavior* 42(4):405-424.
- Williams, Frank P., and Marilyn D. McShane. 2010. Criminological Theory. Upper Saddle River, NJ: Prentice Hall.
- Williams, Timothy. 2012a. "Higher Crime, Fewer Charges on Indian Land." The New York Times. Retrieved on February 4, 2013 (http://www.nytimes.com/2012/02/21/us/on-indian-reservations-higher-crimeand-fewer-prosecutions.html?pagewanted=all&_r=0).
- -----. 2012b. "Washington Steps Back From Policing Indian Lands, Even as Crime Rises" Retrieved on February 4, 2013 http://www.nytimes.com/2012/11/13/us/ascrime-rises-on-indian-lands-policing-is-cut-back.html?emc=eta1&_r=1&

- -----. 2012c. "Brutal Crimes Grip an Indian Reservation." *The New York Times*. Retrieved on February 4, 2013 (http://www.nytimes.com/2012/02/03/us/wind-riverindian-reservation-where-brutality-is-banal.html?_r=1&).
- Williams, Walter L. 1979. "Cherokee History: An Analysis of Recent Studies." American Indian Quarterly 5(4): 47-54.

Williams v. Lee, 358 U.S. 217 (1959)

- Winfree, L. Thomas, Curt T. Griffiths, and Christine S. Sellers. 1989. "Social Learning Theory, Drug Use, and American Indian Youths: A Cross-Cultural Test." Justice Quarterly 6(3):395-417.
- Witmer, Linda F. 1993. The Indian Industrial School, Carlisle, Pennsylvania, 1879-1918. Carlisle, PA: Cumberland County Historical Society.
- Wood, Darryl S. 2002. "Explanations of Employment Turnover among Alaska Village Public Safety Officers." *Journal of Criminal Justice* 30 (3): 197-215.
- Wood, Darryl S., and Lawrence C. Trostle. 1997. "The Nonenforcement Role of Police in Western Alaska and the Eastern Canadian Arctic: An Analysis of Police Tasks in Remote Arctic Communities." *Journal of Criminal Justice* 25(5):367-79.
- Worden, R., and A. Pollitz. 1984. "Police Arrests in Domestic Disturbances: A Further Look." Law and Society Review 18(1):105-19.
- Wyatt, Gail E., Julie Axelrod, Dorothy Chin, Jennifer Vargas Carmona, and Tamra Burns Loeb. 2000. "Examining Patterns of Vulnerability to Domestic Violence among African American Women." *Violence Against Women* 6(5):495-514.
- Yabiku, Scott T., Andrea Dixon Rayle, Scott K. Okamoto, Flavio F. Marsiglia, and Stephen Kulis. 2007. "The Effect of Neighborhood Context on the Drug Use of American Indian Youth of the Southwest." *Journal of Ethnicity in Substance Abuse* 6(2):181-204.

Young, Jock. 2007. The Vertigo of Late Modernity. London: Sage

- 2003. "Merton with Energy, Katz with Structure: The Sociology of Vindictiveness and the Criminology of Transgression." *Theoretical Criminology*. 7(1): 88-414.
- Yuan, Nicole P., Mary P. Koss, Mona Polacca, and David Goldman. 2006. "Risk Factors for Physical Assault and Rape among Six Native American Tribes." *Journal of Interpersonal Violence* 21(12):1566-1590.
- Zatz, Majorie S., Lugan Chiago, and Z.K. Snyder-Joy. 1991. "American Indians and Criminal Justice: Some Conceptual and Methodological Considerations." In *Race*

and Criminal Justice, edited by M. J. Lynch and E. B. Patterson. New York, NY: Harrow and Heston.

- Zion, James W. 2006. "Justice as Phoenix: Traditional Indigenous Law, Restorative Justice, and the Collapse of the State." Pp.51-66 in Native Americans and the Criminal Justice System edited by Jeffrey Ian Ross and Larry Gould. Boulder, CO: Paradigm Publishers.
- -----. 1998. "The Dynamics of Navajo Peacemaking." Journal of Contemporary Criminal Justice 14(1): 58-74.
- Zorza, Joan. 1992. "The Criminal Law of Misdemeanor Domestic Violence, 1970-1990." The Journal of Criminal Law and Criminology 83(1):46-72.

VITA

Favian Alejandro Martín Old Dominion University Department of Sociology and Criminology Norfolk, Virginia 23529

Ph.D. Criminology & Criminal Justice, Old Dominion University, 2013 M.A. Criminal Justice, Pennsylvania State University, Harrisburg, 2009 B.S. Criminal Justice, Pennsylvania State University, Harrisburg, 2007

PUBLICATIONS, PEER REVIEWED

Collins, Victoria. E., Farrell, Amanda. L., McKee, Jesse. R., Martín, Favian A., and Monk-Turner, Elizabeth. 2011. "The State of Coverage: The Media's Representation of International Issues and State Crime." *International Criminal Justice Review*, 21(1), 5-21.

Gabbidon, Shaun. L., Higgins, George. E., Martín, Favian A. Nelson, Matthew., and Brown, Jimmy. 2011. "An Exploratory Analysis of Federal Litigation in the United States Challenging Asset Forfeiture." *Criminal Justice Policy Review*, 22(1), 50-64.

Gabbidon, Shaun. L., and Martín, Favian A. 2010. "An Era-Based Exploration of the Most Significant Books in Criminology/Criminal Justice: A Research Note." *Journal of Criminal Justice Education*, 21(3), 348-369.

Gabbidon, Shaun. L., Higgins, George., and Martín, Favian A. 2010. "Press Rankings in Criminology/Criminal Justice: A Preliminary Assessment of Book Publisher Quality." *Journal of Criminal Justice Education*, 21(3), 229-244.

Higgins, George. E., Gabbidon, Shaun. L., and Martín, Favian A. 2010. "The Role of Race/Ethnicity and Race Relations on Public Opinion Related to the Immigration and Crime Link." *Journal of Criminal Justice*, 38(1), 51-56.

Gabbidon, Shaun. L., Higgins, George. E., and Martín, Favian A. 2010. "Moving through the Faculty Ranks: An Exploratory Study on the Perceived Importance of Book Publishing and Publishing in Peer-Reviewed Journals in Criminology/Criminal Justice." *Journal of Criminal Justice Education*, 22(2), 165-180.